UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2004

or

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to _____

Commission File Number 0-7201



BROWN & BROWN, INC.

(Exact Name of Registrant as Specified in its Charter)

Florida (State or other jurisdiction of incorporation or organization) 59-0864469 (I.R.S. Employer Identification Number)

220 S. Ridgewood Ave., Daytona Beach, FL (Address of Principal Executive Offices) 32114 (Zip Code)

Registrant's telephone number, including area code: (386) 252-9601

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months, and (2) has been subject to such filing requirements for the past 90 days. Yes 🛛 No 🗆

Indicate by check mark whether the Registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes 🗵 No 🗆

The number of shares of the Registrant's common stock, \$.10 par value, outstanding as of August 2, 2004 was 68,904,401.

BROWN & BROWN, INC.

INDEX

PAGE

PART I. FINANCIAL INFORMATION

Item 1.	Financial Statements (Unaudited)	
	Condensed Consolidated Statements of Income for the three and six months ended June 30, 2004 and 2003	3
	Condensed Consolidated Balance Sheets as of June 30, 2004 and December 31, 2003	4
	Condensed Consolidated Statements of Cash Flows for the six months ended June 30, 2004 and 2003	5
	Notes to Condensed Consolidated Financial Statements	6
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	13
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	20
Item 4.	Controls and Procedures	21
PART II. O	THER INFORMATION	
Item 1.	Legal Proceedings	21
Item 4.	Submission of Matters to a Vote of Security Holders	21
Item 6.	Exhibits and Reports on Form 8-K	22
SIGNATUR		24

PART I - FINANCIAL INFORMATION

ITEM 1 - FINANCIAL STATEMENTS (UNAUDITED)

BROWN & BROWN, INC. CONDENSED CONSOLIDATED STATEMENTS OF INCOME (UNAUDITED) (in thousands, except per share data)

		For the three months ended June 30,			For the six months ended June 30,			
		2004		2003		2004		2003
REVENUES								
Commissions and fees	\$	156,749	\$	137,257	\$	321,063	\$	281,509
Investment income		333		442		1,021		775
Other income, net		860		159		1,423		310
Total revenues		157,942		137,858		323,507		282,594
EXPENSES								
Employee compensation and benefits		76,270		66,092		152,552		134,333
Non-cash stock grant compensation		665		632		1,510		1,449
Other operating expenses		19,983		19,229		41,379		38,635
Amortization		5,483		4,416		10,300		8,753
Depreciation		2,269		2,019		4,423		3,946
Interest		743		946		1,454		1,953
Total expenses		105,413		93,334		211,618		189,069
Income before income taxes		52,529		44,524		111,889		93,525
Income taxes		20,376		16,589		43,388		35,054
NET INCOME	\$	32,153	\$	27,935	\$	68,501	\$	58,471
Net income per share:					_		_	
Basic	\$	0.47	\$	0.41	\$	1.00	\$	0.86
Diluted	\$	0.46	\$	0.41	\$	0.99	\$	0.85
			-		_		-	
Weighted average number of shares outstanding:								
Basic	_	68,790		68,270		68,736		68,222
Diluted		69,370		68,943	_	69,283		68,927
Dividends declared per share	\$	0.07	\$	0.0575	\$	0.14	\$	0.115
	_		_		_			

See accompanying notes to condensed consolidated financial statements.

BROWN & BROWN, INC. CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED) (in thousands, except per share data)

	June 30, 2004	December 31 2003
ASSETS		
Current assets:		
Cash and cash equivalents	\$ 36,384	\$ 56,926
Restricted cash	133,194	116,543
Short-term investments	386	382
Premiums, commissions and fees receivable	146,669	146,672
Other current assets	17,833	22,943
Total current assets	334,466	343,466
Fixed assets, net	32,990	32,396
Goodwill	300,732	237,753
Amortizable intangible assets, net	308,361	232,934
Investments	10,353	10,845
Other assets	7,197	8,460
Total assets	\$ 994,099	\$ 865,854
LIABILITIES AND SHAREHOLDERS' EQUITY		
Current liabilities:		
Premiums payable to insurance companies	\$ 216,788	\$ 199,628
Premium deposits and credits due customers	23,769	22,223
Accounts payable	20,163	11,282
Accrued expenses	38,474	49,69
Current portion of long-term debt	68,455	18,692
Total current liabilities	367,649	301,516
Long-term debt	34,313	41,102
Deferred income taxes, net	17,176	15,018
Other liabilities	9,063	10,178
Shareholders' equity:		
Common stock, par value \$.10 per share; authorized 280,000 shares; issued and outstanding, 68,808 shares at		
2004 and 68,561 at 2003	6,881	6,850
Additional paid-in capital	178,577	170,130
Retained earnings	375,695	316,822
Accumulated other comprehensive income	4,745	4,222
Total shareholders' equity	565,898	498,035
Total liabilities and shareholders' equity	\$ 994,099	\$ 865,854

See accompanying notes to condensed consolidated financial statements.

BROWN & BROWN, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (in thousands)

	For the si ended J	
	2004	2003
CASH FLOWS FROM OPERATING ACTIVITIES:		
Net income	\$ 68,501	\$ 58,471
Adjustments to reconcile net income to net cash provided by operating activities:		
Amortization	10,300	8,753
Depreciation	4,423	3,946
Non-cash stock grant compensation	1,510	1,449
Deferred income tax provision (benefit)	1,121	(71)
Income tax benefit from exercise of stock options	56	3,530
Net gains on sales of investments, fixed assets and customer accounts	(1,747)	(86)
Changes in operating assets and liabilities, net of effect from insurance agency acquisitions and disposals:		, í
Restricted cash, (increase)	(16,651)	(7,213)
Premiums, commissions and fees receivable, (increase)	(413)	(8,304)
Other assets, decrease	11,394	1,798
Premiums payable to insurance companies, increase	16,291	12,427
Premium deposits and credits due customers, increase	1,534	3,266
Accounts payable, increase	7,859	2,890
Accrued expenses, (decrease)	(11,777)	(4,906)
Other liabilities, (decrease) increase	(36)	355
NET CASH PROVIDED BY OPERATING ACTIVITIES	92,365	76,305
CASH FLOWS FROM INVESTING ACTIVITIES:		
Additions to fixed assets	(4 E2E)	(3,960)
Payments for businesses acquired, net of cash acquired	(4,535) (143,555)	(66,569)
Proceeds from sales of fixed assets and customer accounts	2,339	1,353
Proceeds from sales of fixed assets and customer accounts Purchases of investments	2,359	
		(5)
Proceeds from sales of investments	738	
NET CASH USED IN INVESTING ACTIVITIES	(145,013)	(69,181)
CASH FLOWS FROM FINANCING ACTIVITIES:		
Borrowings on revolving credit facility	50,000	_
Payments on long-term debt	(8,928)	(11,227)
Issuance of common stock for employee stock benefit plans	662	238
Purchase of common stock for employee stock benefit plans		(2,334)
Cash dividends paid	(9,628)	(7,852)
Cash distribution to minority interest shareholders	(0,020)	(2,890)
		(2,050)
NET CASH PROVIDED BY (USED IN) FINANCING ACTIVITIES	32,106	(24,065)
Net decrease in cash and cash equivalents	(20,542)	(16,941)
Cash and cash equivalent at beginning of period	56,926	68,050
CASH AND CASH EQUIVALENTS AT END OF PERIOD	\$ 36,384	\$ 51,109

See accompanying notes to condensed consolidated financial statements.

BROWN & BROWN, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

Note 1 - Basis of Financial Reporting

The accompanying unaudited, condensed, consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America for interim financial information and with the instructions for Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. These unaudited, condensed, and consolidated financial statements should be read in conjunction with the audited consolidated financial statements and the notes thereto set forth in the Company's Annual Report on Form 10-K for the year ended December 31, 2003.

Results of operations for the three and six months ended June 30, 2004 are not necessarily indicative of the results that may be expected for the year ending December 31, 2004.

Certain amounts for the prior period have been reclassified to conform to the current period presentation.

Note 2 - Stock-Based Compensation and Incentive Plans

The Company applies the intrinsic value-based method of Accounting Principles Board ("APB") Opinion No. 25, "Accounting for Stock Issued to Employees", to account for its stock plans. Accordingly, the Company presents the disclosure requirements of Statement of Financial Accounting Standards ("SFAS") No. 148, "Accounting for Stock-based Compensation - Transition and Disclosure", which requires presentation of pro forma net income and earnings per share information under SFAS No. 123 (same title).

Pursuant to the above disclosure requirements, the following table provides an expanded reconciliation for all periods presented that: 1.) adds back to reported net income the recorded expense under APB No. 25, net of related income tax effects; 2.) deducts the total fair value expense under SFAS No. 123, net of related income tax effects; and 3.) shows the reported and pro forma earnings per share amounts (in thousands, except per share data):

	For the three months ended June 30,			ix months June 30,
	2004	2003	2004	2003
Net income, as reported	\$32,153	\$27,935	\$68,501	\$58,471
Total stock-based employee compensation cost included in the determination of net income, net of related tax effects	408	392	926	898
Total stock-based employee compensation cost determined under fair value method for all awards, net of related tax effects	(1,340)	(950)	(2,790)	(1,859)
Pro forma net income	\$31,221	\$27,377	\$66,637	\$57,510
Earnings per share:				
Basic, as reported	\$ 0.47	\$ 0.41	\$ 1.00	\$ 0.86
Basic, pro forma	\$ 0.45	\$ 0.40	\$ 0.97	\$ 0.84
Diluted, as reported	\$ 0.46	\$ 0.41	\$ 0.99	\$ 0.85
Diluted, pro forma	\$ 0.45	\$ 0.40	\$ 0.96	\$ 0.83

Note 3 - Basic and Diluted Net Income Per Share

The following table sets forth the computation of basic net income per share and diluted net income per share (in thousands, except per-share data):

		For the three months ended June 30,		ix months June 30,
	2004	2003	2004	2003
Net income	\$32,153	\$27,935	\$68,501	\$58,471
Weighted average number of shares outstanding	68,790	68,270	68,736	68,222
Dilutive effect of stock options using the treasury stock method	580	673	547	705
Weighted average number of shares outstanding	69,370	68,943	69,283	68,927
Net income per share:				
Basic	\$ 0.47	\$ 0.41	\$ 1.00	\$ 0.86
Diluted	\$ 0.46	\$ 0.41	\$ 0.99	\$ 0.85

Note 4 – Acquisitions

On May 1, 2004, the Company acquired the outstanding stock of Proctor Homer Warren, Inc., d/b/a Proctor Financial Insurance ("Proctor"). The purchase price for this acquisition was approximately \$33.9 million, including \$31.1 million of net cash payments and the assumption of \$2.8 million of net liabilities. Additionally, the purchase price may be increased by "earn-out" contingent payments of up to \$12.4 million based upon the net operating profits earned by Proctor over a three-year period. This agency provides lender-placed insurance and other specialty insurance products primarily to financial institutions. This acquisition allowed the Company to expand into this new market niche.

Additionally, during the second quarter of 2004, the Company acquired certain assets and liabilities of seven other general insurance agencies, the outstanding stock of one general insurance agency and several books of business (customer accounts). The aggregate purchase price for these acquisitions was approximately \$18.1 million, including \$14.7 million of net cash payments, the issuance of notes payable of \$0.8 million and the assumption of \$2.6 million of net liabilities. Additionally, \$0.7 million was paid during the quarter on the "earn-out" purchase price agreements of prior acquisitions, which included \$1.7 million of cash payments, the issuance of \$0.2 million in notes payable, the forgiveness of a note payable obligation of \$1.3 million and the assumption of \$0.1 million of liabilities.

On February 1, 2004, the Company acquired certain insurance-related assets of Doyle Consulting Group, Inc. and Doyle Consulting Group of New Jersey, Inc. (collectively, "Doyle"). On March 1, 2004, the Company acquired certain insurance-related assets of Waldor Agency, Inc. ("Waldor") and Statfeld Vantage Insurance Group, L.L.C. ("Statfeld"). The assets acquired from Doyle, Waldor and Statfeld were purchased with cash payments of \$10.7 million, \$30.4 million and \$26.6 million, respectively. Additionally, the purchase price of these acquisitions may be increased by "earn-out" contingent payments of up to \$6.7 million, \$9.1 million and \$7.9 million, respectively, based upon the respective net operating profits earned by the Company from the acquired assets over a two-year period. These three agencies are providers of a broad range of property and casualty, and employee benefits insurance products and services. These acquisitions expanded the Company's Pennsylvania and New Jersey retail insurance operations.

Additionally, during the first quarter of 2004, the Company acquired certain assets and liabilities of nine other general insurance agencies and several books of business (customer accounts). The aggregate purchase price for these acquisitions was approximately \$26.2 million, including \$25.6 million of cash payments, the issuance of notes payable of \$0.6 million and the assumption of less then \$0.1 million of liabilities. Additionally, \$9.1 million was paid during the quarter on the "earn-out" purchase price agreements of prior acquisitions, which included \$2.9 million of cash payments and the issuance of 200,000 shares of the Company's stock with an approximate fair market value of \$6.2 million.

The "earn-out" contingent payments represent the maximum amount of additional consideration that could be paid pursuant to the purchase agreements related to these acquisitions. These contingent obligations are primarily based upon future net operating profits earned by the Company from the acquired entities or assets, and were not included in the purchase price that was recorded for these acquisitions at their respective dates. Future payments made under these agreements, if any, will be recorded as upward adjustments to goodwill when the contingencies are settled. Assuming each acquisition maximizes their total potential earn-out, the aggregated amount of unrecorded contingent payables outstanding as of June 30, 2004 for all consummated acquisitions would be \$100.2 million.

The following table summarizes the preliminary allocation of the aggregate purchase price to the fair values of the assets and liabilities acquired, including "earn-out" adjustments on prior acquisitions (in thousands):

	Doyle	Waldor	Statfeld	Proctor	Other	Total
Current assets	\$ —	\$ —	\$ —	\$ 786	\$ 225	\$ 1,011
Fixed assets	100	50	50	200	414	814
Purchased customer accounts	6,385	16,623	16,175	24,499	24,426	88,108
Noncompete agreements	151	31	11	_	306	499
Goodwill	4,069	13,707	10,383	8,410	28,680	65,249
Total assets acquired	10,705	30,411	26,619	33,895	54,051	155,681
Current liabilities	_	_	_	(2,838)	(2,078)	(4,916)
Deferred income tax	—	_	_	_	(658)	(658)
Total liabilities assumed	_		_	(2,838)	(2,736)	(5,574)
Net assets acquired	\$ 10,705	\$ 30,411	\$26,619	\$31,057	\$51,315	\$150,107

The results of operations for the acquisitions completed during 2004 have been combined with those of the Company since their respective acquisition dates. If the acquisitions had occurred as of January 1, 2003, the Company's results of operations would be as shown in the following table (in thousands, except per share data):

	For the three months ended June 30,			For the six months ended June 30,			
	 2004		2003		2004		2003
Total revenues	\$ 160,146	\$ 1	54,632	\$ 3	338,371	\$	318,312
Income before income taxes	53,194		49,992	-	116,833		105,099
Net income	32,560		31,325		71,528		65,647
Net income per share:							
Basic	\$ 0.47	\$	0.46	\$	1.04	\$	0.96
Diluted	\$ 0.47	\$	0.45	\$	1.03	\$	0.95
		_				_	

These pro forma results are not necessarily indicative of the actual results of operations that would have occurred had the acquisitions actually been made at the beginning of the respective periods.

Note 5 - Goodwill and Amortizable Intangible Assets

The Company accounts for goodwill under SFAS No. 142, "Goodwill and Other Intangible Assets". SFAS No. 142 provides for the non-amortization of goodwill. Goodwill is now subject to at least an annual assessment for impairment by applying a fair value-based test. Amortizable intangible assets will be amortized over their useful lives (other than indefinite life assets) and will be subject to a lower of cost or market impairment testing.

SFAS No. 142 requires the Company to compare the fair value of each reporting unit with its carrying amount to determine if there is potential impairment of goodwill. If the fair value of the reporting unit is less than its carrying value, an impairment loss would be recorded to the extent that the fair value of the goodwill within the reporting unit is less than its carrying value. Fair value is estimated based on multiples of revenues, earnings before interest, income taxes, depreciation and amortization (EBITDA), and discounted cash flows. The Company completed its most recent annual assessment as of November 30, 2003 and identified no impairment as a result of the evaluation.

The changes in goodwill, net of accumulated amortization, by business segment, for the six months ended June 30, 2004 are as follows (in thousands):

	Retail	National Programs	Brokerage	Services	Total
Balance as of December 31, 2003	\$168,135	\$60,695	\$ 8,867	\$ 56	\$237,753
Goodwill acquired	55,254	9,825	170		65,249
Goodwill disposed of relating to sale of businesses	(2,270)		—	—	(2,270)
		<u> </u>			
Balance as of June 30, 2004	\$ 221,119	\$70,520	\$ 9,037	\$ 56	\$300,732

Amortizable intangible assets as of June 30, 2004 and December 31, 2003 consisted of the following (in thousands):

	June 30, 2004					December	r 31, 2003	
	Gross Carrying Value	Accumulated Amortization	Net Carrying Value	Weighted Average Life (Yrs)	Gross Carrying Value	Accumulated Amortization	Net Carrying Value	Weighted Average Life (Yrs)
Purchased Customer								
Accounts	\$ 385,512	\$ (86,107)	\$ 299,405	18.8	\$ 300,236	\$ (77,408)	\$ 222,828	18.4
Noncompete								
Agreements	32,734	(23,778)	8,956	7.6	32,283	(22,177)	10,106	7.7
		+ (100 00F)	* 200 201		+ 000 F10		<u> </u>	
Total	\$ 418,246	\$ (109,885)	\$ 308,361		\$ 332,519	\$ (99,585)	\$ 232,934	

Amortization expense for amortizable intangible assets for the years ended December 31, 2004, 2005, 2006, 2007 and 2008 are estimated to be \$21.4 million, \$21.9 million, \$20.6 million, \$20.0 million and \$19.2 million, respectively.

Note 6 - Long-Term Debt

In January 2001, the Company entered into a \$90.0 million unsecured seven-year term loan agreement with a national banking institution, bearing an interest rate based upon the 30-, 60- or 90-day London Interbank Offering Rate (LIBOR) plus 0.50% to 1.00%, depending upon the Company's quarterly ratio of funded debt to earnings before interest, taxes, depreciation and amortization. The 90-day LIBOR was 1.59% as of June 30, 2004. The loan was fully funded on January 3, 2001 and as of June 30, 2004 had an outstanding balance of \$45.0 million. This loan is to be repaid in equal quarterly installments of \$3.2 million through December 2007.

To hedge the risk of increasing interest rates from January 2, 2002 through the remaining six years of its seven-year \$90.0 million term loan, the Company entered into an interest rate swap agreement that effectively converted the floating LIBOR-based interest payments to fixed interest rate payments at 4.53%. This agreement did not affect the required 0.50% to 1.00% credit risk spread portion of the term loan. In accordance with SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities", as amended, the Company recorded a liability as of June 30, 2004 for the fair value of the interest rate swap of approximately \$670,000, net of taxes of approximately \$419,000, with the related change in fair value reflected as other comprehensive income. As of December 31, 2003, the Company recorded a liability for the fair value of the interest rate swap of approximately \$1,344,000, net of taxes of approximately \$824,000. The Company has designated and assessed the derivative as a highly effective cash flow hedge, and accordingly, the effect is reflected in other comprehensive income.

In September 2003, the Company established an unsecured revolving credit facility with a national banking institution that provided for available borrowings of up to \$75 million, with a maturity date of October 2008, bearing an interest rate based upon the 30-, 60- or 90-day LIBOR plus 0.625% to 1.625%, depending upon the Company's quarterly ratio of funded debt to earnings before interest, taxes, depreciation, amortization and non-cash stock grant compensation. A commitment fee of 0.175% to 0.375% per annum is assessed on the unused balance. As of June 30, 2004, there was an outstanding balance of \$50.0 million.

Both of these credit agreements require the Company to maintain certain financial ratios and comply with certain other covenants. The Company was in compliance with all such covenants as of June 30, 2004.

Acquisition and other notes payable as of June 30, 2004 were \$7.8 million, which primarily represents debt incurred to former owners of certain agencies or customer accounts acquired by the Company. These notes are payable in monthly, quarterly or annual installments through February 2014, including interest ranging from 1.34% to 9.00%.

Note 7 - Contingencies

The Company is involved in numerous pending or threatened proceedings by or against the Company or one or more of its subsidiaries that arise in the ordinary course of business. The damages that may be claimed in these various proceedings are substantial, including in many instances claims for punitive or extraordinary damages. Some of these claims and lawsuits have been resolved, others are in the process of being resolved, and others are still in the investigation or discovery phase. The Company will continue to respond appropriately to these claims and lawsuits, and to vigorously protect its interests.

Although the ultimate outcome of the matters referred to above cannot be ascertained and liabilities in indeterminate amounts may be imposed on the Company or its subsidiaries, on the basis of present information, availability of insurance coverages and legal advice received, it is the opinion of management that the disposition or ultimate determination of such claims will not have a material adverse effect on the Company's consolidated financial position or results of operations.

Note 8 - Supplemental Disclosures of Cash Flow Information

		ix months June 30
	2004	2003
Cash paid during the period for (in thousands):		
Interest	\$ 1,477	\$ 1,905
Income taxes	\$40,385	\$37,288

The Company's significant non-cash investing and financing activities are as follows (in thousands):

	For the si ended J	
	2004	2003
Unrealized holding (loss) gain on available-for-sale securities, net of tax benefit of \$25 in 2004 and net of tax effect of \$746 in		
2003	\$ (157)	\$1,216
Net gain (loss) on cash-flow hedging derivative, net of tax effect of \$404 for 2004 and net of tax benefit of \$58 for 2003	675	(95)
Notes payable and other liabilities issued or assumed for purchased customer accounts	1,868	2,954
Notes receivable on sale of fixed assets and customer accounts	4,638	776
Common stock issued for acquisitions	6,244	

Note 9 - Comprehensive Income

The components of comprehensive income, net of related tax, are as follows (in thousands):

		For the three months ended June 30,		For the six months ended June 30,	
	2004	2003	2004	2003	
Net income	\$32,153	\$27,935	\$68,501	\$58,471	
Net change in unrealized holding gain (loss) on available-for-sale securities	858	1,456	(157)	1,216	
Net gain (loss) on cash-flow hedging derivative	742	(149)	675	(95)	
Comprehensive income	\$33,753	\$29,242	\$69,019	\$59,592	

Note 10 - Segment Information

The Company's business is divided into four reportable segments: the Retail Division, which provides a broad range of insurance products and services to commercial, governmental, professional and individual customers; the National Programs Division, which is comprised of two units - Professional Programs, which provides professional liability and related package products for certain professionals, delivered through nationwide networks of independent agents, and Special Programs, which markets targeted products and services designated for specific industries, trade groups, governmental entities and market niches; the Brokerage Division, which provides insurance-related services including third-party administration, consulting for the workers' compensation and employee benefit self-insurance markets, and managed healthcare services. The Company conducts all of its operations within the United States of America.

Summarized financial information concerning the Company's reportable segments is shown in the following table. The "Other" column includes any income and expenses not allocated to reportable segments and corporate-related items, including the inter-company interest expense charge to the reporting segment (in thousands).

Six Months Ended June 30, 2004:	Retail	National Programs	Brokerage	Services	Other	Total
Total revenues	\$ 240,978	\$ 47,356	\$ 20,476	\$14,297	\$ 400	\$ 323,507
Investment income	535	41		_	445	1,021
Amortization	7,229	2,732	242	19	78	10,300
Depreciation	2,929	748	245	170	331	4,423
Interest expense	10,305	3,968	430	67	(13,316)	1,454
Income before income taxes	67,772	12,683	7,243	3,439	20,752	111,889
Total assets	787,390	324,545	82,573	14,940	(215,349)	994,099
Capital expenditures	3,071	671	223	396	203	4,564
Six Months Ended June 30, 2003:	Retail	National Programs	Brokerage	Services	Other	Total
Six Months Ended June 30, 2003: Total revenues	Retail \$ 209,913		Brokerage \$ 17,846	Services 	Other \$ 581	Total \$ 282,594
· · · · · · · · · · · · · · · · · · ·		Programs				
Total revenues	\$ 209,913	Programs \$ 40,070			\$ 581	\$ 282,594
Total revenues Investment income	\$ 209,913 23	Programs \$ 40,070 95	\$ 17,846 	\$14,184	\$ 581 657	\$ 282,594 775
Total revenues Investment income Amortization	\$ 209,913 23 6,252	Programs \$ 40,070 95 2,251	\$ 17,846 	\$ 14,184 	\$ 581 657 78	\$ 282,594 775 8,753
Total revenues Investment income Amortization Depreciation	\$ 209,913 23 6,252 2,798	Programs \$ 40,070 95 2,251 572	\$ 17,846 	\$14,184 19 	\$ 581 657 78 183	\$ 282,594 775 8,753 3,946
Total revenues Investment income Amortization Depreciation Interest expense	\$ 209,913 23 6,252 2,798 9,024	Programs \$ 40,070 95 2,251 572 2,964	\$ 17,846 153 149 375	\$14,184 	\$ 581 657 78 183 (10,493)	\$ 282,594 775 8,753 3,946 1,953

Note 11 – Subsequent Events

On July 15, 2004 the Company completed the first funding of a total issuance of up to \$200 million of unsecured senior notes (the "Notes") through a private placement. The \$200 million is divided into two series: Series A, for \$100 million due in 2011 and bearing interest at 5.57% per year; and Series B, for \$100 million due in 2014 and bearing interest at 6.08% per year. The closing on the Series B Notes occurred on July 15, 2004. The closing on the Series A Notes is expected to occur on September 15, 2004. The Company anticipates using the proceeds from the sales of the Notes for general corporate purposes, including repayment of existing senior debt and acquisitions.

Additionally, from July 1, 2004 through July 31, 2004, the Company acquired the assets and certain liabilities of six general insurance agencies. The aggregate purchase price of these acquisitions was \$26.3 million, including \$26.2 million of cash payments, and the assumption of \$0.1 million of liabilities.

ITEM 2: MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS (MD&A)

THE FOLLOWING DISCUSSION UPDATES THE MD&A CONTAINED IN THE COMPANY'S 2003 ANNUAL REPORT ON FORM 10-K, AND THE TWO DISCUSSIONS SHOULD BE READ TOGETHER.

<u>General</u>

We are a general insurance agency that commenced business in 1939 and are headquartered in Daytona Beach and Tampa, Florida. We market and sell to our customers insurance products and services, primarily in the property, casualty and the employee benefits area. As an agent and broker, we do not assume underwriting risks. Instead, we provide our customers with quality insurance contracts, as well as other targeted, customized risk management products.

We are compensated for our services primarily by commissions paid by insurance underwriters and fees paid by customers for certain services. The commission income that we receive is usually a percentage of the premium paid by the insured. Commission rates generally depend upon the type of insurance, the particular insurance company and the nature of the services provided by us. In some cases, a commission is shared with other agents or brokers who have acted jointly with us in a transaction. We may also receive "contingent commissions" which are revenue-sharing commissions paid by insurance underwriters based primarily on the overall profitability of the aggregate business written with that underwriter, and/or additional factors such as retention ratios and overall volume of business that we place with such insurance underwriters during the prior year.

The Insurance Market

Premium rates are established by insurance underwriters based upon many factors, including reinsurance rates, none of which we control. Beginning in 1986 and continuing through 1999, commission revenues were adversely influenced by a consistent decline in premium rates resulting from intense competition among property and casualty underwriters for market share. Among other factors, this condition of a prevailing decline in premium rates, commonly referred to as a "soft market," generally resulted in flat to reduced commissions on renewal business. The effect of this softness in rates was somewhat offset by our substantial merger and acquisition activity and net new business production. As a result of increasing loss ratios, (the comparison of incurred losses plus loss adjustment expenses against earned premium) of insurance underwriters through 1999, there was a general increase in premium rates commencing in the first quarter of 2000 and continuing into the first half of 2003. Starting in the second half of 2003, as underwriting companies began to experience improved loss ratios, they have again become more competitive on selected risks, resulting in a moderation of premium rate increases, and in some cases reductions in premium rates. Based on what the underwriters are saying and the results of our own renewal and new business activity, we anticipate that this market softening will continue through the balance of 2004 and into 2005.

Critical Accounting Policies

The more critical accounting and reporting policies include our accounting for revenue recognition, business acquisitions and purchase price allocations, intangible assets impairments, reserves for litigation and derivative interests. In particular, the accounting for these areas requires significant judgments to be made by management. Different assumptions in the application of these policies could result in material changes in our consolidated financial position or consolidated results of operations. Refer to Note 1 in the "Notes to Consolidated Financial Statements" in our 2003 Annual Report on Form 10-K on file with the Securities and Exchange Commission for details regarding all of our critical and significant accounting policies.

Results of Operations

The condensed consolidated financial information relating to the three and six month periods ended June 30, 2004 is as follows:

	For the three months ended June 30,			For the six months ended June 30,			
	2004	2003	% Change	2004	2003	% Change	
REVENUES							
Commissions and fees	\$ 156,749	\$ 137,257	14.2%	\$ 321,063	\$ 281,509	14.1%	
Investment income	333	442	(24.7)%	1,021	775	31.7%	
Other income, net	860	159	440.9%	1,423	310	359.0%	
Total revenues	157,942	137,858	14.6%	323,507	282,594	14.5%	
EXPENSES							
Employee compensation and benefits	76,270	66,092	15.4%	152,552	134,333	13.6%	
Non-cash stock grant compensation	665	632	5.2%	1,510	1,449	4.2%	
Other operating expenses	19,983	19,229	3.9%	41,379	38,635	7.1%	
Amortization	5,483	4,416	24.2%	10,300	8,753	17.7%	
Depreciation	2,269	2,019	12.4%	4,423	3,946	12.1%	
Interest	743	946	(21.5)%	1,454	1,953	(25.6)%	
Total expenses	105,413	93,334	12.9%	211,618	189,069	11.9%	
Income before income taxes	52,529	44,524	18.0%	111,889	93,525	19.6%	
Income taxes	20,376	16,589	22.8%	43,388	35,054	23.8%	
NET INCOME	\$ 32,153	\$ 27,935	15.1%	\$ 68,501	\$ 58,471	17.2%	

Net Income. Net income for the second quarter of 2004 was \$32.2 million, or \$0.46 per diluted share, compared with net income in the second quarter of 2003 of \$27.9 million, or \$0.41 per diluted share, a 12.2% increase on a per-share basis. Net income for the six months ended June 30, 2004 was \$68.5 million or \$0.99 diluted share, compared with net income for the comparable period in 2003 of \$58.5 million, or \$0.85 per diluted share, a 16.5% increase on a per share basis.

Commissions & Fees. Commissions and fees for the second quarter of 2004 increased \$19.5 million, or 14.2%, over the same period in 2003. Contingent commissions (revenue-sharing commissions paid by insurance underwriters based primarily on the overall profitability of the aggregate business written with that underwriter, and/or additional factors such as retention ratios and overall volume of business that we place with such insurance underwriters during the prior year) for the quarter decreased \$6.7 million from the second quarter of 2003, primarily due to the fact that more of this revenue was received in the first quarter of 2004 than in the same period in 2003. For the six months ended June 30, 2004, contingent commissions increased \$0.9 million over the comparable period of 2003. Total core

commissions and fees are our total commissions and fees, less (i) contingent commissions and (ii) divested business (commissions and fees generated from offices, books of business or niches sold or terminated. Of the increase in commissions and fees revenue for the second quarter of 2004, approximately \$23.1 million represents core commissions and fees from agencies acquired since the second quarter of 2003 and \$6.3 million represents net new business production. Commissions and fees for the six months ended June 30, 2004 increased \$39.6 million, or 14.1%, over the same period in 2003. Approximately \$34.9 million of this increase represents core commissions and fees from agencies acquired since the comparable period in 2003.

Investment Income. Investment income for the three months ended June 30, 2004 decreased \$0.1 million, or 24.7%, from the same period in 2003. This decrease in investment income was primarily due to lower available investment cash balances along with lower investment yields. Investment income for the six months ended June 30, 2004 increased \$0.2 million, or 31.7%, over the same period in 2003. This increase is primarily related to the sale of investments in various equity securities and partnership interests.

Other Income. Other income for the three months ended June 30, 2004 increased \$0.7 million, or 440.9%, over the same period in 2003. Other income for the six months ended June 30, 2004, increased \$1.1 million, or 359.0%, over the same period in 2003. These increases are primarily due to gains on the sale of various offices and books of business in 2004.

Employee Compensation and Benefits. Employee compensation and benefits for the second quarter of 2004 increased \$10.2 million, or 15.4%, over the same period in 2003. Employee compensation and benefits for the six months ended June 30, 2004 increased \$18.2 million, or 13.6%, over the same period in 2003. These increases are primarily related to the addition of new employees from acquisitions completed since July 1, 2003 and increased compensation that resulted from higher commissions and fees revenue. Employee compensation and benefits as a percentage of total revenue increased to 48.3% for the second quarter of 2004, from 47.9% for the second quarter of 2003. This increased ratio was primarily the result of the decrease in contingent commissions received in the second quarter of 2004. For the six months ended June 30, 2004, employee compensation and benefits as a percentage of total revenue decreased to 47.2%, compared with 47.5% for the same period in 2003. The improved ratio for the six month period was the result of the continued assimilation of the acquisitions completed in 2003 and 2004 into our standard compensation program

Non-Cash Stock Grant Compensation. Non-cash stock grant compensation expense represents the expense required to be recorded under Accounting Principles Board Opinion ("APB") No. 25, "Accounting for Stock Issued to Employees", relating to our stock performance plan. The annual cost of this stock performance plan increases only when our average stock price over a 20 trading-day period increases by increments of 20% or more over the price at the time of the original grant, or when more shares are granted and the aforementioned average stock price increases. Non-cash stock grant compensation expense for the second quarter of 2004 increased less then \$0.1 million, or 5.2%, over the same period in 2003. Non-cash stock grant compensation for the six months ended June 30, 2004 increased \$0.1 million, or 4.2%, over the same period in 2003. This increase is primarily the result of more shares being outstanding during the second quarter of 2004 than were outstanding during the second quarter of 2003.

Other Operating Expenses. Other operating expenses for the second quarter of 2004 increased \$0.8 million, or 3.9%, over the same period in 2003. For the six months ended June 30, 2004, other operating expenses increased \$2.7 million, or 7.1%, over the same period in 2003. This was primarily the result of acquisitions completed since the second quarter of 2003 that had no comparable results in the same period of 2003. Other operating expenses as a percentage of revenues for the second quarter of 2004 decreased to 12.7%, from 13.9%. For the six months ended June 30, 2004, other operating expenses as a percentage of revenue decreased to 12.8%, compared with 13.7% for the same period in 2003. The improved ratios are the result of improved operating efficiencies.

Amortization. Amortization expense for the second quarter of 2004 increased \$1.1 million, or 24.2%, over the second quarter of 2003. For the six months ended June 30, 2004, amortization expense increased \$1.5 million, or 17.7%, over the same period in 2003. These increases are primarily due to acquisitions completed since July 1, 2003.

Depreciation. Depreciation expense for the second quarter of 2004 increased \$0.3 million, or 12.4%, over the second quarter of 2003. For the six months ended June 30, 2004, depreciation expense increased \$0.5 million, or 12.1%, over the same period in 2003. These increases are due to capital expenditures and fixed assets acquired in agency acquisitions completed since July 1, 2003.

Interest Expense. Interest expense for the second quarter of 2004 decreased \$0.2 million, or 21.5%, from the same period in 2003. For the six months ended June 30, 2004, interest expense decreased \$0.5 million, or 25.6%, from the same period in 2003. These decreases are the result of lower outstanding debt balances.

Segment Information

As discussed in Note 10 of the Notes to Condensed Consolidated Financial Statements, we operate in four reportable segments: the Retail, National Programs, Brokerage and Services Division.

Retail

The Retail Division is our insurance agency business, which provides a broad range of insurance products and services directly to commercial, governmental, professional and individual clients. The table below presents the Retail Division's total revenues, expenses and income before income taxes during the three and six months ended June 30, 2004 and 2003.

		For the three months ended June 30,			For the six months ended June 30,		
	2004	2003	% Change	2004	2003	% Change	
Total revenues	\$ 115,952	\$ 103,019	12.6%	\$ 240,978	\$ 209,913	14.8%	
Total expenses	88,195	76,348	15.5%	173,206	152,558	13.5%	
Income before income taxes	\$ 27,757	\$ 26,671	4.1%	\$ 67,772	\$ 57,355	18.2%	

The Retail Division's total revenues during the three months ended June 30, 2004 increased 12.6%, or \$12.9 million, to \$116.0 million. Contingent commissions for the quarter decreased \$4.8 million from the second quarter of 2003, primarily due to the fact that more of this revenue was received in the first quarter of 2004 than in the same period in 2003. Of the increase in revenues, approximately \$16.4 million related to the core commissions and fees from acquisitions that had no comparable revenues in the same period of 2003. The remaining increase is primarily due to net new business growth in core commissions and fees of \$3.6 million. Income before income taxes for the three months ended June 30, 2004 increased 4.1%, or \$1.1 million, to \$27.8 million. This increase is primarily due to the earnings from acquisitions, net new business, but was offset by lower contingent commissions.

The Retail Division's total revenues during the six months ended June 30, 2004 increased 14.8%, or \$31.1 million, to \$241.0 million over the same period in 2003. Contingent commissions for the first six months of the year increased \$2.5 million over the comparable period of 2003. Of the increase in revenues, approximately \$25.1 million related to the core commissions and fees from acquisitions that had no comparable revenues in the same period of 2003. The remaining increase is primarily due to net new business growth in core commissions and fees of \$6.1 million. Income before income taxes for the six months ended June 30, 2004 increased 18.2%, or \$10.4 million, to \$67.8 million over the same period in 2003. This increase is primarily due to earnings from acquisitions, net new business, and an increase in contingent commissions.

National Programs

The National Programs Division is comprised of two units: Professional Programs, which provides professional liability and related package products for certain professionals delivered through nationwide networks of independent agents; and Special Programs, which markets targeted products and services designated for specific industries, trade groups, governmental entities and market niches. The table below presents the National Programs Division's total revenues, expenses and income before income taxes during the three and six months ended June 30, 2004 and 2003.

	F	For the three months ended June 30,			For the six months ended June 30,		
	2004	2003	% Change	2004	2003	% Change	
Total revenues	\$25,156	\$19,128	31.5%	\$47,356	\$40,070	18.2%	
Total expenses	18,992	13,363	42.1%	34,673	26,571	30.5%	
Income before income taxes	\$ 6,164	\$ 5,765	6.9%	\$12,683	\$13,499	(6.0)%	

Total revenues for National Programs for the three months ended June 30, 2004 increased 31.5%, or \$6.0 million, to \$25.2 million. This increase consists of \$5.8 million in core commissions and fees from acquisitions that had no comparable revenues in the same period of 2003. The remaining increase is primarily due to net new business growth offset by a slight decrease in contingent commissions. Income before income taxes for the three months ended June 30, 2004 increased 6.9%, or \$0.4 million, to \$6.2 million, from the same period in 2003, which is primarily due to net new business growth and earnings from acquisitions completed since the second quarter of 2003.

Total revenues for National Programs for the six months ended June 30, 2004 increased 18.2%, or \$7.3 million, to \$47.4 million over the same period in 2003. This increase consists of \$8.1 million in core commissions and fees from acquisitions that had no comparable revenues in the same period of 2003. The remaining increase is primarily due to net new business growth, offset by a \$0.4 million decrease in contingent commissions. Income before income taxes for the six months ended June 30, 2004 decreased 6.0%, or \$0.8 million, to \$12.7 million, which is primarily due to \$1.5 million in additional amortization and interest expense allocated as a result of new acquisitions.

Brokerage

The Brokerage Division markets and sells excess and surplus commercial insurance and reinsurance, primarily through independent agents and brokers. The table below presents the Brokerage Division's total revenues, expenses and income before income taxes during the three and six months ended June 30, 2004 and 2003.

	Fe	For the three months ended June 30,			For the six months ended June 30,		
	2004	2003	% Change	2004	2003	% Change	
Total revenues	\$ 8,393	\$8,187	2.5%	\$20,476	\$17,846	14.7%	
Total expenses	6,236	4,899	27.3%	13,233	10,031	31.9%	
Income before income taxes	\$ 2,157	\$3,288	(34.4)%	\$ 7,243	\$ 7,815	(7.3)%	

The Brokerage Division's total revenues for the three months ended June 30, 2004 increased 2.5%, or \$0.2 million, to \$8.4 million over the same period in 2003. Contingent commissions for the quarter decreased \$1.7 million from the second quarter of 2003, primarily due to the fact that more of this revenue was received in the first quarter of 2004 than in the same period in 2003. Of the increase in revenues, approximately \$0.9 million related to core commissions and fees from acquisitions that had no comparable revenues in the same period of 2003. The remaining increase is primarily due to net new business growth in core commissions and fees of \$1.0 million. Income before income taxes for the three months ended June 30, 2004 decreased 34.4%, or \$1.1 million, to \$2.2 million from the same period in 2003, primarily due to the decrease in contingent commissions received in the second quarter of 2004 compared with the corresponding period in 2003.

The Brokerage Division's total revenues for the six months ended June 30, 2004 increased 14.7%, or \$2.6 million, to \$20.5 million over the same period in 2003. Contingent commissions for the first six months of 2004 decreased \$0.9 million from the same period of 2003. Of the increase in revenues, approximately \$1.7 million related to core commissions and fees from acquisitions that had no comparable revenues in the same period of

2003. The remaining increase is primarily due to net new business growth in core commissions and fees of \$1.9 million. Income before income taxes for the six months ended June 30, 2004 decreased 7.3% or \$0.6 million, to \$7.2 million from the same period in 2003, which is primarily due to the decrease in contingent commissions received in 2004 compared with the corresponding period of 2003.

Services

The Services Division provides insurance-related services, including third-party administration, consulting for the workers' compensation and employee benefit self-insurance markets and managed healthcare services. Unlike our other segments, the majority of the Services Division's revenues are fees, which are not significantly affected by fluctuations in general insurance premiums. The table below presents the Services Division's total revenues, expenses and income before income taxes during the three and six months ended June 30, 2004 and 2003.

	F	For the three months ended June 30,			For the six months ended June 30,		
	2004	2003	% Change	2004	2003	% Change	
Total revenues	\$ 7,833	\$ 7,115	10.1%	\$14,297	\$14,184	0.8%	
Total expenses	5,630	5,905	(4.7)%	10,858	11,874	(8.6)%	
Income before income taxes	\$ 2,203	\$1,210	82.1%	\$ 3,439	\$ 2,310	48.9%	

The Service Division's total revenues for the three months ended June 30, 2004 increased 10.1%, or \$0.7 million, to \$7.8 million from the same period in 2003. On June 30, 2004, we sold our medical services operation in Louisiana, resulting in a \$1.0 million gain. The Louisiana medical services operation had estimated annual revenues of approximately \$3.2 million and therefore future quarterly revenues for the Services Division will be less without this operation. Income before income taxes for the three months ended June 30, 2004 increased 82.1%, or \$1.0 million, to \$2.2 million from the same period in 2003, primarily as a result of the gain on the sale of the medical services operation in Louisiana.

The Service Division's total revenues for the six months ended June 30, 2004 increased 0.8%, or \$0.1 million, to \$14.3 million from the same period in 2003. Income before income taxes for the six months ended June 30, 2004 increased 48.9%, or \$1.1 million, to \$3.4 million from the same period in 2003. This increase was primarily due to the gain on the sale of the medical services operation in Louisiana.

Liquidity and Capital Resources

Our cash and cash equivalents at June 30, 2004 totaled \$36.4 million, a decrease of \$20.5 million from the \$56.9 million balance at December 31, 2003. For the six-month period ended June 30, 2004, \$92.4 million of cash was provided from operating activities and \$50.0 million was borrowed from our revolving credit facility. Also during this period, \$143.6 million was used to acquire other agencies and books of business (customer accounts), \$4.5 million was used for additions to fixed assets, \$8.9 million was used for payments on long-term debt, and \$9.6 million was used for payment of dividends.

As of June 30, 2004, our contractual cash obligations were as follows (in thousands):

		Payments Due by Period				
Contractual Cash Obligations	Total	Less than 1 Year	1-3 Years	4-5 Years	After 5 Years	
Long-term debt	\$ 102,741	\$ 68,449	\$ 27,254	\$ 6,746	\$ 292	
Capital lease obligations	27	6	15	6	_	
Other long term liabilities	9,063	6,006	1,153	651	1,253	
Operating leases	70,968	17,355	25,609	15,757	12,247	
Maximum future acquisition contingent payments	100,202	38,237	61,094	871		
Total contractual cash obligations	\$ 283,001	\$ 130,053	\$ 115,125	\$24,031	\$13,792	

In January 2001, we entered into a \$90 million unsecured seven-year term loan agreement with a national banking institution, bearing an interest rate based upon the 30-, 60- or 90-day London Interbank Offering Rate (LIBOR) plus 0.50% to 1.00%, depending upon our quarterly ratio of funded debt to earnings before interest, taxes, depreciation and amortization. The 90-day LIBOR was 1.59% as of June 30, 2004. The loan was fully funded on January 3, 2001 and as of June 30, 2004 had an outstanding balance of \$45.0 million. This loan is to be repaid in equal quarterly installments of \$3.2 million through December 2007.

To hedge the risk of increasing interest rates from January 2, 2002 through the remaining six years of its seven-year \$90 million term loan, we entered into an interest rate swap agreement that effectively converted the floating LIBOR-based interest payments to fixed interest payments at an annual rate of 4.53%. This agreement did not impact or change the required 0.50% to 1.00% credit risk spread portion of the term loan. In accordance with SFAS No. 133, as amended, we recorded a liability as of June 30, 2004 for the fair value of the interest rate swap at June 30, 2004 of approximately \$670,000, net of taxes of approximately \$419,000. We have designated and assessed the derivative as a highly effective cash flow hedge, and accordingly, the effect is reflected in other comprehensive income.

In September 2003, we established an unsecured revolving credit facility with a national banking institution that provided for available borrowings of up to \$75 million, with a maturity date of October 2008, bearing an interest rate based upon the 30-, 60- or 90-day LIBOR plus 0.625% to 1.625%, depending upon our quarterly ratio of funded debt to earnings before interest, taxes, depreciation, amortization and non-cash stock grant compensation. A commitment fee of 0.175% to 0.375% per annum is assessed on the unused balance. As of June 30, 2004, there was an outstanding balance of \$50.0 million.

We (including our subsidiaries) have never incurred off-balance sheet obligations through the use of, or investment in, off-balance sheet derivative financial instruments or structured finance or special purpose entities organized as corporations, partnerships or limited liability companies, or trusts.

We believe that our existing cash, cash equivalents, short-term investments portfolio, funds generated from operations, and available credit facility borrowings are sufficient to satisfy our normal short-term financial needs. However, to continue to fund potential future acquisitions, we have agreed to borrow \$200 million by issuing senior unsecured notes (the "Notes") with maturities ranging from seven to ten years through a private debt offering. The \$200 million is divided into two series: Series A, for \$100 million due in 2011 and bearing interest at 5.57% per year; and Series B, for \$100 million due in 2014 and bearing interest at 6.08% per year. The closing of the Series B Notes occurred on July 15, 2004. The closing of the Series A Notes is expected to occur on September 15, 2004.

Recent Industry Development

Based on press releases issued by certain insurance brokerage companies, we understand that the Office of the Attorney General of New York has served subpoenas on certain insurance brokerage companies seeking information relating to certain compensation agreements between those insurance brokers and insurance underwriters. As of the date of this report, we have not received a subpoena from the Office of the Attorney General of New York. However, in March and April 2004, one of our New York subsidiaries received an initial inquiry letter and one follow up letter from the State of New York Insurance Department requesting information relating to "placement service agreements" maintained by that New York subsidiary. We have responded to such requests.

As previously disclosed in our public filings, we have contingent commission agreements with certain insurance underwriters. Contingent commissions are revenue-sharing commissions paid by insurance underwriters based primarily on the overall profitability of the aggregate business written with that underwriter, and/or additional factors such as retention ratios and overall volume of business that we place with the underwriter. To a lesser extent, we have some override commission agreements, which allow for commissions to be paid by insurance underwriters in excess of the standard commission rate in specific lines of business, such as group health business. While it is not possible to predict the outcome of these inquires, any decrease in these contingent commissions and override commissions may have a negative effect on our results of operations.

Disclosure Regarding Forward-Looking Statements

We make "forward-looking statements" within the "safe harbor" provision of the Private Securities Litigation Reform Act of 1995 throughout this report and in the documents we incorporate by reference into this report. You can identify these statements by forward-looking words such as "may," "will," "expect," "anticipate," "believe," "estimate," "plan" and "continue" or similar words. We have based these statements on our current expectations about future events. Although we believe that our expectations reflected in or suggested by our forward-looking statements are reasonable, our actual results may differ materially from what we currently expect. Important factors which could cause our actual results to differ materially from the forward-looking statements in this report include:

- material adverse changes in economic conditions in the markets we serve;
- future regulatory actions and conditions in the states in which we conduct our business;
- competition from others in the insurance agency and brokerage business;
- the integration of our operations with those of businesses or assets we have acquired or may acquire in the future and the failure to realize the expected benefits of such integration; and
- other risks and uncertainties as may be detailed from time to time in our public announcements and Securities and Exchange Commission filings.

You should carefully read this report completely and with the understanding that our actual future results may be materially different from what we expect. All forward-looking statements attributable to us are expressly qualified by these cautionary statements.

We do not undertake any obligation to publicly update or revise any forward-looking statements.

ITEM 3: QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Market risk is the potential loss arising from adverse changes in market rates and prices, such as interest rates and equity prices. We are exposed to market risk through our investments, revolving credit line and term loan agreements.

Our invested assets are held as cash and cash equivalents, restricted cash, available-for-sale marketable equity securities, non-marketable equity securities and certificates of deposit. These investments are subject to interest rate risk and equity price risk. The fair values of our cash and cash equivalents, restricted cash, and certificates of deposit at June 30, 2004 and December 31, 2003 approximated their respective carrying values due to their short-term duration and therefore such market risk is not considered to be material.

We do not actively invest or trade in equity securities. In addition, we generally dispose of any significant equity securities received in conjunction with an acquisition shortly after the acquisition date. However, we have no current intention to add to or dispose of any of the 559,970 common stock shares of Rock-Tenn Company, a publicly-held New York Stock Exchange-listed company, which we have owned for over 10 years. The investment in Rock-Tenn Company accounted for 88% and 86% of the total value of available-for-sale marketable equity securities, non-marketable equity securities and certificates of deposit as of June 30, 2004 and December 31, 2003, respectively. Rock-Tenn Company's closing stock price at June 30, 2004 and December 31, 2003 was \$16.95 and \$17.26, respectively. Our exposure to equity price risk is primarily related to the Rock-Tenn Company investment. As of June 30, 2004, the value of the Rock-Tenn Company investment was \$9,491,000.

To hedge the risk of increasing interest rates from January 2, 2002 through the remaining six years of our seven-year \$90 million term loan, on December 5, 2001 we entered into an interest rate swap agreement that effectively converted the floating rate LIBOR-based interest payments to fixed interest rate payments at 4.53%. We do not otherwise enter into derivatives, swaps or other similar financial instruments for trading or speculative purposes.

At June 30, 2004, the interest rate swap agreement was as follows (in thousands, except percentages):

	 ntractual/ onal Amount	Fair Value	Weighted Average Pay Rates	Weighted Average Received Rates
Interest rate swap agreement	\$ 45,000	\$ (1,089)	4.53%	1.11%

ITEM 4: CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

We carried out an evaluation required by Rules 13a-15 and 15d-15 under the Exchange Act (the "Evaluation"), under the supervision and with the participation of our Chief Executive Officer ("CEO") and Chief Financial Officer ("CFO"), of the effectiveness of our disclosure controls and procedures as defined in Rule 13a-15 and 15d-15 under the Exchange Act ("Disclosure Controls"). Although we believe that our pre-existing Disclosure Controls, including our internal controls, were adequate to enable us to comply with our disclosure obligations, as a result of such Evaluation, we implemented minor changes, primarily to formalize, document and update the procedures already in place. Based on the Evaluation, our CEO and CFO concluded that, subject to the limitations noted herein, our Disclosure Controls are effective in timely alerting them to material information required to be included in our periodic SEC reports.

Changes in Internal Controls

There has not been any change in our internal control over financial reporting identified in connection with the Evaluation that occurred during the quarter ended June 30, 2004 that has materially affected, or is reasonably likely to materially affect, those controls.

Limitations on the Effectiveness of Controls

Our management, including our CEO and CFO, does not expect that our Disclosure Controls and internal controls will prevent all error and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Further, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within the Company have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty and that breakdowns can occur because of simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the controls.

The design of any system of controls also is based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions; over time, controls may become inadequate because of changes in conditions, or the degree of compliance with the policies or procedures may deteriorate. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected.

CEO and CFO Certifications

Exhibits 31.1 and 31.2 are the Certifications of the CEO and the CFO, respectively. The Certifications are required in accord with Section 302 of the Sarbanes-Oxley Act of 2002 (the "Section 302 Certifications"). This Item 4, of this report, which you are currently reading, is the information concerning the Evaluation referred to in the Section 302 Certifications and this information should be read in conjunction with the Section 302 Certifications for a more complete understanding of the topics presented.

PART II - OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

We are involved in numerous pending or threatened proceedings by or against us or one or more of our subsidiaries that arise in the ordinary course of business. The damages that may be claimed in these various proceedings are substantial, including in many instances claims for punitive or extraordinary damages. Some of these claims and lawsuits have been resolved, others are in the process of being resolved, and others are still in the investigation or discovery phase. We will continue to respond appropriately to these claims and lawsuits, and to vigorously protect our interests.

Although the ultimate outcome of all matters referred to above cannot be ascertained and liabilities in indeterminate amounts may be imposed on us or our subsidiaries, on the basis of present information, availability of insurance coverages and legal advice received, it is the opinion of management that the disposition or ultimate determination of such claims will not have a material adverse effect on our consolidated financial position.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

The Company's Annual Meeting of Shareholders was held on April 22, 2004. At the meeting, one matter was submitted to a vote of security holders.

1. The election of nine directors

The number of votes cast for, withheld or abstaining with respect to the election of each of the directors is set forth below:

	For	Abstain/ Withheld
J. Hyatt Brown	63,096,879	395,776
Samuel P. Bell, III	62,864,261	628,394
Hugh M. Brown	63,215,494	277,161
Bradley Currey, Jr.	62,868,623	624,032
Jim W. Henderson	63,068,290	424,375
Theodore J. Hoepner	63,090,213	402,442
David H. Hughes	62,869,570	623,085
John R. Riedman	62,726,213	766,442
Jan E. Smith	62,870,115	622,540

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) EXHIBITS

- Exhibit 3.1Articles of Amendment to Articles of Incorporation (adopted April 24, 2003) (incorporated by reference to Exhibit
3a to Form 10-Q for the quarter ended March 31, 2003), and Amended and Restated Articles of Incorporation
(incorporated by reference to Exhibit 3a to Form 10-Q for the quarter ended March 31, 1999).
- Exhibit 3.2 Bylaws (incorporated by reference to Exhibit 3b to Form 10-K for the year ended December 31, 2002).
- Exhibit 4.1 Note Purchase Agreement, dated as of July 15, 2004, among the Company and the listed Purchasers of the 5.57% Series A Senior Notes due September 15, 2011 and 6.08% Series B Senior Notes due July 15, 2014.
- Exhibit 4.2 First Amendment to Amended and Restated Revolving and Term Loan Agreement dated and effective July 15, 2004, by and between Brown & Brown, Inc. and SunTrust Bank.

Exhibit 4.3	Second Amendment to Revolving Loan Agreement dated and effective July 15, 2004, by and between Brown & Brown, Inc. and SunTrust Bank.
Exhibit 4.4	Rights Agreement, dated as of July 30, 1999, between the Company and First Union National Bank, as Rights Agent (incorporated by reference to Exhibit 4.1 to Form 8-K filed on August 2, 1999).
Exhibit 31.1	Section 302 Certification by the Chief Executive Officer of the Company.
Exhibit 31.2	Section 302 Certification by the Chief Financial Officer of the Company.
Exhibit 32.1	Section 1350 Certification by the Chief Executive Officer of the Company.
Exhibit 32.2	Section 1350 Certification by the Chief Financial Officer of the Company.

(b) REPORTS ON FORM 8-K

The Company filed a current report on Form 8-K on April 14, 2004. This current report reported Item 12, which announced that the Company issued a press release on April 14, 2004, relating to the Company's earnings for the first quarter of fiscal year 2004.

The Company filed a current report on Form 8-K on April 23, 2004. This current report reported (a) Item 5, which announced that the Company issued a press release on April 23, 2004, relating to the Company's acquisition of Proctor Homer Warren, Inc. d/b/a Proctor Financial Insurance, and (b) Item 7, (the "Press Release") which attached the Press Release as Exhibit 99 thereto.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

BROWN & BROWN, INC.

/S/ CORY T. WALKER

Date: August 9, 2004

Cory T. Walker Sr. Vice President, Chief Financial Officer and Treasurer (duly authorized officer, principal financial officer and principal accounting officer)

BROWN & BROWN, INC.

\$200,000,000

5.57% Series A Senior Notes due September 15, 2011 6.08% Series B Senior Notes due July 15, 2014

NOTE PURCHASE AGREEMENT

Dated as of July 15, 2004

- 1. AUTHORIZATION OF NOTES.
- 2. SALE AND PURCHASE OF NOTES.
- 3. CLOSINGS.

4.

- 3.1. First Closing.
- 3.2. Second Closing.
- 3.3.
- Failure of the Company to Deliver. CONDITIONS TO CLOSINGS.
- - 4.1. Representations and Warranties.
 - Performance; No Default. 4.2.
 - 4.3. Compliance Certificates.
 - 4.4. Opinions of Counsel.
 - Purchase Permitted By Applicable Law, etc. 4.5.
 - Sale of Other Notes. 4.6.
 - Payment of Special Counsel Fees. 4.7.
 - Private Placement Number. 4.8.
 - 4.9. Changes in Corporate Structure.
 - 4.10. Subsidiary Guaranty.
 - Amendments to Bank Credit Agreement. 4.11.
 - Offeree Letter. 4.12.
 - 4.13. Wire Instructions.
 - Proceedings and Documents. 4.14.

REPRESENTATIONS AND WARRANTIES OF THE COMPANY. 5.

- Organization; Power and Authority. 5.1.
- 5.2. Authorization.
- Disclosure. 5.3.
- Organization and Ownership of Shares of Subsidiaries. 5.4.
- Financial Statements. 5.5.
- 5.6. Compliance with Laws, Other Instruments, etc.
- Governmental Authorizations, etc. 5.7.
- 5.8. Litigation; Observance of Statutes and Orders.
- 5.9. Taxes.
- 5.10. Title to Property; Leases.

	5.11.	Licenses, Permits, etc.	9
	5.12.	Compliance with ERISA.	9
	5.13.	Private Offering by the Company.	10
	5.14.	Use of Proceeds; Margin Regulations.	10
	5.15.	Existing Indebtedness.	11
	5.16.	Foreign Assets Control Regulations, etc.	11
	5.17.	Status under Certain Statutes.	11
6.	REPRES	SENTATIONS OF THE PURCHASER.	12
	6.1.	Purchase for Investment.	12
	6.2.	Source of Funds.	12
7.	INFORM	ATION AS TO COMPANY.	13
	7.1.	Financial and Business Information.	13
	7.2.	Officer's Certificate.	16
	7.3.	Inspection.	16
8.		MENT OF THE NOTES.	17
	8.1.	Required Prepayments.	17
	8.2.	Optional Prepayments with Make-Whole Amount.	17
	8.3.	Prepayment of Notes Upon Change in Control.	17
	8.4.	Offer to Prepay upon the Sale of Certain Assets.	18
	8.5.	Allocation of Partial Prepayments.	19
	8.6.	Maturity; Surrender, etc.	20
	8.7.	Purchase of Notes.	20
	8.8.	Make-Whole Amount.	20
9.		IATIVE COVENANTS.	22
	9.1.	Compliance with Law.	22
	9.2.	Insurance.	22
	9.3.	Maintenance of Properties.	22
	9.4.	Payment of Taxes.	22
	9.5.	Corporate Existence, etc.	23
	9.6.	Subsidiary Guarantors.	23
10.		IVE COVENANTS.	24
	10.1.	Transactions with Affiliates.	24
	10.2.	Merger, Consolidation, etc.	24

ii

		le of Assets.	25
		nitation on Liens.	26
		verage Ratio.	28
		xed Charge Coverage Ratio.	28
		bsidiary Debt.	28
		strictions on Dividends of Subsidiaries, etc.	29
		nitation on Priority Debt.	29
	10.10. No	Limitation on Prepayments or Amendments to Certain Financing Documents.	29
	10.11. Lir	ne of Business.	29
11.	EVENTS OF	DEFAULT.	29
12.	REMEDIES C	DN DEFAULT, ETC.	32
	12.1. Ac	recleration.	32
	12.2. Ot	her Remedies.	32
	12.3. Re	scission.	32
	12.4. No	Waivers or Election of Remedies, Expenses, etc.	33
13.	REGISTRATION; EXCHANGE; SUBSTITUTION OF NOTES.		
	13.1. Re	gistration of Notes.	33
	13.2. Tra	ansfer and Exchange of Notes.	33
	13.3. Re	placement of Notes.	34
14.	PAYMENTS ON NOTES.		
	14.1. Pla	ace of Payment.	34
	14.2. Ho	ome Office Payment.	34
15.	EXPENSES, H	ETC.	35
	15.1. Tra	ansaction Expenses.	35
		rvival.	35
16.	SURVIVAL O	F REPRESENTATIONS AND WARRANTIES; ENTIRE AGREEMENT.	35
17.	AMENDMENT AND WAIVER.		
	17.1. Re	quirements.	36
	17.2. So	licitation of Holders of Notes.	36
	17.3. Bin	nding Effect, etc.	36
		ites held by Company, etc.	37
18.	NOTICES.	5 1 5/	37

٠	٠	٠	
1	1	1	
	-	×	

20. 21. 22.	CONFID SUBSTIT MISCEL 22.1. 22.2. 22.3. 22.4.	DUCTION OF DOCUMENTS. ENTIAL INFORMATION. FUTION OF PURCHASER. LANEOUS. Successors and Assigns. Payments Due on Non-Business Days. Severability. Construction.	37 38 39 39 39 39 39 39 40
	22.4.	Construction.	40
	22.5. 22.6.	Counterparts. Governing Law.	40 40

SCHEDULES AND EXHIBITS

SCHEDULE A	—	Information Relating To Purchasers
SCHEDULE B	—	Defined Terms
SCHEDULE 3	—	Payment Instructions
SCHEDULE 4.9	—	Changes in Corporate Structure
SCHEDULE 4.10	—	Subsidiary Guarantors
SCHEDULE 5.3	—	Disclosure Materials
SCHEDULE 5.4	—	Subsidiaries of the Company and Ownership of Subsidiary Stock
SCHEDULE 5.5	—	Financial Statements
SCHEDULE 5.8	—	Certain Litigation
SCHEDULE 5.11	—	Patents, etc.
SCHEDULE 5.12	—	ERISA Affiliates and Plans
SCHEDULE 5.14	—	Use of Proceeds
SCHEDULE 5.15	—	Existing Indebtedness
EXHIBIT 1(a)	—	Form of 5.57% Series A Senior Note due September 15, 2011
EXHIBIT 1(b)	—	Form of 6.08% Series B Senior Note due July 15, 2014
EXHIBIT 4.4(a)	—	Form of Opinion of Special Counsel for the Company and certain of the Subsidiary Guarantors
EXHIBIT 4.4(b)	—	Form of Opinion of Special Counsel for Arizona Subsidiary Guarantor
EXHIBIT 4.4(c)	—	Form of Opinion of Special Counsel for Pennsylvania Subsidiary Guarantor
EXHIBIT 4.4(d)	—	Form of Opinion of Special Counsel for Nevada Subsidiary Guarantor
EXHIBIT 4.4(e)	—	Form of Opinion of Special Counsel for the Purchasers
EXHIBIT 4.10	—	Form of Subsidiary Guaranty

v

BROWN & BROWN, INC. 220 South Ridgewood Avenue Daytona Beach, Florida 32114

5.57% Series A Senior Notes due September 15, 2011 6.08% Series B Senior Notes due July 15, 2014

July 15, 2004

To each of the Purchasers listed in

the attached Schedule A (the "Purchasers")

Ladies and Gentlemen:

Brown & Brown, Inc., a Florida corporation (together with its successors and assigns, the "Company"), agrees with the Purchasers as follows:

1. AUTHORIZATION OF NOTES.

The Company will authorize the issuance and sale of:

(a) \$100,000,000 aggregate principal amount of its 5.57% Series A Senior Notes due September 15, 2011 (including any amendments, restatements or modifications from time to time, the **"Series A Notes"**, such term to include any such notes issued in substitution therefor pursuant to Section 13 of this Agreement); and

(b) \$100,000,000 aggregate principal amount of its 6.08% Series B Senior Notes due July 15, 2014 (including any amendments, restatements or modifications from time to time, the **"Series B Notes"**, such term to include any such notes issued in substitution therefor pursuant to Section 13 of this Agreement).

The Series A Notes and the Series B Notes are sometimes referred to herein collectively as the "**Notes**." The Series A Notes and the Series B Notes shall be substantially in the respective forms set out in Exhibits 1(a) and Exhibit 1(b), respectively, in each case with such changes thereto, if any, as may be approved by the Purchasers and the Company. Certain capitalized terms used in this Agreement are defined in Schedule B; references to a "Schedule" or an "Exhibit" are, unless otherwise specified, to a Schedule or an Exhibit attached to this Agreement.

2. SALE AND PURCHASE OF NOTES.

Subject to the terms and conditions of this Agreement, the Company will issue and sell to each Purchaser and each Purchaser will purchase from the Company, at the First Closing or the Second Closing, as applicable, as provided for in Section 3, Notes of the Series and in the principal amount specified opposite such Purchaser's name in Schedule A at the purchase price of 100% of the principal amount thereof. The Purchasers' obligations hereunder are several and not joint obligations and no Purchaser shall have any liability to any Person for the performance or non-performance by any other Purchaser hereunder.

3. CLOSINGS.

3.1. First Closing.

The closing of the sale and purchase of \$100,000,000 in aggregate principal amount of the Series B Notes to be purchased by the Purchasers shall occur at the offices of Bingham McCutchen LLP, 399 Park Avenue, New York, New York 10022, at 10:00 a.m., local time, at a closing (the "**First Closing**") on July 15, 2004 (the "**First Closing Date**"). At the First Closing the Company will deliver to the Purchasers the Series B Notes to be purchased by such Purchasers in the form of a single Series B Note (or such greater number of Series B Notes in denominations of at least \$100,000, as the Purchasers may request), dated the First Closing Date and registered in such Purchaser's name (or in the name of such Purchaser's nominee), as indicated in Schedule A, against delivery by such Purchaser to the Company or its order of immediately available funds in the amount of the purchase price therefor as directed by the Company in Schedule 3.

3.2. Second Closing.

The closing of the sale and purchase of \$100,000,000 in the aggregate principal amount of the Series A Notes to be purchased by the Purchasers on the Second Closing Date shall occur at the offices of Bingham McCutchen LLP, 399 Park Avenue, New York, New York 10022, at 10:00 a.m., local time, at a closing (the "**Second Closing**"; the First Closing and the Second Closing being sometimes referred to herein collectively as the "**Closings**" and individually as a "**Closing**") on September 15, 2004 (the "**Second Closing Date**"). At the Second Closing the Company will deliver to the Purchasers the Series A Notes to be purchased by such Purchasers in the form of a single Series A Note (or such greater number of Series A Notes in denominations of at least \$100,000 as the Purchasers may request), dated the Second Closing Date, and registered in such Purchaser's name (or in the name of such Purchaser's nominee), as indicated in Schedule A, against delivery by such Purchaser to the Company or its order of immediately available funds in the amount of the purchase price therefor as directed by the Company in Schedule 3. The failure of the Company to satisfy the conditions precedent set forth in Section 4 at the Second Closing shall relieve the Purchasers of all obligations under this Agreement with respect to the Second Closing.

3.3. Failure of the Company to Deliver.

If on either Closing Date, the Company fails to tender to the Purchasers the Notes to be acquired by such Purchasers on such Closing Date, or if the conditions specified in Section 4 have not been fulfilled to each Purchaser's satisfaction, such Purchaser shall, at its election, be relieved of all further obligations under this Agreement without thereby waiving any rights each such Purchaser may have by reason of such failure or such nonfulfillment. In addition, if such tender shall not be made to any Purchaser on the Second Closing Date, or if any of the conditions specified in Sections 4.1, 4.2, 4.3, 4.4(a), 4.4(b), 4.4(c), 4.4(e), 4.6 or 4.7 have not been fulfilled on such date, and such Purchaser does not purchase such Series A Notes as a result thereof, the Company shall pay to such Purchaser a Second Closing Compensation Amount with respect to the Series A Notes to be purchased by such Purchaser on such date. As used herein the term "Second Closing Compensation Amount" shall mean, with respect to any Series A Note to be purchased at the Second Closing, an amount equal to 0.25% of the aggregate principal amount of such Series A Notes to be purchaser as indicated in Schedule A.

4. CONDITIONS TO CLOSINGS.

Each Purchaser's obligation to purchase and pay for the Notes to be sold to it at each Closing is subject to the fulfillment to each such Purchaser's satisfaction, prior to or at such Closing, of the following conditions:

4.1. Representations and Warranties.

The representations and warranties of the Company in this Agreement shall be correct when made and at the time of each Closing.

4.2. Performance; No Default.

Each Obligor shall have performed and complied with all agreements and conditions contained in this Agreement required to be performed or complied with by it prior to or at each Closing and, after giving effect to the issue and sale of the Notes (and the application of the proceeds thereof as contemplated by Schedule 5.14), no Default or Event of Default shall have occurred and be continuing.

4.3. Compliance Certificates.

(a) <u>Company Officer's Certificate</u>. The Company shall have delivered to each Purchaser an Officer's Certificate, dated the applicable Closing Date, certifying that the conditions specified in Sections 4.1, 4.2 and 4.9 have been fulfilled.

(b) <u>Company Secretary's Certificate</u>. The Company shall have delivered to each Purchaser a certificate, dated the applicable Closing Date, certifying as to the resolutions attached thereto and other corporate proceedings relating to the authorization, execution and delivery of the Notes and the Agreements.

(c) <u>Subsidiary Guarantors' Secretary's Certificate</u>. Each of the Subsidiary Guarantors shall have delivered to each Purchaser a certificate, dated the applicable

Closing Date, certifying as to the resolutions attached thereto and other corporate proceedings relating to the authorization, execution and delivery of the Subsidiary Guaranty by such Subsidiary Guarantor.

4.4. Opinions of Counsel.

Each Purchaser shall have received opinions in form and substance satisfactory to it, dated such applicable Closing Date (a) from Holland & Knight LLP, counsel for the Company and certain of the Subsidiary Guarantors, covering the matters set forth in Exhibit 4.4(a) and covering such other matters incident to the transactions contemplated hereby as the Purchasers or their counsel may reasonably request (and the Company hereby instructs its counsel to deliver such opinion to each Purchaser), (b) from Gallagher & Kennedy, PA., Arizona counsel for Brown & Brown Insurance of Arizona, Inc., covering the matters set forth in Exhibit 4.4(b) and covering such other matters incident to the transactions contemplated hereby as the Purchasers or their counsel for Brown & Brown Insurance of Arizona, Inc., Pennsylvania counsel for Brown & Brown of Lehigh Valley, Inc., covering the matters set forth in Exhibit 4.4(c) and covering such other matters set forth in Exhibit 4.4(c) and covering such other matters contemplated hereby as the Purchasers or their counsel may reasonably request (and the Company hereby instructs its counsel for Brown & Brown of Lehigh Valley, Inc., covering the matters set forth in Exhibit 4.4(c) and covering such other matters incident to the transactions contemplated hereby as the Purchasers or their counsel may reasonably request (and the Company hereby instructs its counsel for Brown & Brown Insurance of Nevada, Inc., covering the matters set forth in Exhibit 4.4(d) and covering such other matters incident to the transactions contemplated hereby as the Purchasers or their counsel for Brown & Brown Insurance of Nevada, Inc., covering the matters set forth in Exhibit 4.4(d) and covering such other matters incident to the transactions contemplated hereby as the Purchasers or their counsel for Brown & Brown Insurance of Nevada, Inc., covering the matters set forth in Exhibit 4.4(d) and covering such other matters incident to the transactions contemplated hereby as the Purchasers or their counsel may reas

4.5. Purchase Permitted By Applicable Law, etc.

On such applicable Closing Date, each Purchaser's purchase of Notes shall (a) be permitted by the laws and regulations of each jurisdiction to which it is subject, without recourse to provisions (such as Section 1405(a)(8) of the New York Insurance Law) permitting limited investments by insurance companies without restriction as to the character of the particular investment, (b) not violate any applicable law or regulation (including, without limitation, Regulation T, U or X of the Board of Governors of the Federal Reserve System) and (c) not subject such Purchaser to any tax, penalty or liability under or pursuant to any applicable law or regulation, which law or regulation was not in effect on the First Closing Date. If so requested, each Purchaser shall have received an Officer's Certificate certifying as to such matters of fact as it may reasonably specify to enable such Purchaser to determine whether such purchase is so permitted.

4.6. Sale of Other Notes.

Contemporaneously with each Closing the Company shall sell to each Purchaser and each Purchaser shall purchase the Notes to be purchased by it at such Closing as specified in Schedule A.

4.7. Payment of Special Counsel Fees.

Without limiting the provisions of Section 15.1, the Company shall have paid on or before such Closing the fees, charges and disbursements of the Purchasers' special counsel referred to in Section 4.4 to the extent reflected in a statement of such counsel rendered to the Company at least one Business Day prior to such applicable Closing Date.

4.8. Private Placement Number.

A Private Placement Number issued by Standard & Poor's CUSIP Service Bureau (in cooperation with the Securities Valuation Office of the National Association of Insurance Commissioners) shall have been obtained for each Series of Notes.

4.9. Changes in Corporate Structure.

Except as specified in Schedule 4.9, no Obligor shall have changed its jurisdiction of incorporation or been a party to any merger or consolidation and shall not have succeeded to all or any substantial part of the liabilities of any other entity, at any time following the date of the most recent financial statements referred to in Schedule 5.5.

4.10. Subsidiary Guaranty.

Each Subsidiary Guarantor listed in Schedule 4.10 shall have executed and delivered to the Purchasers a guaranty agreement (as may be amended, restated or modified from time to time, the "**Subsidiary Guaranty**"), substantially in the form of Exhibit 4.10.

4.11. Amendments to Bank Credit Agreement.

The Company shall have delivered to the Purchasers fully executed copies of (a) that certain First Amendment to Amended and Restated Revolving and Term Loan Agreement dated as of July 15, 2004 between the Company and SunTrust Bank, (b) that certain Agreement Regarding Confirmation and Release of Guarantors in connection with Revolving and Term Loan Agreement between the Company and SunTrust Bank dated as of July 15, 2004, (c) that certain Second Amendment to Revolving Loan Agreement dated as of July 15, 2004 between the Company and SunTrust Bank and (d) that certain Agreement Regarding Confirmation and Release of Guarantors in connection with Revolving Loan Agreement between the Company and SunTrust Bank dated as of July 15, 2004, each certified as true and correct by a Responsible Officer of the Company.

4.12. Offeree Letter.

SunTrust Capital Markets, Inc. shall have delivered to the Company, its counsel, each of the Purchasers and the Purchasers' special counsel an offeree letter, in form and substance satisfactory to each Purchaser and the Company, confirming the manner of the offering of the Notes by SunTrust Capital Markets, Inc.

4.13. Wire Instructions.

No later than two (2) Business Days prior to each Closing, the Company shall have delivered to the Purchasers a letter on the Company's letterhead which provides the wiring instructions for the Purchasers to wire their funds for payment of the Notes.

4.14. Proceedings and Documents.

All corporate and other proceedings in connection with the transactions contemplated by this Agreement and all documents and instruments incident to such transactions shall be satisfactory to each Purchaser and its special counsel, and each Purchaser and its special counsel shall have received all such counterpart originals or certified or other copies of such documents as such Purchaser or its counsel may reasonably request.

5. REPRESENTATIONS AND WARRANTIES OF THE COMPANY.

The Company represents and warrants to each Purchaser that:

5.1. Organization; Power and Authority.

The Company is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by law, other than those jurisdictions as to which the failure to be so qualified or in good standing would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Company has the corporate power and authority to own or hold under lease the properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the Notes and to perform the provisions hereof and thereof.

5.2. Authorization.

(a) This Agreement and the Notes have been duly authorized by all necessary corporate action on the part of the Company, and this Agreement constitutes, and upon execution and delivery thereof each Note will constitute, legal, valid and binding obligations of the Company enforceable against the Company in accordance with their respective terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

(b) The Subsidiary Guaranty has been duly authorized by all necessary corporate action on the part of each Subsidiary Guarantor party thereto, and the Subsidiary Guaranty constitutes the legal, valid and binding obligation of such Subsidiary Guarantor enforceable against such Subsidiary Guarantor in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors'



rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

5.3. Disclosure.

The Company, through its agent, SunTrust Capital Markets, Inc., has delivered to each Purchaser a copy of a Confidential Offering Memorandum, dated June 2004 (the **"Memorandum"**), relating to the transactions contemplated hereby. Except as disclosed in Schedule 5.3, this Agreement, the Memorandum, the documents, certificates or other writings identified in Schedule 5.3 and the financial statements listed in Schedule 5.5, taken as a whole, do not contain any untrue statement of a material fact or omit to state any material fact necessary to make the statements therein not misleading in light of the circumstances under which they were made. Except as disclosed in the Memorandum or as expressly described in Schedule 5.3, or in one of the documents, certificates or other writings identified therein, or in the financial statements listed in Schedule 5.5, since December 31, 2003 there has been no change in the financial condition, operations, business or properties of the Company or any of its Subsidiaries except changes that individually or in the aggregate would not reasonably be expected to have a Material Adverse Effect.

5.4. Organization and Ownership of Shares of Subsidiaries.

(a) Schedule 5.4 is (except as noted therein) a complete and correct list of the Company's Subsidiaries, showing, as to each Subsidiary, the correct name thereof, the jurisdiction of its organization, and the percentage of shares of each class of its Capital Stock outstanding owned by the Company and each other Subsidiary.

(b) All of the outstanding shares of Capital Stock of each Subsidiary shown in Schedule 5.4 as being owned by the Company and its Subsidiaries have been validly issued, are fully paid and nonassessable and are owned by the Company or another Subsidiary free and clear of any Lien (except as otherwise disclosed in Schedule 5.4).

(c) Each Subsidiary identified in Schedule 5.4 is a corporation or other legal entity duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, and is duly qualified as a foreign corporation or other legal entity and is in good standing in each jurisdiction in which such qualification is required by law, other than those jurisdictions as to which the failure to be so qualified or in good standing would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each such Subsidiary has the corporate or other power and authority to own or hold under lease the properties it purports to own or hold under lease and to transact the business it transacts and proposes to transact, to execute and deliver the Subsidiary Guaranty, and to perform the provisions thereof.

5.5. Financial Statements.

The Company has delivered to each Purchaser copies of the financial statements of the Company and its Subsidiaries listed on Schedule 5.5. All of said financial statements (including in each case the related schedules and notes) fairly present in all material respects the consolidated financial position of the Company and its Subsidiaries as of the respective dates

specified in such Schedule and the consolidated results of their operations and cash flows for the respective periods so specified and have been prepared in accordance with GAAP consistently applied throughout the periods involved except as set forth in the notes thereto (subject, in the case of any interim financial statements, to normal year-end adjustments).

5.6. Compliance with Laws, Other Instruments, etc.

The execution, delivery and performance by each Obligor of the Financing Documents to which such Obligor is a party will not (a) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of the Company or any Subsidiary under, any indenture, mortgage, deed of trust, loan, purchase or credit agreement, lease, corporate charter or by-laws, or any other Material agreement or instrument to which the Company or any Subsidiary is bound or by which the Company or any Subsidiary or any of their respective properties may be bound or affected, (b) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental Authority applicable to the Company or any Subsidiary or (c) violate any provision of any statute or other rule or regulation of any Governmental Authority applicable to the Company or any Subsidiary.

5.7. Governmental Authorizations, etc.

No consent, approval or authorization of, or registration, filing or declaration with, any Governmental Authority is required in connection with the execution, delivery or performance by (a) the Company of this Agreement or the Notes and (b) each Subsidiary Guarantor of the Subsidiary Guaranty.

5.8. Litigation; Observance of Statutes and Orders.

(a) Except as disclosed in Schedule 5.8, there are no actions, suits or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company or any Subsidiary or any property of the Company or any Subsidiary in any court or before any arbitrator of any kind or before or by any Governmental Authority that, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect.

(b) Neither the Company nor any Subsidiary is in default under any order, judgment, decree or ruling of any court, arbitrator or Governmental Authority or is in violation of any applicable law, ordinance, rule or regulation (including without limitation Environmental Laws) of any Governmental Authority, which default or violation, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect.

5.9. Taxes.

The Company and its Subsidiaries have filed all income tax returns that are required to have been filed in any jurisdiction, and have paid all taxes shown to be due and payable on such returns and all other taxes and assessments payable by them, to the extent such taxes and assessments have become due and payable and before they have become delinquent, except for

any taxes and assessments (a) the amount of which is not individually or in the aggregate Material or (b) the amount, applicability or validity of which is currently being contested in good faith by appropriate proceedings and with respect to which the Company or a Subsidiary, as the case may be, has established adequate reserves in accordance with GAAP. Neither the Company nor any Subsidiary has entered into any agreement with the Internal Revenue Service tolling the statute of limitations that is currently in effect with respect to any tax returns or any taxes due in respect thereof.

5.10. Title to Property; Leases.

The Company and its Subsidiaries have good and sufficient title to their respective Material properties, including all such properties reflected in the most recent audited balance sheet referred to in Section 5.5 or purported to have been acquired by the Company or any Subsidiary after said date (except as sold or otherwise disposed of in the ordinary course of business), in each case free and clear of Liens prohibited by this Agreement, except for those defects in title and Liens that, individually or in the aggregate, would not have a Material Adverse Effect. All Material leases are valid and subsisting and are in full force and effect in all material respects.

5.11. Licenses, Permits, etc.

Except as disclosed in Schedule 5.11, the Company and its Subsidiaries own or possess all licenses, permits, franchises, authorizations, patents, copyrights, service marks, trademarks and trade names, or rights thereto, that are Material, without known conflict with the rights of others, except for those conflicts that, individually or in the aggregate, would not have a Material Adverse Effect.

5.12. Compliance with ERISA.

(a) The Company and each ERISA Affiliate have operated and administered each Plan in compliance with all applicable laws except for such instances of noncompliance as have not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any ERISA Affiliate has incurred any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans (as defined in section 3 of ERISA), and no event, transaction or condition has occurred or exists that would reasonably be expected to result in the incurrence of any such liability by the Company or any ERISA Affiliate, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate, in either case pursuant to Title I or IV of ERISA or to such penalty or excise tax provisions or to section 401(a)(29) or section 412 of the Code, other than such liabilities or Liens as would not be individually or in the aggregate Material.

(b) The present value of the aggregate benefit liabilities under each of the Plans (other than Multiemployer Plans), determined as of the end of such Plan's most recently ended plan year on the basis of the actuarial assumptions specified for funding purposes in such Plan's most recent actuarial valuation report, did not exceed the aggregate current value of the assets of such Plan allocable to such benefit liabilities by

more than \$3,000,000 in the case of any single Plan and by more than \$5,000,000 in the aggregate for all Plans. The term **"benefit liabilities"** has the meaning specified in section 4001 of ERISA and the terms **"current value"** and **"present value"** have the meaning specified in section 3 of ERISA.

(c) The Company and its ERISA Affiliates have not incurred withdrawal liabilities (and are not subject to contingent withdrawal liabilities) under section 4201 or 4204 of ERISA in respect of Multiemployer Plans that individually or in the aggregate are Material.

(d) The expected postretirement benefit obligation (determined as of the last day of the Company's most recently ended fiscal year in accordance with Financial Accounting Standards Board Statement No. 106, without regard to liabilities attributable to continuation coverage mandated by section 4980B of the Code) of the Company and its Subsidiaries is not Material.

(e) The execution and delivery of this Agreement and the issuance and sale of the Notes hereunder will not involve any transaction that is subject to the prohibitions of section 406 of ERISA or in connection with which a tax could be imposed pursuant to section 4975(c)(1)(A)-(D) of the Code. The representation by the Company in the first sentence of this Section 5.12(e) is made in reliance upon and subject to the accuracy of each Purchaser's representation in Section 6.2 as to the sources of the funds to be used to pay the purchase price of the Notes to be purchased by such Purchaser.

(f) Schedule 5.12 sets forth all ERISA Affiliates and all "employee benefit plans" maintained by the Company (or an "affiliate" thereof) or in respect of which the Notes could constitute an "employer security" ("employee benefit plan" has the meaning specified in section 3 of ERISA, "affiliate" has the meaning specified in section 407(d) of ERISA and section V of the Department of Labor Prohibited Transaction Exemption 95-60 (60 FR 35925, July 12, 1995) and "employer security" has the meaning specified in section 407(d) of ERISA).

5.13. Private Offering by the Company.

Neither the Company nor anyone acting on its behalf has offered the Notes or any similar securities for sale to, or solicited any offer to buy any of the same from, or otherwise approached or negotiated in respect thereof with, any person other than the Purchasers and not more than thirty-eight (38) other Institutional Investors (as defined in clause (c) of the definition of such term), each of which has been offered the Notes at a private sale for investment. Neither the Company nor anyone acting on its behalf has taken, or will take, any action that would subject the issuance or sale of the Notes to the registration requirements of Section 5 of the Securities Act.

5.14. Use of Proceeds; Margin Regulations.

The Company will apply the proceeds of the sale of the Notes as set forth in Schedule 5.14. No part of the proceeds from the sale of the Notes hereunder will be used, directly or indirectly, for the purpose of buying or carrying any margin stock within the meaning of

Regulation U of the Board of Governors of the Federal Reserve System (12 CFR 221), or for the purpose of buying or carrying or trading in any securities under such circumstances as to involve the Company in a violation of Regulation X of said Board (12 CFR 224) or to involve any broker or dealer in a violation of Regulation T of said Board (12 CFR 220). Margin stock does not constitute more than 5% of the value of the consolidated assets of the Company and its Subsidiaries and the Company does not have any present intention that margin stock will constitute more than 5% of the value of such assets. As used in this Section, the terms **"margin stock"** and **"purpose of buying or carrying"** shall have the meanings assigned to them in said Regulation U.

5.15. Existing Indebtedness.

Except as described therein, Schedule 5.15 sets forth a complete and correct list of all outstanding Indebtedness of the Company and its Subsidiaries as of July 14, 2004 and sets forth a listing of the Subsidiaries obligated under the Subsidiary Guaranty. Since July 14, 2004 there has been no Material change in the amounts, interest rates, sinking funds, installment payments or maturities of the Indebtedness of the Company or its Subsidiaries. Neither the Company nor any Subsidiary and no event or condition exists with respect to any Indebtedness of the Company or any Subsidiary the outstanding principal amount of which exceeds \$1,000,000 that would permit (or that with notice or the lapse of time, or both, would permit) one or more Persons to cause such Indebtedness to become due and payable before its stated maturity or before its regularly scheduled dates of payment.

5.16. Foreign Assets Control Regulations, etc.

Neither the sale of the Notes by the Company hereunder nor its use of the proceeds thereof will violate (a) the Trading with the Enemy Act, as amended, or (b) any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto. Without limiting the foregoing, (i) neither the Company nor any Subsidiary is or will become a blocked Person described by Section 1 of Executive Order 13224 of September 24, 2001, Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (31 CFR Part 595 et seq.) and (ii) the Company and its Subsidiaries shall use all commercially reasonable efforts to refrain from engaging in any dealings or transactions, or is otherwise associated, with any such Person.

5.17. Status under Certain Statutes.

Neither the Company nor any Subsidiary is (a) subject to regulation under the Investment Company Act of 1940, as amended, the Public Utility Holding Company Act of 1935, as amended, or the Federal Power Act, as amended or (b) in violation of the USA Patriot Act.

6. REPRESENTATIONS OF THE PURCHASER.

6.1. Purchase for Investment.

Each Purchaser represents that it is purchasing the Notes for its own account or for one or more separate accounts or investments fund maintained or managed by such Purchaser or for the account of one or more pension or trust funds over which such Purchaser has investment discretion and not with a view to the distribution thereof, <u>provided</u> that the disposition of such Purchaser's property shall at all times be within such Purchaser's control. Each Purchaser understands that the Notes have not been registered under the Securities Act and may be resold only if registered pursuant to the provisions of the Securities Act or if an exemption from registration is available, except under circumstances where neither such registration nor such an exemption is required by law, and that the Company is not required to register the Notes.

6.2. Source of Funds.

Each Purchaser represents that at least one of the following statements is an accurate representation as to each source of funds (a "**Source**") to be used by such Purchaser to pay the purchase price of the Notes to be purchased by such Purchaser hereunder:

(a) the Source is an "insurance company general account" (as defined in the United States Department of Labor Prohibited Transaction Exemption (**"PTE"**) 95-60) (issued July 12, 1995) and in respect thereof each Purchaser represents that there is no "employee benefit plan" (as defined in section 3(3) of ERISA and section 4975(e)(1) of the Code, treating as a single plan all plans maintained by the same employer or employee organization or affiliate thereof) with respect to which the amount of the general account reserves and liabilities of all contracts held by or on behalf of such plan exceeds 10% of the total reserves and liabilities of such general account (exclusive of separate account liabilities) plus surplus, as set forth in the National Association of Insurance Commissioners' Annual Statement filed with such Purchaser's state of domicile; or

(b) the Source is a separate account that is maintained solely in connection with such Purchaser's fixed contractual obligations under which the amounts payable, or credited, to any employee benefit plan (or its related trust) that has any interest in such separate account (or to any participant or beneficiary of such employee benefit plan (including any annuitant)) are not affected in any manner by the investment performance of the separate account; or

(c) the Source is either (i) an insurance company pooled separate account, within the meaning of PTE 90-1 (issued January 29, 1990), or (ii) a bank collective investment fund, within the meaning of the PTE 91-38 (issued July 12, 1991) and, except as disclosed by such Purchaser to the Company in writing pursuant to this paragraph (c), no employee benefit plan or group of plans maintained by the same employer or employee organization beneficially owns more than 10% of all assets allocated to such pooled separate account or collective investment fund; or

(d) the Source constitutes assets of an "investment fund" (within the meaning of part V of the QPAM Exemption) managed by a "qualified professional asset manager" or "QPAM" (within the meaning of part V of the QPAM Exemption), no employee benefit plan's assets that are included in such investment fund, when combined with the assets of all other employee benefit plans established or maintained by the same employer or by an affiliate (within the meaning of section V(c)(1) of the QPAM Exemption) of such employer or by the same employee organization and managed by such QPAM, exceed 20% of the total client assets managed by such QPAM, the conditions of part I(c) and (g) of the QPAM Exemption are satisfied, neither the QPAM nor a person controlling or controlled by the QPAM (applying the definition of "control" in section V(e) of the QPAM Exemption) owns a 5% or greater interest in the Company and (i) the identity of such QPAM and (ii) the names of all employee benefit plans whose assets are included in such investment fund have been disclosed to the Company in writing pursuant to this paragraph (d); or

(e) the Source constitutes assets of a "plan(s)" (within the meaning of section IV of PTE 96-23 (the "**INHAM Exemption**")) managed by an "inhouse asset manager" or "INHAM" (within the meaning of part IV of the INHAM exemption), the conditions of part I(a), (g) and (h) of the INHAM Exemption are satisfied, neither the INHAM nor a person controlling or controlled by the INHAM (applying the definition of "control" in section IV(d) of the INHAM Exemption) owns a 5% or greater interest in the Company and (i) the identity of such INHAM and (ii) the name(s) of the employee benefit plan(s) whose assets constitute the Source have been disclosed to the Company in writing pursuant to this paragraph (e); or

(f) the Source is a governmental plan; or

(g) the Source is one or more employee benefit plans, or a separate account or trust fund comprised of one or more employee benefit plans, each of which has been identified to the Company in writing pursuant to this paragraph (g); or

(h) the Source does not include assets of any employee benefit plan, other than a plan exempt from the coverage of ERISA.

As used in this Section 6.2, the terms "governmental plan" and "separate account" shall have the respective meanings assigned to such terms in section 3 of ERISA.

7. INFORMATION AS TO COMPANY.

7.1. Financial and Business Information.

The Company shall deliver to each holder of Notes that is an Institutional Investor:

(a) <u>Quarterly Statements</u>. — within 60 days after the end of each quarterly fiscal period in each fiscal year of the Company (other than the last quarterly fiscal period of each such fiscal year), duplicate copies of,

(i) a consolidated balance sheet of the Company and its Subsidiaries as at the end of such quarter, and

(ii) consolidated statements of income, changes in shareholders' equity and cash flows of the Company and its Subsidiaries, for such quarter and (in the case of the second and third quarters) for the portion of the fiscal year ending with such quarter,

setting forth in each case in comparative form the figures for the corresponding periods in the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP applicable to quarterly financial statements generally, and certified by a Senior Financial Officer as fairly presenting, in all material respects, the consolidated financial position of the companies being reported on and their consolidated results of operations and cash flows, subject to changes resulting from year-end adjustments, <u>provided</u> that delivery within the time period specified above of copies of the Company's Quarterly Report on Form 10-Q prepared in compliance with the requirements therefor and filed with the Securities and Exchange Commission shall be deemed to satisfy the requirements of this Section 7.1(a);

(b) Annual Statements. — within 90 days after the end of each fiscal year of the Company, duplicate copies of,

- (i) a consolidated balance sheet of the Company and its Subsidiaries, as at the end of such year, and
- (ii) consolidated statements of income, changes in shareholders' equity and cash flows of the Company and its Subsidiaries, for such year,

setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP, and accompanied by an opinion thereon of independent certified public accountants of recognized national standing, which opinion shall state that such financial statements present fairly, in all material respects, the financial position of the companies being reported upon and their results of operations and cash flows and have been prepared in conformity with GAAP, and that the examination of such accountants in connection with such financial statements has been made in accordance with generally accepted auditing standards, and that such audit provides a reasonable basis for such opinion in the circumstances, <u>provided</u> that the delivery within the time period specified above of the Company's Annual Report on Form 10-K for such fiscal year (together with the Company's annual report to shareholders, if any, prepared pursuant to Rule 14a-3 under the Exchange Act) prepared in accordance with the requirements therefor and filed with the Securities and Exchange Commission shall be deemed to satisfy the requirements of this Section 7.1(b);

(c) <u>SEC and Other Reports</u>. — promptly upon their becoming available, one copy of (i) each financial statement, report, notice or proxy statement sent by the Company or any Subsidiary to public securities holders generally, and (ii) each regular or periodic report, each registration statement that shall have become effective (without

exhibits except as expressly requested by such holder), and each final prospectus and all amendments thereto filed by the Company or any Subsidiary with the Securities and Exchange Commission;

(d) <u>Notice of Default or Event of Default</u>. — promptly, and in any event within five Business Days after a Responsible Officer becoming aware of the existence of any Default or Event of Default, a written notice specifying the nature and period of existence thereof and what action the Company is taking or proposes to take with respect thereto;

(e) <u>ERISA Matters</u>. — with reasonable promptness, and in any event within five Business Days after a Responsible Officer becoming aware of any of the following, a written notice setting forth the nature thereof and the action, if any, that the Company or an ERISA Affiliate proposes to take with respect thereto:

(i) with respect to any Plan, any reportable event, as defined in section 4043(b) of ERISA and the regulations thereunder, for which notice thereof has not been waived pursuant to such regulations as in effect on the First Closing Date; or

(ii) the taking by the PBGC of steps to institute, or the threatening by the PBGC of the institution of, proceedings under section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan, or the receipt by the Company or any ERISA Affiliate of a notice from a Multiemployer Plan that such action has been taken by the PBGC with respect to such Multiemployer Plan; or

(iii) any event, transaction or condition that could result in the incurrence of any liability by the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or such penalty or excise tax provisions, if such liability or Lien, taken together with any other such liabilities or Liens then existing, would reasonably be expected to have a Material Adverse Effect;

(f) <u>Rule 144A</u>. — with reasonable promptness, after any holder of Notes so requests, such information regarding the Obligors required to satisfy the requirements of Rule 144A under the Securities Act, as amended from time to time, in connection with any contemplated transfer of the Notes pursuant to Rule 144A; and

(g) <u>Requested Information</u>. — with reasonable promptness, such other data and information relating to the business, operations, affairs, financial condition, assets or properties of the Company or any of its Subsidiaries or relating to the ability of any Obligor to perform its obligations under the Financing Documents to which it is a party as from time to time may be reasonably requested by any such holder of Notes.

7.2. Officer's Certificate.

Each set of financial statements delivered to a holder of Notes pursuant to Section 7.1(a) or Section 7.2(b) hereof shall be accompanied by a certificate of a Senior Financial Officer setting forth:

(a) <u>Covenant Compliance</u>. — the information (including detailed calculations) required in order to establish whether the Company was in compliance with the requirements of Section 10.5, Section 10.6 and Section 10.9, during the quarterly or annual period covered by the statements then being furnished (including with respect to each such Section, where applicable, the calculations of the maximum or minimum amount, ratio or percentage, as the case may be, permissible under the terms of such Sections, and the calculation of the amount, ratio or percentage at the end of such period or then in existence); and

(b) <u>Event of Default</u>. — a statement that such officer has reviewed the relevant terms hereof and has made, or caused to be made, under his or her supervision, a review of the transactions and conditions of the Company and its Subsidiaries from the beginning of the quarterly or annual period covered by the statements then being furnished to the date of the certificate and that such review shall not have disclosed the existence during such period of any condition or event that constitutes a Default or an Event of Default or, if any such condition or event existed or exists (including, without limitation, any such event or condition resulting from the failure of the Company or any Subsidiary to comply with any Environmental Law), specifying the nature and period of existence thereof and what action the Company shall have taken or proposes to take with respect thereto.

7.3. Inspection.

The Company shall permit the representatives of each holder of Notes that is an Institutional Investor:

(a) <u>No Default</u>. — if no Default or Event of Default then exists, at the expense of such holder and upon reasonable prior notice to the Company and during normal business hours unless otherwise agreed by the Company, to visit the principal executive office of the Company, to discuss the affairs, finances and accounts of the Company and its Subsidiaries with the Company's officers, and, with the consent of the Company (which consent will not be unreasonably withheld) to visit the other offices and properties of the Company and each Subsidiaries shall be required to disclose information that the Company or such Subsidiary is prohibited from disclosing by the terms of a confidentiality agreement binding upon the Company or such Subsidiary and not entered into in contemplation of this clause (a) provided the Company has made a good faith attempt to obtain consent from the party in whose favor the obligation of confidentiality was made to permit disclosure of the relevant information; and

(b) <u>Default</u>. — if a Default or Event of Default then exists, at the expense of the Company to visit and inspect any of the offices or properties of the Company or any Subsidiary, to examine all their respective books of account, records, reports and other papers, to make copies and extracts therefrom, and to discuss their respective affairs, finances and accounts with their respective officers and independent public accountants (and by this provision the Company authorizes said accountants to discuss the affairs, finances and accounts of the Company and its Subsidiaries), all at such times and as often as may be requested.

8. PREPAYMENT OF THE NOTES.

8.1. Required Prepayments.

The outstanding principal amount, if any, of (a) the Series A Notes shall be repaid by the Company, at par and without payment of the Make-Whole Amount or any premium, on September 15, 2011, and (b) the Series B Notes shall be repaid by the Company, at par and without payment of the Make-Whole Amount or any premium, on July 15, 2014.

8.2. Optional Prepayments with Make-Whole Amount.

The Company may, at its option, upon notice as provided below, prepay at any time all, or from time to time any part of, the Notes, in an amount not less than 5% of the aggregate principal amount of the Notes then outstanding in the case of a partial prepayment, at 100% of the principal amount so prepaid, plus the Make-Whole Amount determined for the prepayment date with respect to such principal amount. The Company will give each holder of Notes written notice of each optional prepayment under this Section 8.2 not less than 30 days and not more than 60 days prior to the date fixed for such prepayment. Each such notice shall specify such date, the aggregate principal amount of the Notes to be prepaid on such date, the principal amount of each Note and of each Series held by such holder to be prepaid (determined in accordance with Section 8.5), and the interest to be paid on the prepayment date with respect to such principal amount being prepaid, and shall be accompanied by a certificate of a Senior Financial Officer as to the estimated Make-Whole Amount due in connection with such prepayment (calculated as if the date of such notice were the date of the prepayment), setting forth the details of such computation. Two Business Days prior to such prepayment, the Company shall deliver to each holder of Notes a certificate of a Senior Financial Officer specifying the calculation of such Make-Whole Amount as of the specified prepayment date.

8.3. Prepayment of Notes Upon Change in Control.

(a) <u>Notice of Change in Control</u>. The Company will, within five Business Days after any Responsible Officer has knowledge of the occurrence of any Change in Control, give written notice of such Change in Control to each holder of Notes. In the case that a Change in Control has occurred, such notice shall contain and constitute an offer to prepay Notes as described in subparagraph (b) of this Section 8.3 and shall be accompanied by the certificate described in subparagraph (e) of this Section 8.3.

(b) <u>Offer to Prepay Notes</u>. The offer to prepay Notes contemplated by subparagraph (a) of this Section 8.3 shall be an offer to prepay, in accordance with and

subject to this Section 8.3, all, but not less than all, of the Notes held by each holder (in this case and in the case of Section 8.4(a) only, "**holder**" in respect of any Note registered in the name of a nominee for a disclosed beneficial owner shall mean such beneficial owner) on a date specified in such offer (the "**Change in Control Prepayment Date**") that is not less than 45 days and not more than 60 days after the date of such offer (if the Change in Control Prepayment Date shall not be specified in such offer, the Change in Control Prepayment Date shall be the 45th day after the date of such offer).

(c) <u>Acceptance</u>; <u>Rejection</u>. A holder of Notes may accept the offer to prepay made pursuant to this Section 8.3 by causing a notice of such acceptance to be delivered to the Company not more than 30 days after the date the written offer notice referred to in subsection (a) of this Section 8.3 is given to the holders of the Notes. A failure by a holder of Notes to respond to an offer to prepay made pursuant to this Section 8.3 shall be deemed to constitute a rejection of such offer by such holder on the last date for acceptance provided for in this subsection (c).

(d) <u>Prepayment</u>. Prepayment of the Notes to be prepaid pursuant to this Section 8.3 shall be at 100% of the principal amount of such Notes, together with interest on such Notes accrued to the applicable Change in Control Prepayment Date. Each prepayment of Notes pursuant to this Section 8.3 shall be made on the applicable Change in Control Prepayment Date.

(e) <u>Officer's Certificate</u>. Each offer to prepay the Notes pursuant to this Section 8.3 shall be accompanied by a certificate, executed by a Senior Financial Officer of the Company and dated the date of such offer, specifying: (i) the proposed Change in Control Prepayment Date; (ii) that such offer is made pursuant to this Section 8.3; (iii) the principal amount of each Note offered to be prepaid; (iv) the interest that would be due on each Note offered to be prepaid as of the Change in Control Prepayment Date; (v) that the conditions of this Section 8.3 have been fulfilled; and (vi) in reasonable detail, the nature and date of the Change in Control (including, if known, the name or names of the Person or Persons acquiring control).

(f) "<u>Change in Control Defined.</u>" A "**Change in Control**" shall occur if any Person or group of Persons acting in concert, together with Affiliates thereof (other than members of the Brown Family), shall in the aggregate, directly or indirectly, control or own (beneficially or otherwise) more than 50% of the issued and outstanding Voting Stock of the Company at any time after the First Closing Date or shall otherwise have the ability to elect a majority of the members of the board of directors of the Company.

8.4. Offer to Prepay upon the Sale of Certain Assets.

(a) <u>Notice and Offer</u>. In the event of any Debt Prepayment Application under Section 10.3 of this Agreement (a "**Debt Prepayment Transfer**"), the Company will offer to prepay the Notes (the "**Transfer Prepayment Offer**") in compliance with the requirements for a Debt Prepayment Application (as set forth in the definition thereof), and give written notice of such offer to each holder of Notes. Such written notice shall contain, and such written notice shall constitute, an irrevocable offer to prepay, at the

election of each holder, a portion of the Notes held by such holder equal to such holder's Ratable Portion of the Net Proceeds in respect of such Debt Prepayment Transfer on a date specified in such notice (the **"Transfer Prepayment Date"**) that is not less than thirty (30) days and not more than sixty (60) days after the date of such notice, together with interest on the amount to be so prepaid accrued to the Transfer Prepayment Date. If the Transfer Prepayment Date shall not be specified in such notice, the Transfer Prepayment Date shall be the fortieth (40th) day after the date of such notice.

(b) <u>Acceptance and Payment</u>. To accept such Transfer Prepayment Offer, a holder of Notes shall cause a notice of such acceptance to be delivered to the Company not later than ten (10) Business Days after the date of such written notice from the Company, provided, that failure to accept such offer in writing within ten (10) Business Days after the date of such written notice shall be deemed to constitute a rejection of the Transfer Prepayment Offer. If so accepted by any holder of a Note, such offered prepayment (equal to not less than such holder's Ratable Portion of the Net Proceeds in respect of such Debt Prepayment Transfer) shall be due and payable on the Transfer Prepayment Date. Such offered prepayment shall be made at one hundred percent (100%) of the principal amount of such Notes being so prepaid, together with interest on such principal amount then being prepaid accrued to the Transfer Prepayment Date.

(c) <u>Officer's Certificate</u>. Each Transfer Prepayment Offer pursuant to this Section 8.4 shall be accompanied by a certificate, executed by a Senior Financial Officer of the Company and dated the date of such offer, specifying (i) the Transfer Prepayment Date, (ii) the Net Proceeds in respect of the applicable Debt Prepayment Transfer, (iii) that such offer is being made pursuant to Section 8.4 and Section 10.3 of this Agreement, (iv) the principal amount of each Note offered to be prepaid, (v) the interest that would be due on each Note offered to be prepaid, accrued to the Transfer Prepayment Date, (vi) the calculation of the Ratable Portion of the Net Proceeds in respect of such Debt Prepayment Transfer and (vii) in reasonable detail, the nature of the Transfer giving rise to such Debt Prepayment Transfer and certifying that no Default or Event of Default exists or would exist after giving effect to the prepayment contemplated by such offer.

(d) <u>Notice Concerning Status of Holders of Notes</u>. Promptly after each Transfer Prepayment Date and the making of all prepayments contemplated on such Transfer Prepayment Date under this Section 8.4 (and, in any event, within thirty (30) days thereafter), the Company shall deliver to each holder of Notes a certificate signed by a Senior Financial Officer of the Company containing a list of the then current holders of Notes (together with their addresses) and setting forth as to each such holder the outstanding principal amount of Notes held by such holder at such time.

8.5. Allocation of Partial Prepayments.

In the case of each partial prepayment of the Notes pursuant to Section 8.2, the principal amount of the Notes to be prepaid shall be allocated among all of the Notes at the time outstanding in proportion, as nearly as practicable, to the respective unpaid principal amounts thereof not theretofore called for prepayment, without regard to the Series of such Notes.

8.6. Maturity; Surrender, etc.

In the case of each prepayment of Notes pursuant to this Section 8, the principal amount of each Note to be prepaid shall mature and become due and payable on the date fixed for such prepayment, together with interest on such principal amount accrued to such date and the applicable Make-Whole Amount, if any. From and after such date, unless the Company shall fail to pay such principal amount when so due and payable, together with the interest and Make-Whole Amount, if any, as aforesaid, interest on such principal amount shall cease to accrue. Any Note paid or prepaid in full shall be surrendered to the Company and cancelled and shall not be reissued, and no Note shall be issued in lieu of any prepaid principal amount of any Note.

8.7. Purchase of Notes.

The Company will not and will not permit any Affiliate to purchase, redeem, prepay or otherwise acquire, directly or indirectly, any of the outstanding Notes except (a) upon the payment or prepayment of the Notes in accordance with the terms of this Agreement and the Notes or (b) pursuant to an offer to purchase made by the Company or an Affiliate pro rata (without regard to Series) to the holders of all Notes at the time outstanding upon the same terms and conditions. Any such offer shall provide each holder with sufficient information to enable it to make an informed decision with respect to such offer, and shall remain open for at least 14 Business Days. If the holders of more than 15% of the principal amount of the Notes then outstanding timely accept such offer, the Company shall promptly notify the remaining holders of such fact and the expiration date for the acceptance by holders of Notes of such offer. The Company will promptly cancel all Notes acquired by it or any Affiliate pursuant to any payment, prepayment or purchase of Notes pursuant to any provision of this Agreement and no Notes may be issued in substitution or exchange for any such Notes.

8.8. Make-Whole Amount.

The term **"Make-Whole Amount"** means, with respect to any Note of either Series, an amount equal to the excess, if any, of the Discounted Value of the Remaining Scheduled Payments with respect to the Called Principal of such Note of such Series over the amount of such Called Principal, <u>provided</u> that the Make-Whole Amount may in no event be less than zero. For the purposes of determining the Make-Whole Amount, the following terms have the following meanings:

"Called Principal" means, with respect to any Note of either Series, the principal of such Note that is to be prepaid pursuant to Section 8.2 or has become or is declared to be immediately due and payable pursuant to Section 12.1, as the context requires.

"Discounted Value" means, with respect to the Called Principal of any Note of either Series, the amount obtained by discounting all Remaining Scheduled Payments with respect to such Called Principal from their respective scheduled due dates to the Settlement Date with respect to such Called Principal, in accordance with accepted financial practice and at a discount factor (applied on the same periodic basis as that on

which interest on such Series of Notes is payable) equal to the Reinvestment Yield with respect to such Called Principal.

"Reinvestment Yield" means, with respect to the Called Principal of any Note of either Series, 0.50% over the yield to maturity implied by (i) the yields reported, as of 10:00 A.M. (New York City time) on the second Business Day preceding the Settlement Date with respect to such Called Principal, on the display designated as "Page PX1" on the Bloomberg Financial Market Service (or such other display as may replace Page PX1 on Bloomberg Financial Market Service) for actively traded U.S. Treasury securities having a maturity equal to the Remaining Average Life of such Called Principal as of such Settlement Date, or (ii) if such yields are not reported as of such time or the yields reported as of such time are not ascertainable, the Treasury Constant Maturity Series Yields reported, for the latest day for which such yields have been so reported as of the second Business Day preceding the Settlement Date with respect to such Called Principal, in Federal Reserve Statistical Release H.15 (519) (or any comparable successor publication) for actively traded U.S. Treasury securities having a constant maturity equal to the Remaining Average Life of such Settlement Date. Such implied yield will be determined, if necessary, by (a) converting U.S. Treasury bill quotations to bond-equivalent yields in accordance with accepted financial practice and (b) interpolating linearly between (1) the actively traded U.S. Treasury security with the duration closest to and greater than the Remaining Average Life and (2) the actively traded U.S. Treasury security with the duration closest to and greater than the Remaining Average Life. The Reinvestment Yield will be rounded to the number of decimals as appears in the coupon for the applicable Series of Notes.

"Remaining Average Life" means, with respect to any Called Principal of either Series of Notes, the number of years (calculated to the nearest one-twelfth year) obtained by dividing (i) such Called Principal into (ii) the sum of the products obtained by multiplying (a) the principal component of each Remaining Scheduled Payment with respect to such Called Principal by (b) the number of years (calculated to the nearest one-twelfth year) that will elapse between the Settlement Date with respect to such Called Principal and the scheduled due date of such Remaining Scheduled Payment.

"Remaining Scheduled Payments" means, with respect to the Called Principal of any Note of either Series, all payments of such Called Principal and interest thereon that would be due after the Settlement Date with respect to such Called Principal if no payment of such Called Principal were made prior to its scheduled due date, <u>provided</u> that if such Settlement Date is not a date on which interest payments are due to be made under the terms of the Notes of such Series, then the amount of the next succeeding scheduled interest payment will be reduced by the amount of interest accrued to such Settlement Date and required to be paid on such Settlement Date pursuant to Section 8.2 or Section 12.1.

"Settlement Date" means, with respect to the Called Principal of any Note of either Series, the date on which such Called Principal is to be prepaid pursuant to Section 8.2 or has become or is declared to be immediately due and payable pursuant to Section 12.1, as the context requires.

9. AFFIRMATIVE COVENANTS.

The Company covenants that so long as any of the Notes are outstanding:

9.1. Compliance with Law.

The Company will and will cause each of its Subsidiaries to comply with all laws, ordinances or governmental rules or regulations to which each of them is subject, including, without limitation, Environmental Laws, and will obtain and maintain in effect all licenses, certificates, permits, franchises and other governmental authorizations necessary to the ownership of their respective properties or to the conduct of their respective businesses, in each case to the extent necessary to ensure that non-compliance with such laws, ordinances or governmental rules or regulations or failures to obtain or maintain in effect such licenses, certificates, permits, franchises and other governmental authorizations would not reasonably be expected, individually or in the aggregate, to have a materially adverse effect on the business, operations, affairs, financial condition, properties or assets of the Company and its Subsidiaries taken as a whole.

9.2. Insurance.

The Company will and will cause each of its Subsidiaries to maintain, with financially sound and reputable insurers, insurance with respect to their respective properties and businesses against such casualties and contingencies, of such types, on such terms and in such amounts (including deductibles, co-insurance and self- insurance, if adequate reserves are maintained with respect thereto) as is customary in the case of entities of established reputations engaged in the same or a similar business and similarly situated.

9.3. Maintenance of Properties.

The Company will and will cause each of its Subsidiaries to maintain and keep, or cause to be maintained and kept, their respective properties in good repair, working order and condition (other than ordinary wear and tear), so that the business carried on in connection therewith may be properly conducted at all times, provided that this Section 9.3 shall not prevent the Company or any Subsidiary from discontinuing the operation and the maintenance of any of its properties if such discontinuance is desirable in the conduct of its business and the Company has concluded that such discontinuance would not, individually or in the aggregate, have a materially adverse effect on the business, operations, affairs, financial condition, properties or assets of the Company and its Subsidiaries taken as a whole.

9.4. Payment of Taxes.

The Company will and will cause each of its Subsidiaries to file all income tax or similar tax returns required to be filed in any jurisdiction and to pay and discharge all taxes shown to be due and payable on such returns and all other taxes, assessments, governmental charges, or levies payable by any of them, to the extent such taxes and assessments have become due and payable and before they have become delinquent, <u>provided</u> that neither the Company nor any Subsidiary need pay any such tax or assessment if (a) the amount, applicability or validity thereof is contested by the Company or such Subsidiary on a timely basis in good faith and in appropriate

proceedings, and the Company or a Subsidiary has established adequate reserves therefor in accordance with GAAP on the books of the Company or such Subsidiary or (b) the nonpayment of all such taxes and assessments in the aggregate would not reasonably be expected to have a materially adverse effect on the business, operations, affairs, financial condition, properties or assets of the Company and its Subsidiaries taken as a whole.

9.5. Corporate Existence, etc.

The Company will at all times preserve and keep in full force and effect its corporate existence. Except as provided in Section 10.2 and Section 10.3, the Company will at all times preserve and keep in full force and effect the corporate existence of each of its Subsidiaries (unless merged into the Company or a Subsidiary) and all rights and franchises of the Company and its Subsidiaries unless, in the good faith judgment of the Company, the termination of or failure to preserve and keep in full force and effect such corporate existence, right or franchise would not, individually or in the aggregate, have a materially adverse effect on the business, operations, affairs, financial condition, properties or assets of the Company and its Subsidiaries taken as a whole.

9.6. Subsidiary Guarantors.

(a) The Company will cause each Subsidiary which has provided or does provide a Guaranty of the Indebtedness of the Company or any Subsidiary under a Bank Credit Agreement (regardless of whether it has previously been discharged from any such Guaranty and becomes liable under a new Guaranty) to become a Subsidiary Guarantor on a joint and several basis with all other Subsidiary Guarantors under the Subsidiary Guaranty as promptly as practicable after (but in any event within 30 days of) the date such Subsidiary first guaranties such Indebtedness, by causing such Subsidiary to execute and deliver to the holders of the Notes a joinder agreement to the Subsidiary Guaranty in the form attached to the Subsidiary Guaranty, together with all documents and opinions which the Required Holders may reasonably request relating to the existence of such Subsidiary, the corporate or other authority for and the validity of the Subsidiary Guaranty, and any other matters reasonably determined by the Required Holders to be relevant thereto, all in form and substance reasonably satisfactory to the Required Holders.

(b) The holders of the Notes agree that if the obligations of a Subsidiary Guarantor, whether direct or indirect, as a co-borrower, guarantor or otherwise, of any Indebtedness of the Company or its Subsidiaries under any Bank Credit Agreement shall, at any time after the First Closing Date, be terminated by the holders of such Indebtedness, the holders of the Notes shall, within 30 days of receipt of a written request of the Company, take such action and execute such documents as the Company or such Subsidiary shall reasonably request to give effect to the termination, release and discharge of such Subsidiary's obligations under the Subsidiary Guaranty so long as no Default or Event of Default is continuing;; *provided, however*, that such Subsidiary Guarantor shall not be released from its obligations as a Subsidiary Guarantor if in connection with the release of such Subsidiary Guarantor from its obligations under its Guaranty of the Indebtedness of the Company or any Subsidiaries under any Bank Credit

Agreement, the Company or any of its Subsidiaries pays any consideration to the holders of such Indebtedness in consideration of such release, unless the holders of Notes are paid equivalent consideration for such release; and *provided*, *further*, that in the event any such Subsidiary Guarantor shall at any time after the release provided for in this Section 9.6 enter into a Guaranty of, or otherwise become directly or indirectly liable for (whether by way of becoming a co-borrower, guarantor or otherwise), all or any part of the Indebtedness of the Company or its Subsidiaries under a Bank Credit Agreement, the Company will cause such Subsidiary Guarantor contemporaneously with entering into any such Guaranty or incurring such liability (and in any event within 30 days thereafter) to execute and deliver to the holders of the Notes, (i) a joinder agreement to the Subsidiary Guaranty in accordance with the provisions thereof, and (ii) such opinions of counsel, certificates accompanying authorizing resolutions and corporate or similar documents, and such other agreements, instruments and other documents as the Required Holders may reasonably request, each of foregoing in form and substance reasonably satisfactory to the Required Holders.

10. NEGATIVE COVENANTS.

The Company covenants that so long as any of the Notes are outstanding:

10.1. Transactions with Affiliates.

The Company will not and will not permit any Subsidiary to enter into directly or indirectly any Material transaction or Material group of related transactions (including without limitation the purchase, lease, sale or exchange of properties of any kind or the rendering of any service) with any Affiliate (other than the Company or another Subsidiary), except pursuant to the reasonable requirements of the Company's or such Subsidiary's business and upon fair and reasonable terms no less favorable to the Company or such Subsidiary than would be obtainable in a comparable arm's-length transaction with a Person not an Affiliate.

10.2. Merger, Consolidation, etc.

The Company will not, nor will it permit any Subsidiary Guarantor to, consolidate with or merge with any other Person or convey, transfer or lease all or substantially all of its assets or a controlling equity interest in a Subsidiary Guarantor in a single transaction or series of transactions to any Person (except that a Subsidiary Guarantor may (x) consolidate with or merge with, or convey, transfer or lease all or substantially all of its assets in a single transaction or series of transactions to, the Company or another Subsidiary of the Company so long as in each case, the survivor of such merger or consolidation or the transferee of such assets shall have assumed such Guaranty and (y) convey, transfer or lease all or substantially all of its assets (including any such transaction effected by means of a merger or consolidation of such Subsidiary Guarantor with or into another Person) in compliance with the provisions of Section 10.3), provided that the foregoing restriction does not apply to the consolidation or merger of the Company or a Subsidiary Guarantor with the Company or another Subsidiary Guarantor, or the conveyance, transfer or lease of all or substantially all of the assets of the Company or a Subsidiary Guarantor, or a controlling equity interest in a Subsidiary Guarantor, in a single transaction or series of transactions to any Person so long as:

(a) the successor formed by such consolidation or the survivor of such merger or the Person that acquires by conveyance, transfer or lease all or substantially all of the assets of the Company or such Subsidiary Guarantor, or a controlling equity interest in a Subsidiary Guarantor, as the case may be (the "**Successor Corporation**"), shall be a solvent entity organized and existing under the laws of the United States or any State thereof (including the District of Columbia);

(b) if the Company or a Subsidiary Guarantor is party to such transaction and is not the Successor Corporation, such Person shall (i) have executed and delivered to each holder of Notes its assumption of the due and punctual performance and observance of each covenant and condition of such Obligor, as the case may be, under the applicable Financing Documents in form and substance satisfactory to the Required Holders and (ii) have caused to be delivered to each holder of any Notes an opinion reasonably satisfactory to the Required Holders of nationally recognized independent counsel, or other independent counsel reasonably satisfactory to the Required Holders, to the effect that all agreements or instruments effecting such assumption are enforceable in accordance with their respective terms and comply with the terms hereof;

(c) to the extent the Company is a party and is not the surviving Person of such transaction, each Subsidiary Guarantor shall have executed and delivered to each holder of Notes its reaffirmation of its obligations under the Subsidiary Guaranty in form and substance satisfactory to the Required Holders; and

(d) immediately after giving effect to such transaction, no Default or Event of Default shall have occurred and be continuing, to be determined on a Pro Forma Basis with respect to compliance with Sections 10.5 and 10.6, and the Company shall have delivered to each holder of the Notes computations evidencing compliance with such Sections.

No such conveyance, transfer or lease of all or substantially all of the assets of any Obligor shall have the effect of releasing such Obligor or any Successor Corporation that shall theretofore have become such in the manner prescribed in this Section 10.2 from its liability under the applicable Financing Documents.

10.3. Sale of Assets.

Except as permitted under Section 10.2, the Company will not, and will not permit any Subsidiary to, make any Asset Disposition unless:

(a) in the good faith opinion of the Company, the Asset Disposition is in exchange for consideration having a Fair Market Value at least equal to that of the property exchanged and is in the best interest of the Company or such Subsidiary provided, however, that the Company and its Subsidiaries shall be permitted to engage in Asset Dispositions that are in the best interest of the disposing Person, regardless of whether such Person receives Fair Market Value for the property subject thereto, so long as the aggregate Disposition Value of all property subject to any such Asset Disposition

in any fiscal year is not in excess of 2% of Consolidated Total Assets (determined as of the end of the then most recently ended fiscal year of the Company);

(b) immediately after giving effect to the Asset Disposition, no Default or Event of Default would exist; and

(c) immediately after giving effect to the Asset Disposition, the Disposition Value of all property that was the subject of any Asset Disposition occurring in the then current fiscal year of the Company would not exceed 15% of Consolidated Total Assets (determined as of the end of the then most recently ended fiscal year of the Company).

If an amount equal to the Net Proceeds arising from any Asset Disposition is applied to a Debt Prepayment Application or a Property Reinvestment Application within 365 days after the date of such Asset Disposition, then such Asset Disposition, only for the purpose of determining compliance with subsection (c) of this Section 10.3 as of any date, shall be deemed not to be an Asset Disposition as of the date of such application.

10.4. Limitation on Liens.

The Company will not, and will not permit any Subsidiary to, directly or indirectly create, incur, assume or permit to exist any Lien on or with respect to any property or assets (including, without limitation, any document or instrument in respect of goods or accounts receivable) of the Company or any Subsidiary whether now owned or held or hereafter acquired, or any income or profits therefrom, or assign or otherwise convey any right to receive income or profits except for the following:

(a) Liens for taxes, assessments or other governmental charges which are not yet due and payable or the payment of which is not at the time required by Section 9.4;

(b) Liens incidental to the normal conduct of the business of the Company or any Subsidiary or the ownership of its property which are not incurred in connection with the incurrence of Indebtedness and rights of set-off incurred in the ordinary course of business and which do not, in the aggregate, materially impair the use of such property in the operation of the business of the Company and its Subsidiaries taken as a whole or the value of such property for the purposes of such business (including without limitation, contingent liens arising in favor of insurance carriers under agency agreements with the Company or any Subsidiary;

(c) minor survey exceptions or minor encumbrances which are necessary for the conduct of the activities of the Company and its Subsidiaries or which customarily exist on properties of corporations engaged in similar activities, which do not materially impair their use in operations of the business of the Company and its Subsidiaries;

(d) Liens created by or resulting from any litigation or legal proceeding which is currently being contested in good faith by appropriate proceedings and for which adequate reserves have been made;

(e) Liens on property or assets of the Company or any of its Subsidiaries securing Indebtedness owing to the Company or to a Wholly-Owned Subsidiary;

(f) Liens in existence at on the First Closing Date and securing the Indebtedness of the Company and its Subsidiaries as set forth in Schedule 5.15;

(g) Liens securing any obligations of a Person existing at the time such Person becomes a Subsidiary or is merged into or consolidated with the Company or a Subsidiary or Liens on an asset existing at the time such asset shall have first been acquired by the Company or any Subsidiary, *provided* that (i) such Liens shall not extend to or cover any property other than the property subject to such Liens immediately prior to such time, (ii) such Liens shall not have been created in contemplation of such merger, consolidation or acquisition or such Person becoming a Subsidiary, (iii) the amount of the commitment in respect of the obligations secured by such Liens is not increased after such time and (iv) the principal amount of the obligations secured by any such Lien shall not exceed the Fair Market Value (as determined in good faith by a Responsible Officer of the Company) of such property and any improvements thereon at the time such Person becomes a Subsidiary or is merged into or consolidated with the Company or a Subsidiary or such asset is acquired;

(h) any Lien created on tangible real or personal property (or any improvement thereon) to secure all or any part of the purchase price or cost of construction, improvement or development of such tangible real or personal property (or any improvement thereon), or to secure Indebtedness incurred or assumed to pay all or any part of the purchase price or the cost of construction of tangible real or personal property (or any improvement thereon) acquired or constructed by the Company or any Subsidiary after the First Closing Date, *provided* that

(i) the principal amount of the Indebtedness secured by any such Lien shall at no time exceed an amount equal to the lesser of (A) the cost to the Company or such Subsidiary of the property (or improvement thereon) so acquired or constructed and (B) the Fair Market Value (as determined in good faith by a Responsible Officer of such Person) of such property and any improvements thereon at the time of such acquisition or construction;

(ii) each such Lien shall extend solely to the item or items of property (or improvement thereon) so acquired or constructed and, if required by the terms of the instrument originally creating such Lien, other property (or improvement thereon) which is an improvement to or is acquired for specific use in connection with such acquired or constructed property (or improvement thereon); and

(iii) any such Lien shall be created contemporaneously with, or within 180 days after, the acquisition or construction of such property (or improvement thereon);

(i) any Lien renewing, extending or refunding Liens permitted by paragraphs (g) and (h) of this Section 10.4, *provided* that (i) the amount of the commitment in respect

of the Indebtedness secured by such Lien immediately prior to such renewal, extension or refunding is not increased or the maturity thereof reduced, (ii) such Lien is not extended to any other property, and (iii) immediately after such extension, renewal, or refunding, no Default or Event of Default would exist; and

(j) Liens not otherwise permitted by subsections (a) through (j) above, *provided* that after giving effect to any such Liens Priority Debt will not at any time exceed 20% of Consolidated Net Worth (determined as of the last day of the then most recently ended fiscal quarter of the Company).

10.5. Leverage Ratio.

The Company will not, at any time, permit the ratio of Consolidated Net Indebtedness as of the end of any fiscal quarter of the Company to Consolidated EBITDA, for the period of four consecutive fiscal quarters of the Company ending with and including such fiscal quarter, to be greater than 2.75 to 1.00.

10.6. Fixed Charge Coverage Ratio.

The Company will not permit the Fixed Charge Coverage Ratio, to be less than 1.50 to 1.0 at the end of any fiscal quarter.

10.7. Subsidiary Debt.

The Company will not at any time permit any Subsidiary to, directly or indirectly, create, incur, assume, guarantee, have outstanding, or otherwise become or remain directly or indirectly liable with respect to, any Indebtedness other than:

(a) Indebtedness of a Subsidiary outstanding on the First Closing Date and disclosed in Schedule 5.15 and any extension, renewal or refunding thereof, *provided* that the amount of the commitment in respect of such Indebtedness in effect immediately before giving effect to such extension, renewal or refunding is not increased, the maturity thereof is not reduced and no Default or Event of Default exists at the time of such extension, renewal or refunding;

(b) Indebtedness of a Subsidiary owed to the Company or a Wholly-Owned Subsidiary;

(c) Indebtedness of a Subsidiary outstanding at the time such Subsidiary becomes a Subsidiary, *provided* that (*i*) such Indebtedness shall not have been incurred in contemplation of such Subsidiary becoming a Subsidiary, (*ii*) immediately after such Person becomes a Subsidiary no Default or Event of Default shall exist, and (iii) such Indebtedness shall cease to be permitted under this clause (c) to the extent that such Indebtedness remains Indebtedness of a Subsidiary on the 365th day after such Person became a Subsidiary, and such Indebtedness may be extended, renewed or refunded if immediately after such extension, renewal or refunding no Default or Event of Default would exist but shall cease to be permitted under this clause (c) on the 365th day after such Person becomes a Subsidiary; and

(d) Indebtedness of a Subsidiary in addition to that otherwise permitted by the foregoing provisions of this Section 10.7, *provided* that on the date the Subsidiary incurs or otherwise becomes liable with respect to any such additional Indebtedness and immediately after giving effect thereto and the concurrent retirement of any other Indebtedness, no Default or Event of Default exists.

10.8. Restrictions on Dividends of Subsidiaries, etc.

The Company will not, and will not permit any of its Subsidiaries to, directly or indirectly, enter into or permit to exist any agreement or other arrangements that would prohibit, restrict or impose any condition upon such Subsidiary's ability or right to pay dividends or other distributions to, or make advances to or Investments in, the Company, or, if such Subsidiary is not directly owned by the Company, the "parent" Subsidiary of such Subsidiary; provided that (x) the foregoing shall not apply to restrictions and conditions imposed by law or by this Agreement and (y) the foregoing shall not apply to customary restrictions and conditions contained in agreements relating to the sale of a Subsidiary pending such sale, provided such restrictions and conditions apply only to the Subsidiary that is to be sold and such sale is permitted under the terms of this Agreement.

10.9. Limitation on Priority Debt.

The Company will not permit Priority Debt, determined at any time, to exceed 20% of Consolidated Net Worth, determined as of the last day of the then most recently ended fiscal quarter of the Company.

10.10. No Limitation on Prepayments or Amendments to Certain Financing Documents.

The Company will not, nor will it permit any Subsidiary to, be a party to any agreement or instrument limiting its rights (a) to make payments or prepayments on the Notes, whether optional or mandatory, under this Agreement, or (b) to amend or waive any term or provision of this Agreement, the Notes or the Subsidiary Guaranty.

10.11. Line of Business.

The Company will not, and will not permit any of its Subsidiaries to, engage to any substantial extent in any business other than the businesses in which the Company and its Subsidiaries are engaged on the date of this Agreement as described in the Memorandum and businesses reasonably related thereto, in furtherance thereof or ancillary thereto.

11. EVENTS OF DEFAULT.

An "Event of Default" shall exist if any of the following conditions or events shall occur and be continuing:

(a) the Company defaults in the payment of any principal or Make-Whole Amount, if any, on any Note when the same becomes due and payable, whether at maturity or at a date fixed for prepayment or by declaration or otherwise; or

(b) the Company defaults in the payment of any interest on any Note for more than five Business Days after the same becomes due and payable; or

(c) the Company defaults in the performance of or compliance with any term contained in Section 7.1(d) or Section 10; or

(d) the Company defaults in the performance of or compliance with any term contained herein (other than those terms referred to in paragraphs (a), (b) and (c) of this Section 11) and such default is not remedied within 30 days after the earlier of (i) a Responsible Officer obtaining actual knowledge of such default and (ii) the Company receiving written notice of such default from any holder of a Note (any such written notice to be identified as a "notice of default" and to refer specifically to this paragraph (d) of Section 11); or

(e) any representation or warranty made in writing by or on behalf of any Obligor or by any officer of such Obligor in any Financing Document or in any writing furnished in connection with the transactions contemplated hereby proves to have been false or incorrect in any Material respect on the date as of which made; or

(f) (i) the Company or any Subsidiary is in default (as principal or as guarantor or other surety) in the payment of any principal of or premium or make-whole amount or interest on any Indebtedness that is outstanding in an aggregate principal amount of at least \$25,000,000 beyond any period of grace provided with respect thereto, or (ii) the Company or any Subsidiary is in default in the performance of or compliance with any term of any evidence of any Indebtedness in an aggregate outstanding principal amount of at least \$25,000,000 or of any mortgage, indenture or other agreement relating thereto or any other condition exists, and as a consequence of such default or condition such Indebtedness has become, or has been declared, due and payable before its stated maturity or before its regularly scheduled dates of payment, or (iii) as a consequence of the occurrence or continuation of any event or condition (other than the passage of time or the right of the holder of Indebtedness to convert such Indebtedness into equity interests), (x) the Company or any Subsidiary has become obligated to purchase or repay Indebtedness before its regular maturity or before its regularly scheduled dates of payment in an aggregate outstanding principal amount of at least \$25,000,000, or (y) one or more Persons have required that the Company or any Subsidiary purchase or repay such Indebtedness; or

(g) the Company or any Significant Subsidiary (i) is generally not paying, or admits in writing its inability to pay, its debts as they become due, (ii) files, or consents by answer or otherwise to the filing against it of, a petition for relief or reorganization or arrangement or any other petition in bankruptcy, for liquidation or to take advantage of any bankruptcy, insolvency, reorganization, moratorium or other similar law of any jurisdiction, (iii) makes an assignment for the benefit of its creditors, (iv) consents to the appointment of a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, (v) is adjudicated as insolvent or to be liquidated, or (vi) takes corporate action for the purpose of any of the foregoing; or

(h) a court or governmental authority of competent jurisdiction enters an order appointing, without consent by the Company or any of its Significant Subsidiaries, a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, or constituting an order for relief or approving a petition for relief or reorganization or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy or insolvency law of any jurisdiction, or ordering the dissolution, winding-up or liquidation of the Company or any of its Significant Subsidiaries, or any such petition shall be filed against the Company or any of its Significant Subsidiaries and such petition shall not be dismissed within 60 days; or

(i) a final judgment or judgments for the payment of money aggregating in excess of \$25,000,000 are rendered against one or more of the Company and its Significant Subsidiaries and which judgments are not, within 60 days after entry thereof, bonded, discharged or stayed pending appeal, or are not discharged within 60 days after the expiration of such stay; or

(j) any Subsidiary Guarantor fails or neglects in any material respect to observe, perform or comply with any term, provision or covenant contained in the Subsidiary Guaranty; or

(k) the Subsidiary Guaranty is not or ceases to be effective against any Subsidiary Guarantor or is alleged by any Obligor to be ineffective against any Subsidiary Guarantor for any reason unless, in each case, the Company is in compliance with Section 9.6 without such Subsidiary being a Subsidiary Guarantor; or

(l) if (i) any Plan shall fail to satisfy the minimum funding standards of ERISA or the Code for any plan year or part thereof or a waiver of such standards or extension of any amortization period is sought or granted under section 412 of the Code, (ii) a notice of intent to terminate any Plan shall have been or is reasonably expected to be filed with the PBGC or the PBGC shall have instituted proceedings under ERISA section 4042 to terminate or appoint a trustee to administer any Plan or the PBGC shall have notified the Company or any ERISA Affiliate that a Plan may become a subject of any such proceedings, (iii) the aggregate "amount of unfunded benefit liabilities" (within the meaning of section 4001(a)(18) of ERISA) under all Plans, determined in accordance with Title IV of ERISA, shall exceed \$25,000,000, (iv) the Company or any ERISA Affiliate shall have incurred or is reasonably expected to incur any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, (v) the Company or any ERISA Affiliate withdraws from any Multiemployer Plan, or (vi) the Company or any Subsidiary establishes or amends any employee welfare benefit plan that provides post-employment welfare benefits in a manner that would increase the liability of the Company or any Subsidiary thereunder; and any such event or events described in clauses (i) through (vi) above, either individually or together with any other such event or events, would reasonably be expected to have a Materially Adverse Effect.

As used in Section 11(l), the terms **"employee benefit plan"** and **"employee welfare benefit plan"** shall have the respective meanings assigned to such terms in Section 3 of ERISA.

12. REMEDIES ON DEFAULT, ETC.

12.1. Acceleration.

(a) If an Event of Default with respect to the Company described in Section 11(g) or 11(h) (other than an Event of Default described in clause (i) of paragraph (g) or described in clause (vi) of paragraph (g) by virtue of the fact that such clause encompasses clause (i) of paragraph (g)) has occurred, all the Notes then outstanding shall automatically become immediately due and payable.

(b) If any other Event of Default has occurred and is continuing, the Required Holders may at any time at its or their option, by notice or notices to the Company, declare all the Notes then outstanding to be immediately due and payable.

(c) If any Event of Default described in paragraph (a) or (b) of Section 11 has occurred and is continuing, any holder or holders of Notes at the time outstanding affected by such Event of Default may at any time, at its or their option, by notice or notices to the Company, declare all the Notes held by it or them to be immediately due and payable.

Upon any Notes becoming due and payable under this Section 12.1, whether automatically or by declaration, such Notes will forthwith mature and the entire unpaid principal amount of such Notes, plus (x) all accrued and unpaid interest thereon and (y) the Make-Whole Amount determined in respect of such principal amount (to the full extent permitted by applicable law), shall all be immediately due and payable, in each and every case without presentment, demand, protest or further notice, all of which are hereby waived. The Company acknowledges, and the parties hereto agree, that each holder of a Note has the right to maintain its investment in the Notes free from repayment by the Company (except as herein specifically provided for) and that the provision for payment of a Make-Whole Amount by the Company in the event that the Notes are prepaid or are accelerated as a result of an Event of Default, is intended to provide compensation for the deprivation of such right under such circumstances.

12.2. Other Remedies.

If any Default or Event of Default has occurred and is continuing, and irrespective of whether any Notes have become or have been declared immediately due and payable under Section 12.1, the holder of any Note at the time outstanding may proceed to protect and enforce the rights of such holder by an action at law, suit in equity or other appropriate proceeding, whether for the specific performance of any agreement contained herein or in any Note, or for an injunction against a violation of any of the terms hereof or thereof, or in aid of the exercise of any power granted hereby or thereby or by law or otherwise.

12.3. Rescission.

At any time after any Notes have been declared due and payable pursuant to clause (b) or (c) of Section 12.1, the Required Holders, by written notice to the Company, may rescind and annul any such declaration and its consequences if (a) the Company has paid all overdue interest on the Notes, all principal of and Make-Whole Amount, if any, on any Notes that are due and

payable and are unpaid other than by reason of such declaration, and all interest on such overdue principal and Make-Whole Amount, if any, and (to the extent permitted by applicable law) any overdue interest in respect of either Series of the Notes, at the Default Rate for such Series, (b) all Events of Default and Defaults, other than non-payment of amounts that have become due solely by reason of such declaration, have been cured or have been waived pursuant to Section 17, and (c) no judgment or decree has been entered for the payment of any monies due pursuant hereto or to the Notes. No rescission and annulment under this Section 12.3 will extend to or affect any subsequent Event of Default or Default or impair any right consequent thereon.

12.4. No Waivers or Election of Remedies, Expenses, etc.

No course of dealing and no delay on the part of any holder of any Note in exercising any right, power or remedy shall operate as a waiver thereof or otherwise prejudice such holder's rights, powers or remedies. No right, power or remedy conferred by this Agreement or by any Note upon any holder thereof shall be exclusive of any other right, power or remedy referred to herein or therein or now or hereafter available at law, in equity, by statute or otherwise. Without limiting the obligations of the Company under Section 15, the Company will pay to the holder of each Note on demand such further amount as shall be sufficient to cover all costs and expenses of such holder incurred in any enforcement or collection under this Section 12, including, without limitation, reasonable attorneys' fees, expenses and disbursements.

13. REGISTRATION; EXCHANGE; SUBSTITUTION OF NOTES.

13.1. Registration of Notes.

The Company shall keep at its principal executive office a register for the registration and registration of transfers of Notes. The name and address of each holder of one or more Notes, each transfer thereof and the name and address of each transferee of one or more Notes shall be registered in such register. Prior to due presentment for registration of transfer, the Person in whose name any Note shall be registered shall be deemed and treated as the owner and holder thereof for all purposes hereof, and the Company shall not be affected by any notice or knowledge to the contrary. The Company shall give to any holder of a Note that is an Institutional Investor promptly upon request therefor, a complete and correct copy of the names and addresses of all registered holders of Notes.

13.2. Transfer and Exchange of Notes.

Upon surrender of any Note at the principal executive office of the Company for registration of transfer or exchange (and in the case of a surrender for registration of transfer, duly endorsed or accompanied by a written instrument of transfer duly executed by the registered holder of such Note or its attorney duly authorized in writing and accompanied by the address for notices of each transferee of such Note or part thereof), the Company shall execute and deliver, at the Company's expense (except as provided below), one or more new Notes of the same Series (as requested by the holder thereof) in exchange therefor, in an aggregate principal amount equal to the unpaid principal amount of the surrendered Note. Each such new Note shall be payable to such Person as such holder may request and shall be substantially in the form of Notes for such Series set forth in Exhibit 1(a) or Exhibit 1(b), as the case may be. Each such new Note shall be

dated and bear interest from the date to which interest shall have been paid on the surrendered Note or dated the date of the surrendered Note if no interest shall have been paid thereon. The Company may require payment of a sum sufficient to cover any stamp tax or governmental charge imposed in respect of any such transfer of Notes. Notes shall not be transferred in denominations of less than \$100,000, provided that if necessary to enable the registration of transfer by a holder of its entire holding of Notes, one Note may be in a denomination of less than \$100,000. Any transferee, by its acceptance of a Note registered in its name (or the name of its nominee), shall be deemed to have made the representation set forth in Section 6.2.

13.3. Replacement of Notes.

Upon receipt by the Company of evidence reasonably satisfactory to it of the ownership of and the loss, theft, destruction or mutilation of any Note (which evidence shall be, in the case of an Institutional Investor, notice from such Institutional Investor of such ownership and such loss, theft, destruction or mutilation), and

(a) in the case of loss, theft or destruction, of indemnity reasonably satisfactory to it (<u>provided</u> that if the holder of such Note is, or is a nominee for, an original Purchaser or another holder of a Note with a minimum net worth of at least \$100,000,000, such Person's own unsecured agreement of indemnity shall be deemed to be satisfactory), or

(b) in the case of mutilation, upon surrender and cancellation thereof,

the Company at its own expense shall execute and deliver, in lieu thereof, a new Note of the same Series, dated and bearing interest from the date to which interest shall have been paid on such lost, stolen, destroyed or mutilated Note or dated the date of such lost, stolen, destroyed or mutilated Note if no interest shall have been paid thereon.

14. PAYMENTS ON NOTES.

14.1. Place of Payment.

Subject to Section 14.2, payments of principal, Make-Whole Amount, if any, and interest becoming due and payable on the Notes shall be made in Daytona Beach, Florida at the principal office of the Company in such jurisdiction. The Company may at any time, by notice to each holder of a Note, change the place of payment of the Notes so long as such place of payment shall be either the principal office of the Company in such jurisdiction or the principal office of a bank or trust company in such jurisdiction.

14.2. Home Office Payment.

So long as any Purchaser or its nominee shall be the holder of any Note, and notwithstanding anything contained in Section 14.1 or in such Note to the contrary, the Company will pay all sums becoming due on such Note for principal, Make-Whole Amount, if any, and interest by the method and at the address specified for such purpose below such Purchaser's name in Schedule A, or by such other method or at such other address as such Purchaser shall have from time to time specified to the Company in writing for such purpose, without the

presentation or surrender of such Note or the making of any notation thereon, except that upon written request of the Company made concurrently with or reasonably promptly after payment or prepayment in full of any Note, such Purchaser shall surrender such Note for cancellation, reasonably promptly after any such request, to the Company at its principal executive office or at the place of payment most recently designated by the Company pursuant to Section 14.1. Prior to any sale or other disposition of any Note held by any Purchaser or its nominee such Purchaser will, at its election, either endorse thereon the amount of principal paid thereon and the last date to which interest has been paid thereon or surrender such Note to the Company in exchange for a new Note or Notes pursuant to Section 13.2. The Company will afford the benefits of this Section 14.2 to any Institutional Investor that is the direct or indirect transferee of any Note purchased by such Purchaser under this Agreement and that has made the same agreement relating to such Note as such Purchaser has made in this Section 14.2.

15. EXPENSES, ETC.

15.1. Transaction Expenses.

Whether or not the transactions contemplated hereby are consummated, the Company will pay all costs and expenses (including reasonable attorneys' fees of a special counsel and, if reasonably required, local or other counsel) incurred by each Purchaser or holder of a Note in connection with such transactions and in connection with any amendments, waivers or consents under or in respect of this Agreement, the Notes or the Subsidiary Guaranty (whether or not such amendment, waiver or consent becomes effective), including, without limitation: (a) the costs and expenses incurred in enforcing or defending (or determining whether or how to enforce or defend) any rights under this Agreement, the Notes or the Subsidiary Guaranty or in responding to any subpoena or other legal process or informal investigative demand issued in connection with this Agreement, the Notes or the Subsidiary Guaranty, or by reason of being a holder of any Note, and (b) the costs and expenses, including reasonable financial advisors' fees, incurred in connection with the insolvency or bankruptcy of the Company or any Subsidiary or in connection with any work-out or restructuring of the transactions contemplated hereby and by the Notes. The Company will pay, and will save each Purchaser and each other holder of a Note harmless from, all claims in respect of any fees, costs or expenses if any, of brokers and finders (other than those retained by any Purchaser).

15.2. Survival.

The obligations of the Company under this Section 15 will survive the payment or transfer of any Note, the enforcement, amendment or waiver of any provision of this Agreement or the Notes, and the termination of this Agreement.

16. SURVIVAL OF REPRESENTATIONS AND WARRANTIES; ENTIRE AGREEMENT.

All representations and warranties contained herein shall survive the execution and delivery of this Agreement and the Notes, the purchase or transfer by any Purchaser of any Note or portion thereof or interest therein and the payment of any Note, and may be relied upon by any subsequent holder of a Note, regardless of any investigation made at any time by or on

behalf of such Purchaser or any other holder of a Note. All statements contained in any certificate or other instrument delivered by or on behalf of the Company pursuant to this Agreement shall be deemed representations and warranties of the Company under this Agreement. Subject to the preceding sentence, this Agreement and the Notes embody the entire agreement and understanding between each Purchaser and the Company and supersede all prior agreements and understandings relating to the subject matter hereof.

17. AMENDMENT AND WAIVER.

17.1. Requirements.

This Agreement and the Notes may be amended, and the observance of any term hereof or of the Notes may be waived (either retroactively or prospectively), with (and only with) the written consent of the Company and the Required Holders, except that (a) no amendment or waiver of any of the provisions of Sections 1, 2, 3, 4, 5, 6 or 21 hereof, or any defined term (as it is used therein), will be effective as to any holder unless consented to by such holder in writing, and (b) no such amendment or waiver may, without the written consent of the holder of each Note at the time outstanding affected thereby, (i) subject to the provisions of Section 12 relating to acceleration or rescission, change the amount or time of any prepayment or payment of principal of, or reduce the rate or change the time of payment or method of computation of interest or of the Make-Whole Amount on, either Series of the Notes, (ii) change the percentage of the principal amount of the Notes the holders of which are required to consent to any such amendment or waiver, or (iii) amend any of Sections 8, 11(a), 11(b), 12, 17 or 20.

17.2. Solicitation of Holders of Notes.

(a) <u>Solicitation</u>. The Company will provide each holder of the Notes (irrespective of the amount of Notes then owned by it) with sufficient information, sufficiently far in advance of the date a decision is required, to enable such holder to make an informed and considered decision with respect to any proposed amendment, waiver or consent in respect of any of the provisions hereof or of the Notes. The Company will deliver executed or true and correct copies of each amendment, waiver or consent effected pursuant to the provisions of this Section 17 to each holder of outstanding Notes promptly following the date on which it is executed and delivered by, or receives the consent or approval of, the requisite holders of Notes.

(b) <u>Payment</u>. The Company will not directly or indirectly pay or cause to be paid any remuneration, whether by way of supplemental or additional interest, fee or otherwise, or grant any security, to any holder of Notes as consideration for or as an inducement to the entering into by any holder of Notes of any waiver or amendment of any of the terms and provisions hereof unless such remuneration is concurrently paid, or security is concurrently granted, on the same terms, ratably to each holder of Notes then outstanding even if such holder did not consent to such waiver or amendment.

17.3. Binding Effect, etc.

Any amendment or waiver consented to as provided in this Section 17 applies equally to all holders of Notes and is binding upon them and upon each future holder of any Note and upon

the Company without regard to whether such Note has been marked to indicate such amendment or waiver. No such amendment or waiver will extend to or affect any obligation, covenant, agreement, Default or Event of Default not expressly amended or waived or impair any right consequent thereon. No course of dealing between the Company and the holder of any Note nor any delay in exercising any rights hereunder or under any Note shall operate as a waiver of any rights of any holder of such Note. As used herein, the term **"this Agreement"** and references thereto shall mean this Agreement as it may from time to time be amended or supplemented.

17.4. Notes held by Company, etc.

Solely for the purpose of determining whether the holders of the requisite percentage of the aggregate principal amount of Notes then outstanding approved or consented to any amendment, waiver or consent to be given under this Agreement or the Notes, or have directed the taking of any action provided herein or in the Notes to be taken upon the direction of the holders of a specified percentage of the aggregate principal amount of Notes then outstanding, Notes directly or indirectly owned by the Company or any of its Affiliates shall be deemed not to be outstanding.

18. NOTICES.

All notices and communications provided for hereunder shall be in writing and sent (a) by telecopy if the sender on the same day sends a confirming copy of such notice by a recognized overnight delivery service (charges prepaid), or (b) by registered or certified mail with return receipt requested (postage prepaid), or (c) by a recognized overnight delivery service (with charges prepaid). Any such notice must be sent:

(i) if to any Purchaser or its nominee, to such Purchaser or its nominee at the address specified for such communications in Schedule A, or at such other address as such Purchaser or its nominee shall have specified to the Company in writing,

(ii) if to any other holder of any Note, to such holder at such address as such other holder shall have specified to the Company in writing, or

(iii) if to the Company, to the Company at its address set forth at the beginning hereof to the attention of Cory T. Walker, with a copy to the attention of Laurel Grammig, or at such other address as the Company shall have specified to the holder of each Note in writing.

Notices under this Section 18 will be deemed given only when actually received.

19. REPRODUCTION OF DOCUMENTS.

This Agreement and all documents relating thereto, including, without limitation, (a) consents, waivers and modifications that may hereafter be executed, (b) documents received by any Purchaser at each Closing (except the Notes themselves), and (c) financial statements, certificates and other information previously or hereafter furnished to the holders of Notes, may be reproduced by the holders of the Notes by any photographic, photostatic, microfilm,

microcard, miniature photographic or other similar process and the holders of Notes may destroy any original document so reproduced. The Company agrees and stipulates that, to the extent permitted by applicable law, any such reproduction shall be admissible in evidence as the original itself in any judicial or administrative proceeding (whether or not the original is in existence and whether or not such reproduction was made by such holder in the regular course of business) and any enlargement, facsimile or further reproduction of such reproduction shall likewise be admissible in evidence. This Section 19 shall not prohibit the Company or any other holder of Notes from contesting any such reproduction to the same extent that it could contest the original, or from introducing evidence to demonstrate the inaccuracy of any such reproduction.

20. CONFIDENTIAL INFORMATION.

For the purposes of this Section 20, "Confidential Information" means information delivered to any Purchaser or holder or nominee by or on behalf of the Company or any Subsidiary in connection with the transactions contemplated by or otherwise pursuant to this Agreement that is proprietary in nature and that was clearly marked or labeled or otherwise adequately identified when received by such Purchaser as being confidential information of the Company or such Subsidiary, provided that such term does not include information that (a) was publicly known or otherwise known to such Purchaser prior to the time of such disclosure, (b) subsequently becomes publicly known through no act or omission by such Purchaser or any person acting on its behalf, (c) otherwise becomes known to such Purchaser other than through disclosure by the Company or any Subsidiary or by any Person known by such Purchaser as owing a duty of confidentiality with respect to such information to the Company or any Subsidiary, or (d) constitutes financial statements delivered to such Purchaser under Section 7.1 that are otherwise publicly available. Each Purchaser will maintain the confidentiality of such Confidential Information in accordance with procedures adopted by such Purchaser in good faith to protect confidential information of third parties delivered to such Purchaser, provided that such Purchaser may deliver or disclose Confidential Information to (i) its directors, trustees, officers, employees, agents, attorneys and affiliates (to the extent such disclosure reasonably relates to the administration of the investment represented by such Purchaser's Notes), except that the permission set forth in this clause (i) shall not extend to any such person engaged in the design, sale, underwriting or management of insurance products, (ii) its financial advisors and other professional advisors who agree to hold confidential the Confidential Information substantially in accordance with the terms of this Section 20, (iii) any other holder of any Note, (iv) any Institutional Investor to which such Purchaser sells or offers to sell such Note or any part thereof or any participation therein (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 20), (v) any Person from which such Purchaser offers to purchase any security of the Company (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 20), (vi) any federal or state regulatory authority having jurisdiction over such Purchaser, (vii) the National Association of Insurance Commissioners or any similar organization, or any nationally recognized rating agency that requires access to information about such Purchaser's investment portfolio, or (viii) any other Person to which such delivery or disclosure may be necessary or appropriate (w) to effect compliance with any law, rule, regulation or order applicable to such Purchaser, (x) in response to any subpoena or other legal process, (y) in connection with any litigation to which such Purchaser is a party or (z) if an Event of Default has occurred and is continuing, to the extent such Purchaser may reasonably

determine such delivery and disclosure to be necessary or appropriate in the enforcement or for the protection of its rights and remedies under its Notes and this Agreement. Each holder of a Note, by its acceptance of a Note, will be deemed to have agreed to be bound by and to be entitled to the benefits of this Section 20 as though it were a party to this Agreement. On reasonable request by the Company in connection with the delivery to any holder of a Note of information required to be delivered to such holder under this Agreement or requested by such holder (other than a holder that is a party to this Agreement or its nominee), such holder will enter into an agreement with the Company embodying the provisions of this Section 20.

21. SUBSTITUTION OF PURCHASER.

Each Purchaser shall have the right to substitute any one of its Affiliates as the purchaser of the Notes that such Purchaser has agreed to purchase hereunder, by written notice to the Company, which notice shall be signed by both such Purchaser and such Affiliate, shall contain such Affiliate's agreement to be bound by this Agreement and shall contain a confirmation by such Affiliate of the accuracy with respect to it of the representations set forth in Section 6. Upon receipt of such notice, wherever the word "Purchaser" is used in this Agreement (other than in this Section 21), such word shall be deemed to refer to such Affiliate in lieu of such original Purchaser. In the event that such Affiliate is so substituted as a purchaser hereunder and such Affiliate thereafter transfers to such Purchaser all of the Notes then held by such Affiliate, upon receipt by the Company of notice of such transfer, wherever the word "Purchaser" is used in this Agreement (other than in this Section 21), such word shall no longer be deemed to refer to such Affiliate, but shall refer to such Purchaser, and such Purchaser shall have all the rights of an original holder of the Notes under this Agreement.

22. MISCELLANEOUS.

22.1. Successors and Assigns.

All covenants and other agreements contained in this Agreement by or on behalf of any of the parties hereto bind and inure to the benefit of their respective successors and assigns (including, without limitation, any subsequent holder of a Note) whether so expressed or not.

22.2. Payments Due on Non-Business Days.

Anything in this Agreement or the Notes to the contrary notwithstanding, any payment of principal of or Make-Whole Amount or interest on any Note that is due on a date other than a Business Day shall be made on the next succeeding Business Day without including the additional days elapsed in the computation of the interest payable on such next succeeding Business Day.

22.3. Severability.

Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall (to the full extent permitted by law) not invalidate or render unenforceable such provision in any other jurisdiction.

22.4. Construction.

Each covenant contained herein shall be construed (absent express provision to the contrary) as being independent of each other covenant contained herein, so that compliance with any one covenant shall not (absent such an express contrary provision) be deemed to excuse compliance with any other covenant. Where any provision herein refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether such action is taken directly or indirectly by such Person.

22.5. Counterparts.

This Agreement may be executed in any number of counterparts, each of which shall be an original but all of which together shall constitute one instrument. Each counterpart may consist of a number of copies hereof, each signed by less than all, but together signed by all, of the parties hereto.

22.6. Governing Law.

THIS AGREEMENT SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, AND THE RIGHTS OF THE PARTIES SHALL BE GOVERNED BY, THE LAW OF THE STATE OF NEW YORK EXCLUDING CHOICE-OF-LAW PRINCIPLES OF THE LAW OF SUCH STATE THAT WOULD REQUIRE THE APPLICATION OF THE LAWS OF A JURISDICTION OTHER THAN SUCH STATE.

[Remainder of Page Intentionally Left Blank; Next Page is Signature Page]

If each Purchaser is in agreement with the foregoing, please sign the form of agreement on the accompanying counterpart of this Agreement and return it to the Company, whereupon the foregoing shall become a binding agreement between the Purchasers and the Company.

Very truly yours,

BROWN & BROWN, INC.

By: /S/ CORY T. WALKER

Name: Cory T. Walker Title: Treasurer

Schedule A-1

THE VARIABLE ANNUITY LIFE INSURANCE COMPANY

THE UNITED STATES LIFE INSURANCE COMPANY IN THE CITY OF NEW YORK

AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY

OF NEW YORK

AIG LIFE INSURANCE COMPANY

BY: AIG Global Investment Corp., investment adviser

By: /S/ GERALD F. HERMAN

Name: Gerald F. Herman Title: Vice President

Schedule A-2

NEW YORK LIFE INSURANCE COMPANY

By: /S/ POST HOWLAND

Name: A. Post Howland

Title: Investment Vice President

NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION

By: New York Life Investment Management, LLC, Its Investment Manager

By: /S/ POST HOWLAND

Name: A. Post Howland Title: Director

NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION INSTITUTIONALLY OWNED LIFE INSURANCE SEPARATE ACCOUNT

By: New York Life Investment Management LLC, Its Investment Manager

By: /S/ POST HOWLAND

Name: A. Post Howland Title: Director

CONNECTICUT GENERAL LIFE INSURANCE COMPANY

By: CIGNA Investments, Inc.

By: /S/ DEBRA J. HEIGHT

Name: Debra J. Height Title: Managing Director

CIGNA LIFE INSURANCE COMPANY OF NEW YORK

By: CIGNA Investments, Inc.

By: /S/ DEBRA J. HEIGHT

Name: Debra J. Height Title: Managing Director

LIFE INSURANCE COMPANY OF NORTH AMERICA

By: CIGNA Investments, Inc.

By: /S/ DEBRA J. HEIGHT

Name: Debra J. Height Title: Managing Director

ALLSTATE LIFE INSURANCE COMPANY

By: /S/ ROBERT BODETT

Name: ROBERT B. BODETT

By: /S/ JERRY D. ZINKULA

Name: JERRY D. ZINKULA Authorized Signatories

ALLSTATE INSURANCE COMPANY

By: /S/ ROBERT BODETT

Name: ROBERT B. BODETT

By: /S/ JERRY D. ZINKULA

Name: JERRY D. ZINKULA Authorized Signatories

UNITED OF OMAHA LIFE INSURANCE COMPANY

By: /S/ EDWIN H. GARRISON, JR.

Name: Edwin H. Garrison, Jr. Title: First Vice President

MUTUAL OF OMAHA INSURANCE COMPANY

By: /S/ EDWIN H. GARRISON, JR.

Name: Edwin H. Garrison, Jr. Title: First Vice President

COMPANION LIFE INSURANCE COMPANY

By: /S/ EDWIN H. GARRISON, JR.

Name: Edwin H. Garrison, Jr. Title: Authorized Representative

THE PRUDENTIAL INSURANCE COMPANY OF AMERICA

By: /S/ BILLY GREER

Name: Billy B. Greer Title: Vice President

RGA REINSURANCE COMPANY

- By: Prudential Private Placement Investors, L.P. (as Investment Advisor)
- By: Prudential Private Placement Investors, Inc. (as its General Partner)

By: /S/ BILLY GREER

Name: Billy B. Greer Title: Vice President

AMERICAN BANKERS INSURANCE COMPANY OF FLORIDA, INC.

- By: Prudential Private Placement Investors, L.P. (as Investment Advisor)
- By: Prudential Private Placement Investors, Inc. (as its General Partner)
- By: /S/ BILLY GREER

Name: Billy B. Greer Title: Vice President

AMERICAN MEMORIAL LIFE INSURANCE COMPANY

- By: Prudential Private Placement Investors, L.P. (as Investment Advisor)
- By: Prudential Private Placement Investors, Inc. (as its General Partner)

By: /S/ BILLY GREER

Name: Billy B. Greer Title: Vice President

FORTIS INSURANCE COMPANY

- By: Prudential Private Placement Investors, L.P. (as Investment Advisor)
- By: Prudential Private Placement Investors, Inc. (as its General Partner)

By: /S/ BILLY GREER

Name: Billy B. Greer Title: Vice President

JOHN ALDEN LIFE INSURANCE COMPANY

- By: Prudential Private Placement Investors, L.P. (as Investment Advisor)
- By: Prudential Private Placement Investors, Inc. (as its General Partner)

By: /S/ BILLY GREER

Name: Billy B. Greer Title: Vice President

PHOENIX LIFE INSURACNE COMPANY

By: /S/ CAROLE A. MASTERS

Name: Carole A. Masters Title: V.P.

PHL VARIABLE INSURANCE COMPANY

By: /S/ BENITO J. CUEVO

Name: Benito J. Cuevo Title: Second V.P.

LIFE INSURANCE COMPANY OF THE SOUTHWEST

By: /S/ SCOTT HIGGINS

Name: R. Scott Higgins Title: Vice President, NL Capital Management

SECURITY FINANCIAL LIFE INSURANCE CO.

By: /S/ KEVIN W. HAMMOND

Name: Kevin W. Hammond Title: Vice President Chief Investment Officer

SCHEDULE A

INFORMATION RELATING TO PURCHASERS

Purchaser Name	THE VARIABLE ANNUITY LIFE INSURANCE COMPANY		
Name in which to register Notes	THE VARIABLE ANNUITY LIFE INSURANCE COMPANY		
Series B Note registration number; principal amount	RB-1; \$32,500,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	The Bank of New York ABA # 021-000-018 BNF Account #: IOC 566 Credit to: The Variable Annuity Life Insurance Company - Private Placement Ref: PPN # and Prin: \$ Int: \$ Re: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and application (as among principal, premium and interest) of the paymen made.		
Address / Fax # for notices related to payments	AIG Global Investment Group c/o The Bank of New York Attn: P & I Department P.O. Box 19266 Newark, NJ 07195 Telephone: (212) 437-6557 Fax: (212) 437-6467		
	Duplicate payment notices to:		
	The Variable Annuity Life Insurance Company - Private Placement c/o AIG Global Investment Group 2929 Allen Parkway, A36-04 Houston, Texas 77019 Attn: Private Placement Department		

Purchaser Name	THE VARIABLE ANNUITY LIFE INSURANCE COMPANY	
Address / Fax # for all other notices	AIG Global Investment Group 2929 Allen Parkway, Suite A36-01 Houston, TX 77019 Legal Department – Investment Management Fax: (713) 831-2328	
	with copies of compliance information to:	
	The Variable Annuity Life Insurance Company - Private Placement c/o AIG Global Investment Group 2929 Allen Parkway, A36-04 Houston, Texas 77019 Attn: Private Placement Department	
Instructions re Delivery of Notes	The Bank of New York One Wall Street - 3 rd Floor Window A New York, NY 10286 Attn: Mr. Arthur Cassidy (212-635-4927) Account Name: The Variable Annuity Life Insurance Company - Private Placement Account Number: 260735	
Signature Block	THE VARIABLE ANNUITY LIFE INSURANCE COMPANY	
	THE UNITED STATES LIFE INSURANCE COMPANY	
	AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY IN THE CITY OF NEW YORK	
	AIG LIFE INSURANCE COMPANY	
	BY: AIG Global Investment Corp., investment adviser	
	By:	
	Name: Gerald F. Herman Title: Vice President	
Tax identification number	74-1625348	
	Schedule A-13	

Purchaser Name	THE UNITED STATES LIFE INSURANCE COMPANY IN THE CITY OF NEW YORK		
Name in which to register Notes	THE UNITED STATES LIFE INSURANCE COMPANY IN THE CITY OF NEW YORK		
Series B Note registration number; principal amount	RB-2; \$7,500,000		
Payment on account of Note			
Method	Federal Funds Wire Tran	isfer	
Account information	State Street Bank and Tr Boston, MA 02101 ABA # 011000028 Re: The United States Li A/C: 6956-534-9 OBI = PPN # and descri Fund Number PA 77 Re: (See "Accompanying	fe Insurance Company in the City of New York ption of payment	
Accompanying information	Name of Issuer: Description of Security: PPN: Due date and application made.	BROWN & BROWN, INC. 6.08% Series B Senior Notes Due July 15, 2014 115236 A@ 0 (as among principal, premium and interest) of the payment being	
Address / Fax # for notices related to payments	The United States Life Insurance Co. in the City of New York and PA 77 c/o State Street Bank Corporation Insurance Services 801 Pennsylvania Kansas City, MO 64105 Fax: (816) 691-3619		
	Duplicate payment notic	es to:	
	The United States Life In c/o AIG Global Investme 2929 Allen Parkway, A3 Houston, Texas 77019 Attn: Private Placement Fax: (713) 831-1072	6-04	
Schedule A-14			

Purchaser Name	THE UNITED STATES LIFE INSURANCE COMPANY IN THE CITY OF NEW YORK		
Address / Fax # for all other notices	AIG Global Investment Group 2929 Allen Parkway, Suite A36-01 Houston, TX 77019 Legal Department – Investment Management Fax: (713) 831-2328		
	with copies of compliance information to:		
	The United States Life Insurance Company in the City of New York and PA77 c/o AIG Global Investment Corporation		
	2929 Allen Parkway, A36-04 Houston, Tayaa 77010		
	Houston, Texas 77019 Attn: Private Placement Department		
	Fax: (713) 831-1072		
Instructions re Delivery of Notes	DTC / New York Window 55 Water Street New York, N.Y. 10041 Attention: Robert Mendez Account: State Street Fund Name: THE UNITED STATES LIFE INSURANCE CO. IN THE CITY OF NEW YORK Fund Number: PA 77		
Signature Block	THE VARIABLE ANNUITY LIFE INSURANCE COMPANY		
	THE UNITED STATES LIFE INSURANCE COMPANY		
	AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY IN THE CITY OF NEW YORK		
	AIG LIFE INSURANCE COMPANY		
	BY: AIG Global Investment Corp., investment adviser		
	By:		
	Name: Gerald F. Herman Title: Vice President		
Tax identification number	13-5459480		

Purchaser Name	AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY OF NEW YORK	
Name in which to register Notes	AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY OF NEW YORK	
Series B Note registration number; principal amount	RB-3; \$5,000,000	
Payment on account of Note		
Method	Federal Funds Wire Transfer	
Account information	Account Number: A OBI= <i>PPN # and des</i> P \$, I \$	BOS SAFE DEP LIFE ASSURANCE COMPANY OF N.Y. GIFLNY0372 scription of payment
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
	Due date and applica made.	ation (as among principal, premium and interest) of the payment being
Address / Fax # for notices related to payments	AIG Global Investment Group ATTN: Jennifer Lee / Kathleen Cosgrove 160 Water Street, 15th Floor New York, NY 10038 Tel: 212-820-4899 / 4913 Fax: 212-820-4925 Duplicate payment notices to:	
	American Internatio c/o AIG Global Inve 2929 Allen Parkway Houston, Texas 770 Attn: Private Placen Fax: (713) 831-1072	r, A36-04 19 eent Group
	Sch	nedule A-16

Purchaser Name	AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY OF NEW YORK				
Address / Fax # for all other notices	AIG Global Investment Group Legal Department – Investment Management 2929 Allen Parkway, Suite A36-01 Houston, TX 77019-2155 Fax (713) 831-2328				
	with copies of compliance information to:				
	American International Life Assurance Company of New York c/o AIG Global Investment Group 2929 Allen Parkway, A36-04 Houston, Texas 77019 Attn: Private Placement Group Fax: (713) 831-1072				
Instructions re Delivery of Notes	Mellon Bank Mellon Bank Securities Trust 120 Broadway - 13 th Floor New York, NY 10271 Attn: Sue Klein Account Name: AI LIFE ASSURANCE COMPANY OF N.Y. Account Number: AGIFLNY0372				
Signature Block	THE VARIABLE ANNUITY LIFE INSURANCE COMPANY				
0	THE UNITED STATES LIFE INSURANCE COMPANY				
	AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY IN THE CITY OF NEW YORK				
	AIG LIFE INSURANCE COMPANY				
	BY: AIG Global Investment Corp., investment adviser				
	By:				
	Name:Gerald F. HermanTitle:Vice President				
Tax identification number	13-6101875				
	Schedule A-17				

Purchaser Name	AIG LIFE INSURANCE COMPANY	
Name in which to register Notes	AIG LIFE INSURANCE COMPANY	
Series B Note registration number; principal amount	RB-4; \$5,000,000	
Payment on account of Note		
Method	Federal Funds Wire Transfe	r
Account information	Federal Reserve Bank of Boston Boston, MA ABA # 011001234 / BOS SAFE DEP DDA # 169064 Cost Center 1178 Account Name: AIG LIFE INSURANCE COMPANY Account Number: AGIFALI0392 OBI=PPN # and description of payment P \$, I \$ Re: (See "Accompanying information" below)	
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
Address / Fax # for notices related to payments	 Due date and application (as among principal, premium and interest) of the payment being made. AIG Global Investment Group ATTN: Jennifer Lee / Kathleen Cosgrove 160 Water Street, 15th Floor New York, NY 10038 Tel: 212-820-4899 / 4913 Fax: 212-820-4925 Duplicate payment notices to: 	
	AIG Life Insurance Company c/o AIG Global Investment Group 2929 Allen Parkway, A36-04 Houston, Texas 77019 Attn: Private Placement Group Fax: (713) 831-1072	
Schedule A-18		

Purchaser Name	AIG LIFE INSURANCE COMPANY			
Address / Fax # for all other notices	AIG Global Investment Group Legal Department – Investment Management 2929 Allen Parkway, Suite A36-01 Houston, TX 77019-2155 Fax (713) 831-2328			
	with copies of compliance information to:			
	AIG Life Insurance Company c/o AIG Global Investment Group 2929 Allen Parkway, A36-04 Houston, Texas 77019 Attn: Private Placement Group Fax: (713) 831-1072			
Instructions re Delivery of Notes	Mellon Bank Mellon Bank Securities Trust 120 Broadway - 13 th Floor New York, NY 10271 Attn: Sue Klein Account Name: AIG LIFE INSURANCE COMPANY Account Number: AGIFALI0392			
Signature Block	THE VARIABLE ANNUITY LIFE INSURANCE COMPANY			
	THE UNITED STATES LIFE INSURANCE COMPANY			
	AMERICAN INTERNATIONAL LIFE ASSURANCE COMPANY IN THE CITY OF NEW YORK			
	AIG LIFE INSURANCE COMPANY			
	BY: AIG Global Investment Corp., investment adviser			
	By:			
	Name:Gerald F. HermanTitle:Vice President			
Tax identification number	25-1118523			
	Schedule A-19			

Purchaser Name	NEW YORK LIFE INSURANCE COMPANY		
Name in which to register Notes	NEW YORK LIFE INSURANCE COMPANY		
Series A Note registration number; principal amount	RA-1; \$9,500,000		
Series B Note registration number; principal amount	RB-5; \$9,500,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	JPMorgan Chase Bank New York, New York 10019 ABA No.: 021-000-021 Credit: New York Life Insurance Company General Account No. 008-9-00687 Re: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Senior Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and application (as among principal, premium and interest) of the payment being made.		
Address / Fax # for notices related to payments	New York Life Insurance Company c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010-1603 Attention: Financial Management and Operations Group Securities Operations, 2 nd Floor Fax #: (212) 447-4160		
Address / Fax # for all other notices	 New York Life Insurance Company c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010 Attention: Securities Investment Group Private Finance, 2nd Floor Fax #: (212) 447-4122 with a copy of any notices regarding defaults or Events of Default under the operative documents to: Attention: Office of General Counsel Investment Section, Room 1104 Fax #: (212) 576-8340 		

Purchaser Name	NEW YORK LIFE INSURANCE COMPANY
Signature Block	NEW YORK LIFE INSURANCE COMPANY
	By:
	Name:
	Title:
Instructions re Delivery of Notes	New York Life Insurance Company
	c/o New York Life Investment Management LLC
	51 Madison Avenue
	New York, New York 10010
	Attn: Donnamarie Cristina, Esq.
Tax identification number	13-5582869
	Schedule A-21

Purchaser Name	NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION	
Name in which to register Notes	NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION	
Series A Note registration number; principal amount	RA-2; \$5,000,000	
Series B Note registration number; principal amount	RB-6; 5,000,000	
Payment on account of Note		
Method	Federal Funds Wire Transfer	
Account information	JPMorgan Chase Bank New York, New York 10019 ABA No.: 021-000-021 For the account of New York Life Insurance and Annuity Corporation General Account No. 323-8-47382 Re: (See "Accompanying information" below)	
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	5.57% Series A Senior Notes Due September 15, 2011
	PPN:	115236 A* 2
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
	Due date and applica made.	tion (as among principal, premium and interest) of the payment being
Address / Fax # for notices related to payments	New York Life Insurance and Annuity Corporation c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010-1603 Attention: Financial Management and Operations Group Securities Operations, 2 nd Floor Fax #: (212) 447-4160	
Address / Fax # for all other notices	 New York Life Insurance and Annuity Corporation c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010-1603 Attention: Securities Investment Group Private Finance, 2nd Floor Fax #: (212) 447-4122 with a copy of any notices regarding defaults or Events of Default under the operative documents to: Attention: Office of General Counsel Investment Section, Room 1104 Fax #: (212) 576-8340 	

Purchaser Name

Signature Block

NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION

NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION By: New York Life Investment Management LLC, Its Investment Manager

By:

29.

Name: Title:

Instructions re Delivery of Notes

New York Life Insurance Company c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010 Attn: Donnamarie Cristina, Esq.

Tax identification number

13-3044743

Purchaser Name	NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION INSTITUTIONALLY OWNED LIFE INSURANCE SEPARATE ACCOUNT		
Name in which to register Notes	NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION INSTITUTIONALLY OWNED LIFE INSURANCE SEPARATE ACCOUNT		
Series A Note registration number; principal amount	RA-3; \$500,000		
Series B Note registration number; principal amount	RB-7; \$500,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	JPMorgan Chase Bank New York, New York 10019 ABA No.: 021-000-021 Credit: NYLIAC SEPARATE BOLI 3 BROAD FIXED General Account No. 323-8-39002 Re: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Senior Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and applicat made.	tion (as among principal, premium and interest) of the payment being	
Address / Fax # for notices related to payments	New York Life Insurance and Annuity Corporation Institutionally Owned Life Insurance Separate Account c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010-1603 Attention: Financial Management and Operations Group Securities Operations, 2 nd Floor Fax #: (212) 447-4160		
	Sc	hodulo A-21	

Purchaser Name	NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION INSTITUTIONALLY OWNED LIFE INSURANCE SEPARATE ACCOUNT			
Address / Fax # for all other notices	New York Life Insurance and Annuity Corporation Institutionally Owned Life Insurance Separate Account c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010-1603 Attention: Securities Investment Group Private Finance, 2 nd Floor Fax #: (212) 447-4122			
	with a copy of any notices regarding defaults or Events of Default under the operative documents to:			
	Attention: Office of General Counsel Investment Section, Room 1104 Fax #: (212) 576-8340			
Signature Block	NEW YORK LIFE INSURANCE AND ANNUITY CORPORATION INSTITUTIONALLY OWNED LIFE INSURANCE SEPARATE ACCOUNT			
	By: New York Life Investment Management LLC, Its Investment Manager			
	By:			
	Name: Title:			
Instructions re Delivery of Notes	New York Life Insurance Company c/o New York Life Investment Management LLC 51 Madison Avenue New York, New York 10010 Attn: Donnamarie Cristina, Esq.			
Tax identification number	13-3044743			
	Schedule A-25			

Purchaser Name	CONNECTICUT GENERAL LIFE INSURANCE COMPANY		
Name in which to register Notes	CIG & CO.		
Series A Note registration number; principal amount	RA-4; \$10,000,000 RA-5; \$ 3,000,000		
Payment on account of Note			
Method	Federal Funds Wir	re Transfer	
Account information	J.P. Morgan Chase Bank BNF=CIGNA Private Placeme ABA # 021000021 Ref: (See "Accompanying info		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Due date and application (as an made.	nong principal, premium and interest) of the payment being	
Address / Fax # for notices related to payments	CIG & Co. c/o CIGNA Retirement & Investments Services Attn: Private and Alternative Investments, H16B 280 Trumbull Street Hartford, CT 06103 Fax: (860) 534-7203		
	with a copy to:		
	J.P. Morgan Chase Bank 14201 Dallas Parkway, 13th Fl Dallas, TX 75254 Attn: Heather Frisna, Mail Coc Fax: (469) 477-1904		
Address / Fax # for all other notices	CIG & Co. c/o CIGNA Retirement & Investments Services Attn: Private and Alternative Investments, H16B 280 Trumbull Street Hartford, CT 06103 Fax: (860) 534-7203		
Instructions re Delivery of Notes	J.P. Morgan Chase Bank 4 New York Plaza New York, NY 10004 Attn: John Bouquet Together with Transmittal to Securities Custodian Letter		
Signature Block	CONNECTICUT GENERAL	LIFE INSURANCE COMPANY	
	By: CIGNA Investments, Inc	<u>.</u>	
	By:		
	Name: Title:		
Tax identification number	13-3574027 (CIG & Co.)		

Purchaser Name	CIGNA LIFE INSURANCE COMPANY OF NEW YORK	
Name in which to register Notes	CIG & CO.	
Series A Note registration number; principal amount	RA-6; \$3,000,000	
Payment on account of Note		
Method	Federal Fund	s Wire Transfer
Account information	J.P. Morgan Chase Bank BNF=CIGNA Private Pla ABA # 021000021 Ref: (See "Accompanying	cements / AC=9009001802 g information" below)
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	5.57% Series A Notes Due September 15, 2011
	PPN:	115236 A* 2
	Due date and application payment being made.	(as among principal, premium and interest) of the
Address / Fax # for notices related to payments	CIG & Co. c/o CIGNA Retirement & Attn: Private and Alternat 280 Trumbull Street Hartford, CT 06103 Fax: (860) 534-7203	
	with a copy to:	
	J.P. Morgan Chase Bank 14201 Dallas Parkway, 13 Dallas, TX 75254 Attn: Heather Frisna, Mai Fax: (469) 477-1904	
Address / Fax # for all other notices	CIG & Co. c/o CIGNA Retirement & Investments Services Attn: Private and Alternative Investments, H16B 280 Trumbull Street Hartford, CT 06103 Fax: (860) 534-7203	
Instructions re Delivery of Notes	J.P. Morgan Chase Bank 4 New York Plaza New York, NY 10004 Attn: John Bouquet Together with Transmittal to Securities Custodian Letter	
Signature Block	CIGNA LIFE INSURAN	CE COMPANY OF NEW YORK
0	By: CIGNA Investme	ents, Inc.
	By:	
	Name: Title:	
Tax identification number	13-3574027 (CIC	G & Co.)
	Schedule A-27	

Purchaser Name	LIFE INSURANCE COMPANY OF NORTH AMERICA		
Name in which to register Notes	CIG & CO.		
Series A Note registration number; principal amount	RA-7; \$14,000,000		
Payment on account of Note			
Method	Federal Fund	s Wire Transfer	
Account information	ABA # 021000021	ank e Placements / AC=9009001802 nying information" below)	
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Due date and applica made.	tion (as among principal, premium and interest) of the payment being	
Address / Fax # for notices related to payments	CIG & Co. c/o CIGNA Retirement & Investments Services Attn: Private and Alternative Investments, H16B 280 Trumbull Street Hartford, CT 06103 Fax: (860) 534-7203		
	with a copy to:		
	J.P. Morgan Chase B 14201 Dallas Parkwa Dallas, TX 75254 Attn: Heather Frisna Fax: (469) 477-1904	ay, 13 th Floor , Mail Code 300-116	
Address / Fax # for all other notices	CIG & Co. c/o CIGNA Retirement & Investments Services Attn: Private and Alternative Investments, H16B 280 Trumbull Street Hartford, CT 06103 Fax: (860) 534-7203		
Instructions re Delivery of Notes	J.P. Morgan Chase Bank 4 New York Plaza New York, NY 10004 Attn: John Bouquet Together with Transmittal to Securities Custodian Letter		
Signature Block	LIFE INSURANCE By: CIGNA Invest	COMPANY OF NORTH AMERICA tments, Inc.	
	By:		
	Name: Title:		
Tax identification number	13-3574027 (CIG &	Co.)	
	S	chedule A-28	

Purchaser Name	ALLSTATE LIFE INSURANCE COMPANY		
Name in which to register Notes	ALLSTATE LIFE INSURANCE COMPANY		
Series A Note registration number; principal amount	RA-8; \$20,000,000		
Payment on account of Note			
Method	Fe	deral Funds Wire Transfer	
Account information	BBK = Harris Trust and Savings Bank ABA #: 071000288 BNF = Allstate Life Insurance Company Collection Account #168-117-0 ORG = (Enter Issuer Name) OBI = DPP - (Enter Private Placement No., if available) Payment Due Date (MM/DD/YY) P (Enter "P" and amount of principal being remitted, for example, P500000.00) - I (Enter "I" and amount of interest being remitted, for example, I225000.00) Ref: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Due date and applica payment being made.	tion (as among principal, premium and interest) of the	
Address / Fax # for notices related to payments	Allstate Investments LLC Investment Operations - Private Placements 3075 Sanders Road, STE G4A Northbrook, IL 60062-7127 Telephone: (847) 402-6672 Private Placements Telecopy: (847) 326-7032		
Address / Fax # for all other notices including notices of Prepayments:	Allstate Investments LLC Private Placements Department 3075 Sanders Road, STE G5D Northbrook, IL 60062-7127 Telephone: (847) 402-7117 Telecopy: (847) 402-3092		
Instructions re Delivery of Notes	Citibank N.A. 333 West 34th Street 3rd Floor Securities Vault New York, N.Y. 10001 Attn: Danny Reyes For Allstate Life Insurance Company/Safekeeping Account No. 846627		
Signature Block	ALLSTATE LIFE IN	SURANCE COMPANY	
	By:		
	Name:		
	By:		
	Name:		
	Authoriz	ed Signatories	
	S	chedule A-29	

Purchaser Name

ALLSTATE LIFE INSURANCE COMPANY

36-2554642

Tax identification number

Purchaser Name	ALLSTATE INSURANCE COMPANY		
Name in which to register Notes	ALLSTATE INSURANCE COMPANY		
Series A Note registration number; principal amount	RA-9; \$5,000,000		
Payment on account of Note			
Method		Federal Funds Wire Transfer	
Account information	BBK = Harris Trust and Savings Bank ABA #: 071000288 BNF = Allstate Insurance Company Collection Account #168-114-7 ORG = (Enter Issuer Name) OBI = DPP - (Enter Private Placement No., if available) Payment Due Date (MM/DD/YY) - P (Enter "P" and amount of principal being remitted, for example, P500000.00) - I (Enter "I" and amount of interest being remitted, for example, I225000.00) Ref: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Due date and applic made.	cation (as among principal, premium and interest) of the payment being	
Address / Fax # for notices related to payments	Allstate Investments LLC Investment Operations - Private Placements 3075 Sanders Road, STE G4A Northbrook, IL 60062-7127 Telephone: (847) 402-6672 Private Placements Telecopy: (847) 326-7032		
Address / Fax # for all other notices including notices of Prepayments:	Allstate Investments LLC Private Placements Department 3075 Sanders Road, STE G5D Northbrook, IL 60062-7127 Telephone: (847) 402-7117 Telecopy: (847) 402-3092		
Instructions re Delivery of Notes	Citibank N.A. 333 West 34th Street 3rd Floor Securities Vault New York, N.Y. 10001 Attn: Danny Reyes For Allstate Insurance Company/Safekeeping Account No. 846626		
Signature Block	ALLSTATE INSU	RANCE COMPANY	
	By:		
	Name:		
	By:		
	Name:		
	Author	rized Signatories	
	S	Schedule A-31	

Purchaser Name

ALLSTATE INSURANCE COMPANY

36-0719665

Tax identification number

Purchaser Name	UNITED OF OMAHA LIFE INSURANCE COMPANY	
Name in which to register Notes	UNITED OF OMAHA LIFE INSURANCE COMPANY	
Series A Note registration number; principal amount	RA-10; \$17,000,000	
Payment on account of Note		
Method	Federal Funds Wire Transfer	r
Account information	JPMorgan Chase Bank ABA #021000021 Private Income Processing For credit to: United of Omaha Life Insu Account # 900-9000200 a/c: G07097 Ref: (See "Accompanying information" l	
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	5.57% Series A Notes Due September 15, 2011
	PPN:	115236 A* 2
	Due date and application (as among prine being made.	cipal, premium and interest) of the payment
Address / Fax # for notices related to payments	JPMorgan Chase Bank 14201 Dallas Parkway - 13th Floor Dallas, TX 75254-2917 Attn: Income Processing – G. Ruiz a/c: G07097	
Address / Fax # for all other notices including notices of Prepayments:	4 - Investment Loan Administration United of Omaha Life Insurance Compar Mutual of Omaha Plaza Omaha, NE 68175-1011	ıy
Instructions re Delivery of Notes	JPMorgan Chase Bank North America Insurance 6th Floor Attn: Patricia A. Radzicki 3 Chase Metrotech Center Brooklyn, NY 11245	
Signature Block	UNITED OF OMAHA LIFE INSURAN	CE COMPANY
	By:	
	Name: Title:	
Tax identification number	47-0322111	
	Schedule A-33	

Purchaser Name	MUTUAL OF OMAHA LIFE INSURANCE COMPANY		
Name in which to register Notes	MUTUAL OF OMAHA LIFE INSURANCE COMPANY		
Series A Note registration number; principal amount	RA-11; \$6,000,000		
Payment on account of Note			
Method	Federal	Funds Wire Transfer	
Account information	JPMorgan Chase Bank ABA #021000021 Private Income Processing For credit to: Mutual of Omaha Life Insurance Company Account # 900-9000200 a/c: G07096 Ref: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Due date and applica made.	ation (as among principal, premium and interest) of the payment being	
Address / Fax # for notices related to payments	JPMorgan Chase Ba 14201 Dallas Parkwa Dallas, TX 75254-29 Attn: Income Proces a/c: G07096	ay - 13th Floor 917	
Address / Fax # for all other notices including notices of Prepayments:	4 - Investment Loan Administration United of Omaha Life Insurance Company Mutual of Omaha Plaza Omaha, NE 68175-1011		
Instructions re Delivery of Notes	JPMorgan Chase Bank North America Insurance 6th Floor Attn: Patricia A. Radzicki 3 Chase Metrotech Center Brooklyn, NY 11245		
Signature Block	MUTUAL OF OMA	HA LIFE INSURANCE COMPANY	
	By:		
	Name: Title:		
Tax identification number	47-0246511		

Purchaser Name	COMPANION LIFE INSURANCE COMPANY	
Name in which to register Notes	COMPANION LIFE INSURANCE COMPANY	
Series A Note registration number; principal amount	RA-12; \$2,000,000	
Payment on account of Note		
Method	Federal Funds Wire Transfer	
Account information	JPMorgan Chase Bank ABA #021000021 Private Income Processing For credit to: Companion Life Insurat Account # 900-9000200 a/c: G07903 Ref: (See "Accompanying information	
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	5.57% Series A Notes Due September 15, 2011
	PPN:	115236 A* 2
	Due date and application (as among p made.	rincipal, premium and interest) of the payment being
Address / Fax # for notices related to payments	JPMorgan Chase Bank 14201 Dallas Parkway 13th Floor Dallas, TX 75254-2917 Attn: Income Processing – G. Ruiz a/c: G07903	
Address / Fax # for all other notices including notices of Prepayments:	4 - Investment Loan Administration United of Omaha Life Insurance Con Mutual of Omaha Plaza Omaha, NE 68175-1011	ipany
Instructions re Delivery of Notes	JPMorgan Chase Bank North America Insurance - 6th Floor Attn: Patricia A. Radzicki 3 Chase Metrotech Center Brooklyn, NY 11245	
Signature Block	COMPANION LIFE INSURANCE O	COMPANY
	Ву:	
	Name: Title:	
Tax identification number	13-1595128	

Purchaser Name	THE PRUDENTIAL INSURANCE CO	MPANY OF AMERICA
Name in which to register Notes	THE PRUDENTIAL INSURAN	ICE COMPANY OF AMERICA
Series B Note registration number; principal amount	RB-8; \$10,500,000	
Payment on Account of Note		
Method	Federal Funds Wire Transfer	
Account Information	The Bank of New York New York, New York ABA # 021-000-018 Acct. # 890-0304-391 Re: (See "Accompanying inform	ation" below)
Accompanying Information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
	Due date and application (as amo	ong principal, premium and interest) of the payment being made.
Address for Notices Related to Payments	The Prudential Insurance Compa c/o Investment Operations Group Gateway Center Two, 10 th Floor 100 Mulberry Street Newark, New Jersey 07102-407 Attn: Manager, Billings and Coll Fax: 800-224-2278	7 Jections
	with telephonic prepayment notic	<u>ces to</u> :
	Manager, Trade Management Gr Tel: 973-367-3141 Fax: 800-224-2278	oup
Address for All Other Notices	The Prudential Insurance Compa c/o Prudential Capital Group 1170 Peachtree Street, Suite 500 Atlanta, GA 30309 Attn: Managing Director	
Instructions re Delivery of Notes	Prudential Capital Group 1114 Avenue of the Americas, 3(New York, NY 10036 Attn: Michael Fierro (212-626-2	
Signature Block	THE PRUDENTIAL INSURAN	ICE COMPANY OF AMERICA
	By:	
	Name: Title: Vice President	
Tax Identification Number	22-1211670	
	Schedule A	-36

Purchaser Name	RGA REINSURANCE COMPANY	
Name in which to register Notes	HARE & CO.	
Series B Note registration number; principal amount	RB-9; \$3,850,000	
Payment on Account of Note		
Method	Federal Funds Wire Transfer	
Account Information	The Bank of New York New York, New York ABA # 021-000-018 Acct. # 128863 RGA Private Placement Account Re: (See "Accompanying informati	on" below)
Accompanying Information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
	Due date and application (as among made.	g principal, premium and interest) of the payment being
Address for Notices Related to Payments	RGA Reinsurance Company Attn: Banking Dept. 1370 Timberlake Manor Parkway Chesterfield, MO 63017-6039	
Address for All Other Notices	Prudential Private Placement Invest 4 Gateway Center, 100 Mulberry St Newark, NJ 07102	
	Attention: Albert Trank, Managing Director	
	Telephone: (973) 802-8608 Facsimile: (973) 624-6432	

Purchaser Name	RGA REINSURANCE COMPANY	
Instructions re Delivery of Notes	Bank of New York Securities Department One Wall Street, 3rd Floor – "A" New York, NY 10286 Attention: Lucille Del Terzo Telephone: (212) 437-6764 Facsimile: (212) 437-6602 or (212) 437-6603 E-mail: <u>Idelterzo@bankofny.com</u>	
	Ref: account number (RGA Private Placement Account; Account Number: 128863).	
	Send copy by nationwide overnight delivery service to:	
	Prudential Capital Group Gateway Center 4 100 Mulberry, 7th Floor Newark, NJ 07102 Attention: Trade Management, Manager Telephone: (973) 367-3141	
Signature Block	RGA REINSURANCE COMPANY	
	By: Prudential Private Placement Investors, L.P. (as Investment Advisor)	
	By: Prudential Private Placement Investors, Inc. (as its General Partner)	
	By:	
	Name: Title: Vice President	
Tax Identification Number	43-1235868	
	Schedule A 20	

RB-10; \$2,000,000 Federal Funds Wire Trans JP Morgan Chase Bank	INSURANCE COMPANY OF FLORIDA, INC. fer	
Federal Funds Wire Trans JP Morgan Chase Bank	fer	
JP Morgan Chase Bank	fer	
JP Morgan Chase Bank	fer	
-	Federal Funds Wire Transfer	
JP Morgan Chase Bank ABA No.: 021000021 Account No.: 900-9000168 Account Name: JP Morgan Chase For further credit to Account No.: G09887 Account Name: ABIC - Prudential Private Placements Re: (See "Accompanying information" below)		
Name of Issuer:	BROWN & BROWN, INC.	
Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
PPN:	115236 A@ 0	
Due date and application (being made.	(as among principal, premium and interest) of the paymen	
JP Morgan Chase Bank Investor Services 3 Chase Metrotech Center North America Insurance, Brooklyn, NY 11245 Attention: Anna Marie Ma Telephone: (718) 242-539 Facsimile: (718) 242-8328	5S5 azza 9	
and		
Fortis, Inc. One Chase Manhattan Pla New York, NY 10005 Attention: Kevin P. Mahon AVP, Investme Telephone: (212) 859-718 Facsimile: (212) 859-7043	ney nt Accounting & Treasury Operations 4	
Prudential Private Placem 4 Gateway Center, 100 Mi Newark, NJ 07102 Attention: Albert Trank, N Telephone: (973) 802-860 Facsimile: (973) 624-6432	ulberry Street Aanaging Director 8	
	 Account No.: 900-900016 Account Name: JP Morga For further credit to Account Name: ABIC - Prese: (See "Accompanying" Name of Issuer: Description of Security: PPN: Due date and application (being made. JP Morgan Chase Bank Investor Services 3 Chase Metrotech Center North America Insurance, Brooklyn, NY 11245 Attention: Anna Marie Matter (718) 242-539 Facsimile: (718) 242-8328 and Fortis, Inc. One Chase Manhattan Pla New York, NY 10005 Attention: Kevin P. Mahona AVP, Investme Telephone: (212) 859-718 Facsimile: (212) 859-718 Facsimile: (212) 859-718 Facsimile: (212) 859-7043 Prudential Private Placem 4 Gateway Center, 100 M Newark, NJ 07102 Attention: Albert Trank, M Telephone: (973) 802-860 	

Purchaser Name	AMERICAN BANKERS INSURANCE COMPANY OF FLORIDA, INC.		
Instructions re Delivery of Notes	JP Morgan Chase Bank 4 New York Plaza, Ground Floor Window New York, NY 10004 Attention: Receive Window		
	Ref: account number (ABIC - Prudential Private Placements; Account Number: G09887)		
	Send copy by nationwide overnight delivery service to:		
	Prudential Capital Group Gateway Center 4 100 Mulberry, 7th Floor Newark, NJ 07102 Attention: Trade Management, Manager Telephone: (973) 367-3141		
Signature Block	AMERICAN BANKERS INSURANCE COMPANY OF FLORIDA, INC.		
	By: Prudential Private Placement Investors, L.P. (as Investment Advisor)		
	By: Prudential Private Placement Investors, Inc. (as its General Partner)		
	By:		
	Name: Title: Vice President		
Tax Identification Number	59-0593886		
	Schedule A-10		

Purchaser Name	AMERICAN MEMORIAL LIFE INSURANCE COMPANY		
Name in which to register Notes	AMERICAN MEMORIAL LIFE INSURANCE COMPANY		
Series B Note registration number; principal amount	RB-11; \$2,000,000		
Payment on Account of Note			
Method	Federal Funds Wire Transfer		
Account Information	M&I Marshall & Ilsley Bank Milwaukee, WI ABA No.: 075000051 DDA Account No.: 27006 Account Name: General Trust Fund For further credit to Account No.: 89-0035-77-7 Account Name: AMLIC Prudential Private Placements Re: (See "Accompanying information" below)		
Accompanying Information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and application (as and payment being made.	ong principal, premium and interest) of the	
Address for Notices Related to Payments	Marshall & Ilsley Trust Company 1000 North Water Street Milwaukee, WI 53202 Attention: Kim Palleon Telephone: (414) 287-7084 Facsimile: (414) 287-7125		
	and		
	Fortis, Inc. One Chase Manhattan Plaza New York, NY 10005 Attention: Kevin P. Mahoney AVP, Investment Acc Telephone: (212) 859-7184 Facsimile: (212) 859-7043	ounting & Treasury Operations	
Address for All Other Notices	Prudential Private Placement Investors, L.P. 4 Gateway Center, 100 Mulberry Street Newark, NJ 07102 Attention: Albert Trank, Managing Director Telephone: (973) 802-8608 Facsimile: (973) 624-6432		
Schedule A-41		-41	

Purchaser Name	AMERICAN MEMORIAL LIFE INSURANCE COMPANY				
Instructions re Delivery of Notes	Marshall & Ilsley Trust Company, N.A. 1000 N. Water Street, Asset Booking, TR14 Milwaukee, WI 53202 Attention: Margaret A. Armstrong Telephone: (414) 287-8531 Ref: account number (AMLIC - Prudential Private Placements; Account Number: 89 0035 77 7).				
	Send copy by nationwide overnight delivery service to:				
	Prudential Capital Group Gateway Center 4 100 Mulberry, 7th Floor Newark, NJ 07102 Attention: Trade Management, Manager Telephone: (973) 367-3141				
Signature Block	AMERICAN MEMORIAL LIFE INSURANCE COMPANY				
	By: Prudential Private Placement Investors, L.P. (as Investment Advisor)				
	By: Prudential Private Placement Investors, Inc. (as its General Partner)				
	By:				
	Name: Title: Vice President				
Tax Identification Number	46-0260270				

Purchaser Name	FORTIS INSURANCE COMPANY	
Name in which to register Notes	FORTIS INSURANCE COMPANY	
Series B Note registration number; principal amount	RB-12; \$1,150,000	
Payment on Account of Note		
Method	Federal Funds Wire Transfer	
Account Information	M&I Marshall & Ilsley Ba Milwaukee, WI ABA No.: 075000051 DDA Account No.: 27006 Account Name: General Th For further credit to Accou Account Name: Fortis Insu Re: (See "Accompanying i	rust Fund nt No.: 89-0035-78-5 ırance Prudential Private Placements
Accompanying Information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
	Due date and application (a made.	as among principal, premium and interest) of the payment being
Address for Notices Related to Payments	Marshall & Ilsley Trust Co 1000 North Water Street Milwaukee, WI 53202 Attention: Kim Palleon Telephone: (414) 287-7084 Facsimile: (414) 287-7125	•
	and Fortis, Inc.	
	One Chase Manhattan Plaz New York, NY 10005 Attention: Kevin P. Mahon	ey
	Telephone: (212) 859-7184 Facsimile: (212) 859-7043	
Address for All Other Notices	Prudential Private Placeme 4 Gateway Center, 100 Mu Newark, NJ 07102 Attention: Albert Trank, M Telephone: (973) 802-8608 Facsimile: (973) 624-6432	lberry Street anaging Director

Purchaser Name	FORTIS INSURANCE COMPANY		
Instructions re Delivery of Notes	Marshall & Ilsley Trust Company, N.A. 1000 N. Water Street, Asset Booking, TR14 Milwaukee, WI 53202 Attention: Margaret A. Armstrong Telephone: (414) 287-8531 Ref: account number (Fortis Insurance - Prudential Private Placements; Account Number: 89 0035 78 5).		
	Send copy by nationwide overnight delivery service to:		
	Prudential Capital Group Gateway Center 4 100 Mulberry, 7th Floor Newark, NJ 07102 Attention: Trade Management, Manager Telephone: (973) 367-3141		
Signature Block	FORTIS INSURANCE COMPANY		
	By: Prudential Private Placement Investors, L.P. (as Investment Advisor)		
	By: Prudential Private Placement Investors, Inc. (as its General Partner) By:		
	Title: Vice President		
Tax Identification Number	39-0658730		
	Schedule A-44		

Purchaser Name	JOHN ALDEN LIFE INSURANCH	E COMPANY
Name in which to register Notes	JOHN ALDEN LIFE INSURANCE COMPANY	
Series B Note registration number; principal amount	RB-13; \$1,500,000	
Payment on Account of Note		
Method	Federal Funds Wire Transfer	
Account Information	M&I Marshall & Ilsley Bank Milwaukee, WI ABA No.: 075000051 DDA Account No.: 27006 Account Name: General Trust Fund For further credit to Account No.: 89-0035-79-3 Account Name: JALIC Prudential Private Placements Re: (See "Accompanying information" below)	
Accompanying Information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
	Due date and application (as made.	among principal, premium and interest) of the payment being
Address for Notices Related to Payments	Marshall & Ilsley Trust Com 1000 North Water Street Milwaukee, WI 53202 Attention: Kim Palleon Telephone: (414) 287-7084 Facsimile: (414) 287-7125	pany
	and	
	Fortis, Inc. One Chase Manhattan Plaza New York, NY 10005 Attention: Kevin P. Mahoney AVP, Investment Telephone: (212) 859-7184 Facsimile: (212) 859-7043	7 Accounting & Treasury Operations
Address for All Other Notices	Prudential Private Placement 4 Gateway Center, 100 Mulb Newark, NJ 07102 Attention: Albert Trank, Mar Telephone: (973) 802-8608 Facsimile: (973) 624-6432	erry Street

Purchaser Name	JOHN ALDEN LIFE INSURANCE COMPANY				
Instructions re Delivery of Notes	Marshall & Ilsley Trust Company, N.A. 1000 N. Water Street, Asset Booking, TR14 Milwaukee, WI 53202 Attention: Margaret A. Armstrong Telephone: (414) 287-8531 Ref: account number (JALIC - Prudential Private Placements; Account Number: 89 0035 79 3).				
	Send copy by nationwide overnight delivery service to:				
	Prudential Capital Group Gateway Center 4 100 Mulberry, 7th Floor Newark, NJ 07102 Attention: Trade Management, Manager Telephone: (973) 367-3141				
Signature Block	JOHN ALDEN LIFE INSURANCE COMPANY				
	By: Prudential Private Placement Investors, L.P. (as Investment Advisor)				
	By: Prudential Private Placement Investors, Inc. (as its General Partner)				
	By:				
	Name: Title: Vice President				
Tax Identification Number	41-0999752				
	Schedule A-46				

Purchaser Name	PHOENIX LIFE INSURANCE COMPANY	
Name in which to register Notes	PHOENIX LIFE INSURANCE COMPANY	
Series A Note registration number; principal amount	RA-13; \$3,000,000	
Series B Note registration number; principal amount	RB-14; \$4,000,000	
Payment on account of Note		
Method	Federal Funds Wire Transfer	
Account information	JPMorgan Chase Bank New York, New York ABA No.: 021-000-021 Acct. No. 900 9002 859 Acct Name: Income Processing Ref: G05123, Phoenix Life Insur. Further Ref: (See "Accompanying information" below)	
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.
	Description of Security:	5.57% Series A Senior Notes Due September 15, 2011
	PPN:	115236 A* 2
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014
	PPN:	115236 A@ 0
	Due date and application (as an made.	nong principal, premium and interest) of the payment being
Address / Fax # for notices related to payments	Phoenix Investment Partners Attn: Private Placement Depar 56 Prospect Street Hartford, CT 06115	tment
Address / Fax # for all other notices	Phoenix Investment Partners Attn: Private Placement Depar 56 Prospect Street Hartford, CT 06115	tment
	with a copy to:	
	Phoenix Life Insurance Compa Attn: John Mulrain One American Row PO Box 5056 Hartford, CT 06102-5056	iny
Signature Block	PHOENIX LIFE INSURANC	E COMPANY
	By:	
	Name: Title:	
	Schedule A-47	

Purchaser Name

Instructions re Delivery of Notes

PHOENIX LIFE INSURANCE COMPANY

Phoenix Life Insurance Company Attn: John Mulrain One American Row Hartford, CT 06115

Tax identification number

06-0493340

Purchaser Name	PHL VARIABLE INSURANCE COMPANY		
Name in which to register Notes	PHL VARIABLE INSURANCE COMPANY		
Series A Note registration number; principal amount	RA-14; \$500,000		
Series B Note registration number; principal amount	RB-15; \$500,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	JPMorgan Chase Bank New York, New York ABA No.: 021-000-021 Acct. No. 900 9002 859 Acct Name: Income Processing Ref: G09766, Phoenix Life Insur. Further Ref: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Senior Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and application (as among principal, premium and interest) of the payment being made.		
Address / Fax # for notices related to payments	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115		
Address / Fax # for all other notices	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115		
	with a copy to:		
	Phoenix Life Insurance Company Attn: John Mulrain One American Row PO Box 5056 Hartford, CT 06102-5056		
Signature Block	PHL VARIABLE INSUF	ANCE COMPANY	
	Ву:		
	Name: Title:		
	Schedule A-49		

Purchaser Name

Instructions re Delivery of Notes

PHL VARIABLE INSURANCE COMPANY

Phoenix Life Insurance Company Attn: John Mulrain One American Row Hartford, CT 06115

Tax identification number

06-1045829

Purchaser Name	PHL VARIABLE INSURANCE COMPANY		
Name in which to register Notes	PHL VARIABLE INSURANCE COMPANY		
Series A Note registration number; principal amount	RA-15; \$1,000,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	JPMorgan Chase Bank New York, New York ABA No.: 021-000-021 Acct. No. 900 9002 859 Acct Name: Income Processing Ref: G09390, Phoenix Life Insur. Further Ref: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Due date and application made.	(as among principal, premium and interest) of the payment being	
Address / Fax # for notices related to payments	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115		
Address / Fax # for all other notices	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115		
	with a copy to:		
	Phoenix Life Insurance Company Attn: John Mulrain One American Row		
	PO Box 5056 Hartford, CT 06102-5056	õ	
Signature Block	PHL VARIABLE INSUF	RANCE COMPANY	
	By:		
	Name: Title:		
Instructions re Delivery of Notes	Phoenix Life Insurance Company Attn: John Mulrain One American Row Hartford, CT 06115		
Tax identification number	06-1045829		
	Schedule A	A-51	

Purchaser Name	PHL VARIABLE INSURANCE COMPANY		
Name in which to register Notes	PHL VARIABLE INSURANCE COMPANY		
Series A Note registration number; principal amount	RA-16; \$500,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	JPMorgan Chase Bank New York, New York ABA No.: 021-000-021 Acct. No. 900 9002 859 Acct Name: Income Processing Ref: G09389, Phoenix Life Insur. Further Ref: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	5.57% Series A Notes Due September 15, 2011	
	PPN:	115236 A* 2	
	Due date and application (as among principal, premium and interest) of the payment being made.		
Address / Fax # for notices related to payments	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115		
Address / Fax # for all other notices	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115 with a copy to:		
	Phoenix Life Insurance Company		
	Attn: John Mulrain One American Row		
	PO Box 5056		
	Hartford, CT 06102-5056		
Signature Block	PHL VARIABLE INSURANCE COMPAN	Y	
	By:		
	Name: Title:		
Instructions re Delivery of Notes	Phoenix Life Insurance Company Attn: John Mulrain One American Row Hartford, CT 06115		
Toy identification work			
Tax identification number	06-1045829		
	Schedule A-52		

Durchasor Name	PHL VARIABLE INSURANCE COMPANY		
Purchaser Name	PIL VARIABLE INSUF	VANCE COMPANY	
Name in which to register Notes	PHL VARIABLE INSURANCE COMPANY		
Series B Note registration number; principal amount	RB-16; \$1,500,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	JPMorgan Chase Bank New York, New York ABA No.: 021-000-021 Acct. No. 900 9002 859 Acct Name: Income Processing Ref: G05948, Phoenix Life Insur. Further Ref: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and application (as among principal, premium and interest) of the payment being made.		
Address / Fax # for notices related to payments	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115		
Address / Fax # for all other notices	Phoenix Investment Partners Attn: Private Placement Department 56 Prospect Street Hartford, CT 06115		
	with a copy to:		
	Phoenix Life Insurat Attn: John Mulrain One American Row PO Box 5056 Hartford, CT 06102-		
Signature Block	PHL VARIABLE INSURANCE COMPANY By:		
	Name: Title:		
Instructions re Delivery of Notes	Phoenix Life Insurance Company Attn: John Mulrain One American Row Hartford, CT 06115		
Tax identification number	06-1045829		
	So	chedule A-53	

Purchaser Name	LIFE INSURANCE COMPANY OF THE SOUTHWEST		
Name in which to register Notes	LIFE INSURANCE COMPANY OF THE SOUTHWEST		
Series B Note registration number; principal amount	RB-17; \$6,000,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	J.P. Morgan Chase & Co. New York, NY 10010 ABA #: 021000021 Account No.: 910-2-754349 Re: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and application (as among principal, premium and interest) of the payment being made.		
Address / Fax # for notices related to payments	Life Insurance Company of the Southwest c/o National Life Insurance Company One National Drive Montpelier, VT 05604 Attn: Private Placements Fax: (802) 223-9332		
Address / Fax # for all other notices	Life Insurance Company of the Southwest c/o National Life Insurance Company One National Drive Montpelier, VT 05604 Attn: Private Placements Fax: (802) 223-9332		
Instructions re Delivery of Notes	Life Insurance Company of the Southwest c/o National Life Insurance Company One National Drive Montpelier, VT 05604 Attn: R. Scott Higgins		
Signature Block	LIFE INSURAN	LIFE INSURANCE COMPANY OF THE SOUTHWEST By: Name: Title:	
	By:		
Tax identification number	75-0953004		

Purchaser Name	SECURITY FINANCIAL LIFE INSURANCE CO.		
Name in which to register Notes	SECURITY FINANCIAL LIFE INSURANCE CO.		
Series B Note registration number; principal amount	RB-18; \$2,000,000		
Payment on account of Note			
Method	Federal Funds Wire Transfer		
Account information	Union Bank & Trust Company 4732 Calvert Street Lincoln NE 68501-2535 ABA No. 1049 1079 5 Account of: Security Financial Life Insurance Co. Account Number: 338 1480 Re: (See "Accompanying information" below)		
Accompanying information	Name of Issuer:	BROWN & BROWN, INC.	
	Description of Security:	6.08% Series B Senior Notes Due July 15, 2014	
	PPN:	115236 A@ 0	
	Due date and application (as among principal, premium and interest) of the made.		
Address / Fax # for notices related to payments	Security Financial Life Insurance Co. 4000 Pine Lake Road Lincoln, NE 68516 Attention: Investment Division Fax: (402) 458-2170 Phone: (402) 434-9600		
Address / Fax # for all other notices	Security Financial Life Insurance Co. 4000 Pine Lake Road P.O. Box 82248 Lincoln, NE 68501-2248		
Instructions re Delivery of Notes	Security Financial Life Insurance Co. 4000 Pine Lake Road Lincoln, NE 68516 Attention: Svetlana Ungar		
Signature Block	SECURITY FINANCIAL LIFE INSURANCE CO. By:		
	Name: Title:		
Tax identification number	47-0293990		
		Schedule A-55	

SCHEDULES B

DEFINED TERMS

As used herein, the following terms have the respective meanings set forth below or set forth in the Section hereof following such term:

"Affiliate" means, at any time, and with respect to any Person, any other Person that at such time directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, such first Person. As used in this definition, "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. Unless the context otherwise clearly requires, any reference to an "Affiliate" is a reference to an Affiliate of the Company.

"Asset Disposition" means any Transfer except:

(a) any

(i) Transfer from a Subsidiary to the Company or a Wholly-Owned Subsidiary;

(ii) Transfer from the Company to a Wholly-Owned Subsidiary; and

(iii) Transfer from the Company or a Wholly-Owned Subsidiary to a Subsidiary (other than a Wholly-Owned Subsidiary) or from a Subsidiary to another Subsidiary, which in either case is for Fair Market Value;

so long as immediately before and immediately after the consummation of any such Transfer and after giving effect thereto, no Default or Event of Default exists; and

(b) any Transfer made in the ordinary course of business and involving only property that is either (i) inventory held for sale or (ii) equipment, fixtures, supplies or materials that are obsolete.

"**Bank Credit Agreement**" means, collectively, that certain Revolving Loan Agreement, dated as of September 29, 2003, between the Company and SunTrust Bank and that certain Amended and Restated Revolving and Term Loan Agreement, dated as of January 3, 2001, between the Company and SunTrust Bank, each as amended, refinanced or replaced from time to time.

"Book of Business Sales" means the sale by the Company or any Subsidiary in the ordinary course of business of a book of business, either by the sale of assets or Capital Stock, which may include the sale of what is characterized as its profit center operations (i.e. office) that are made from time to time and are consistent with past practices, and where the value is less than \$10,000,000.

"Brown Family" means, collectively, (a) J. Hyatt Brown and Celia Brown and their children and grandchildren and (b) Persons controlled by or for the benefit of such individuals.

"Business Day" means any day other than a Saturday, a Sunday or a day on which commercial banks in New York, New York are required or authorized to be closed.

"Capital Lease" means, at any time, a lease with respect to which the lessee is required concurrently to recognize the acquisition of an asset and the incurrence of a liability in accordance with GAAP.

"Capital Stock" means any class of capital stock, share capital, limited liability company interest, general or limited partnership interest or any other similar equity interest of a Person.

"Change in Control" is defined in Section 8.3(f).

"Change in Control Prepayment Date" is defined in Section 8.3(b).

"Closing" is defined in Section 3.

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.

"Company" is defined in the introductory paragraph of this Agreement.

"Confidential Information" is defined in Section 20.

"Consolidated EBITDA" means, for any period, the result of (a) Consolidated Net Income for such period plus, to the extent deducted in determining Consolidated Net Income for such period, the aggregate amount of (i) Consolidated Interest Expense for such period, (ii) income tax expense for such period, (iii) depreciation and amortization for such period and (iv) non-cash charges, including all non-cash compensation, minus (b) gains on sales of assets (excluding sales in the ordinary course of business, which would include Book of Business Sales) and other extraordinary gains and other one-time non-cash gains, all as determined in accordance with GAAP.

"Consolidated Interest Expense" means, for any period, the sum, for the Company and its Subsidiaries (determined in accordance with GAAP), of all interest expense in respect of Indebtedness (including, without limitation, the interest component of any payments in respect of Capital Lease obligations) accrued or capitalized during such period; provided, however, that in calculating Consolidated Interest Expense effect shall be given to any interest rate protection agreements, whether or not such agreements constitute debits or credits in accordance with GAAP.

"**Consolidated Net Income**" means, for any period, the net income (or loss) of the Company and its Subsidiaries, determined on a consolidated basis for such period and as determined in accordance with GAAP, adjusted to exclude the effect of any extraordinary gain or loss.

"**Consolidated Net Indebtedness**" means, as of any date of determination, the total of all Indebtedness of the Company and its Subsidiaries outstanding on such date, after eliminating all offsetting debits and credits between the Company and its Subsidiaries and all other items required to be eliminated in the course of preparation of consolidated financial statements of the Company and its Subsidiaries in accordance with GAAP, less the aggregate amount of Permitted Investments in excess of \$25,000,000.

"Consolidated Net Worth" means, as of any date, total stockholders' equity of the Company and its Subsidiaries as of such date, determined on a consolidated basis in accordance with GAAP.

"Consolidated Rental Expense" means, for any period, the sum of all rentals paid or payable under operating lease obligations of the Company and its Subsidiaries (including, without limitation, all payments which the lessee is obligated to make to the lessor as the result of the termination of a lease obligation or any surrender of the property subject to such lease obligation, but excluding therefrom any amounts required to be paid by the Company or a Subsidiary (whether or not designated as rents or additional rents) on account of maintenance, repairs, insurance, taxes and similar charges), all determined on a consolidated basis for such period.

"**Consolidated Total Assets**" means, as of any date, the total assets of the Company and its Subsidiaries which would be shown as assets on a consolidated balance sheet of the Company and its Subsidiaries as of such date prepared in accordance with GAAP, after eliminating all amounts properly attributable to minority interests, if any, in the stock and surplus of Subsidiaries.

"Debt Prepayment Application" means, with respect to any Asset Disposition of any property, the application by the Company or any Subsidiary, as the case may be, of cash in an amount equal to the Net Proceeds with respect to such Asset Disposition to pay Senior Debt (other than (a) Senior Debt owing to the Company or any of its Subsidiaries or any Affiliate and (b) Senior Debt in respect of any revolving credit or similar facility providing any Obligor or any such Subsidiary with the right to obtain loans or other extensions of credit from time to time, unless in connection with such payment of Senior Debt the availability of credit under such credit facility is permanently reduced by an amount not less than the amount of such proceeds applied to the payment of such Senior Debt), *provided* that in the course of making such application the Company shall offer to prepay each outstanding Note, in accordance with Section 8.4, in a principal amount equal to the Ratable Portion of the holder of such Note in respect of such Asset Disposition. If any holder of a Note fails to accept such offer of prepayment, then, for purposes of the preceding sentence only, the Company nevertheless will be deemed to have paid Senior Debt in an amount equal to the Ratable Portion of the holder of such Asset Disposition.

"Debt Prepayment Transfer" is defined in Section 8.4(a).

"Default" means an event or condition the occurrence or existence of which would, with the lapse of time or the giving of notice or both, become an Event of Default.

"**Default Rate**" means that rate of interest in respect of a Series of Notes that is the greater of (a) 2% per annum above the rate of interest stated in clause (a) of the first paragraph of the Notes of such Series or (b) 2% over the rate of interest publicly announced by The Bank of New York in New York, New York as its "base" or "prime" rate.

"Disposition Value" means, at any time, with respect to any property

(a) in the case of property that does not constitute Subsidiary Stock, the book value thereof, valued at the time of such disposition in good faith by the Company, and

(b) in the case of property that constitutes Subsidiary Stock, an amount equal to that percentage of book value of the assets of the Subsidiary that issued such stock as is equal to the percentage that the book value of such Subsidiary Stock represents of the book value of all of the outstanding Capital Stock of such Subsidiary (assuming, in making such calculations, that all Securities convertible into such Capital Stock are so converted and giving full effect to all transactions that would occur or be required in connection with such conversion) determined at the time of the disposition thereof in good faith by the Company.

"Environmental Laws" means any and all Federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including but not limited to those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the rules and regulations promulgated thereunder from time to time in effect.

"ERISA Affiliate" means any trade or business (whether or not incorporated) that is treated as a single employer together with the Company under section 414 of the Code.

"Event of Default" is defined in Section 11.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Fair Market Value" means, at any date of determination and with respect to any property, the sale value of such property that would be realized in an arm's-length sale at such time between an informed and willing buyer and an informed and willing seller (neither being under a compulsion to buy or sell) as determined by a Responsible Officer of the Company acting in good faith.

"Financing Documents" means this Agreement, the Notes and the Subsidiary Guaranty, as each may be amended, restated or otherwise modified from time to time, and all other documents to be executed and/or delivered in favor of any holders of Notes, or all of them, by the Company, any of its Subsidiaries, or any other Person in connection with this Agreement.

"First Closing" is defined in Section 3.1.

"First Closing Date" is defined in Section 3.1.

"**Fixed Charge Coverage Ratio**" means, at the end of any fiscal quarter, the ratio of (a) the sum of (i) Consolidated EBITDA *plus* (ii) Consolidated Rental Expense, both calculated for the period of four consecutive fiscal quarters then ended to (b) the sum of (i) Consolidated Interest Expense *plus* (ii) Consolidated Rental Expense, both calculated for such period.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States of America.

"Governmental Authority" means

(a) the government of

(i) the United States of America or any State or other political subdivision thereof, or

(ii) any jurisdiction in which the Company or any Subsidiary conducts all or any part of its business, or which asserts jurisdiction over any properties of the Company or any Subsidiary, or

(b) any entity exercising executive, legislative, judicial, regulatory or administrative functions of, or pertaining to, any such government.

"Guaranty" means, with respect to any Person, any obligation (except the endorsement in the ordinary course of business of negotiable instruments for deposit or collection) of such Person guaranteeing or in effect guaranteeing any indebtedness, dividend or other obligation of any other Person in any manner, whether directly or indirectly, including (without limitation) obligations incurred through an agreement, contingent or otherwise, by such Person:

(a) to purchase such indebtedness or obligation or any property constituting security therefor;

(b) to advance or supply funds (i) for the purchase or payment of such indebtedness or obligation, or (ii) to maintain any working capital or other balance sheet condition or any income statement condition of any other Person or otherwise to advance or make available funds for the purchase or payment of such indebtedness or obligation;

(c) to lease properties or to purchase properties or services primarily for the purpose of assuring the owner of such indebtedness or obligation of the ability of any other Person to make payment of the indebtedness or obligation; or

(d) otherwise to assure the owner of such indebtedness or obligation against loss in respect thereof.

In any computation of the indebtedness or other liabilities of the obligor under any Guaranty, the indebtedness or other obligations that are the subject of such Guaranty shall be assumed to be direct obligations of such obligor.

"holder" means, with respect to any Note, the Person in whose name such Note is registered in the register maintained by the Company pursuant to Section 13.1.

"Indebtedness" means, with respect to any Person, at any time, without duplication,

(a) its liabilities for borrowed money and its redemption obligations in respect of mandatorily redeemable Preferred Stock;

(b) its liabilities for the deferred purchase price of property acquired by such Person (including all liabilities created or arising under any conditional sale or other title retention agreement with respect to any such property but excluding (i) accounts payable arising in the ordinary course of business and (ii) payment obligations in respect of acquisitions of any Person (whether by acquisition of a majority of the Voting Stock of such Person or all or substantially all of its assets) to the extent that the amount thereof is dependent on the future financial performance of such Person (or assets) until such time as any such obligation would be required to be shown on a balance sheet of such Person prepared at such time in accordance with GAAP (at which time such obligation shall constitute "Indebtedness"));

(c) all liabilities appearing on its balance sheet in accordance with GAAP in respect of Capital Leases;

(d) all liabilities for borrowed money secured by any Lien with respect to any property owned by such Person (whether or not it has assumed or otherwise become liable for such liabilities);

(e) any Guaranty of such Person with respect to liabilities of a type described in any of clauses (a) through (d) hereof excluding letters of credit except for standby letters of credit constituting credit support for any Indebtedness described in clauses (a) through (d) above.

Indebtedness of any Person shall include all obligations of such Person of the character described in clauses (a) through (e) to the extent such Person remains legally liable in respect thereof notwithstanding that any such obligation is deemed to be extinguished under GAAP.

"INHAM Exemption" is defined in Section 6.2(e).

"**Institutional Investor**" means (a) any original purchaser of a Note, (b) any holder of a Note holding more than 5% of the aggregate principal amount of the Notes then outstanding, and (c) any bank, trust company, savings and loan association or other financial institution, any pension plan, any investment company, any insurance company, any broker or dealer, or any other similar financial institution or entity, regardless of legal form.

"Lien" means, with respect to any Person, any mortgage, lien, pledge, charge, security interest or other encumbrance, or any interest or title of any vendor, lessor, lender or other secured party to or of such Person under any conditional sale or other title retention agreement or Capital Lease, upon or with respect to any property or asset of such Person (including in the case of stock, stockholder agreements, voting trust agreements and all similar arrangements).

"Make-Whole Amount" is defined in Section 8.8.

"Material" means material in relation to the business, operations, affairs, financial condition, assets, or properties of the Company and its Subsidiaries taken as a whole.

"**Material Adverse Effect**" means a material adverse effect on (a) the business, operations, affairs, financial condition, assets or properties of the Company and its Subsidiaries taken as a whole, (b) the ability of the Company to perform its obligations under this Agreement or the Notes or (c) the validity or enforceability of this Agreement, the Notes or the Subsidiary Guaranty.

"Memorandum" is defined in Section 5.3.

"Multiemployer Plan" means any Plan that is a "multiemployer plan" (as such term is defined in section 4001(a)(3) of ERISA).

"Net Proceeds" means, with respect to any Transfer of any property by any Person, an amount equal to the difference of

(a) the aggregate amount of the consideration (valued at the Fair Market Value of such consideration at the time of the consummation of such Transfer) received by such Person in respect of such Transfer, *minus*

(b) all ordinary and reasonable out-of-pocket costs and expenses actually incurred by such Person in connection with such Transfer.

"Notes" is defined in Section 1.

"Obligors" means, collectively, the Company and the Subsidiary Guarantors.

"Officer's Certificate" means a certificate of a Senior Financial Officer or of any other officer of the Company whose responsibilities extend to the subject matter of such certificate.

"PBGC" means the Pension Benefit Guaranty Corporation referred to and defined in ERISA or any successor thereto.

"Permitted Investments" means the aggregate amount of the following: (to the extent not subject to set-off rights, otherwise subject to Liens, or securing liabilities that do not constitute Indebtedness):

(a) cash balances maintained by the Company or any Subsidiary with Prime Banks;

(b) the principal amount of all negotiable certificates of deposit, commercial paper or promissory notes held by the Company or any Subsidiary having a maturity not exceeding one year issued by (i) any United States or any state or governmental agency that is rated at least "A" by Moody's or "A" by Standard and Poor's or (ii) any other entity the short-term debt obligations of which are rated at least "A" by Moody's or "A" by Standard and Poor's or (iii) any Prime Bank; and

(c) investments in investment funds, investment companies or similar entities sponsored or managed by Prime Banks or other financial institutions which meet the requirements set forth in clause (b) above and 80% or more of the portfolio of which constitutes cash or other securities of the type specified in clause (b) above.

For the avoidance of doubt "Permitted Investments" shall not include any "Restricted Cash" as identified on the Company's consolidated balance sheet.

"**Person**" means an individual, partnership, corporation, limited liability company, association, trust, unincorporated organization, or a government or agency or political subdivision thereof.

"**Plan**" means an "employee benefit plan" (as defined in section 3(3) of ERISA) that is or, within the preceding five years, has been established or maintained, or to which contributions are or, within the preceding five years, have been made or required to be made, by the Company or any ERISA Affiliate or with respect to which the Company or any ERISA Affiliate may have any liability.

"**Preferred Stock**" means any class of capital stock of a corporation that is preferred over any other class of capital stock of such corporation as to the payment of dividends or the payment of any amount upon liquidation or dissolution of such corporation.

"Prime Bank" means a bank or trust company whose long-term unsecured debt is rated "A2" or better by Moody's Investors Service, Inc. or "A" or better by Standard & Poor's, Inc. ratings services.

"**Priority Debt**" means, as of any date, (without duplication) the sum of (a) all outstanding Indebtedness of any Subsidiary (other than Indebtedness permitted under clauses (a) through (c) of Section 10.7) and (b) all Indebtedness of the Company secured by any Lien pursuant to clause (j) of Section 10.4).

"**Pro Forma Basis**" means, in connection with a merger or consolidation or any conveyance of assets and any determination of "Consolidated EBITDA", "Consolidated Net Indebtedness", "Consolidated Interest Expense" and "Consolidated Rental Expense" for any period of four consecutive fiscal quarters of the Company or a Successor Corporation (and compliance with any covenant using any of such terms), that such determination shall be made on the assumptions that any merger, acquisition, consolidation or conveyance shall be deemed to have occurred on the first day of the fourth full fiscal quarter preceding the date of determination and all Indebtedness incurred or paid (or to be incurred or paid) by all such Persons in connection with all such transactions (x) was incurred or paid on the first day of such four fiscal quarter period, as the case may be, and (y) if incurred, was outstanding in full at all times during such period and had in effect at all times during such period (or any portion of such period during which such Indebtedness was not actually outstanding) an interest rate equal to the interest rate in effect on the date of the actual incurrence thereof (regardless of whether such interest rate is a floating rate or would otherwise change over time by reference to a formula or for any other reason).

"property" or "properties" means, unless otherwise specifically limited, real or personal property of any kind, tangible or intangible, choate or inchoate.

"**Property Reinvestment Application**" means, with respect to any Transfer of property, the application of an amount equal to the Net Proceeds with respect to such Transfer to the acquisition by such Person of operating assets of such Person (excluding, for the avoidance of doubt, cash and cash equivalents), and of at least equivalent Fair Market Value, to the property so Transferred, to be used in the principal business of such Person as conducted immediately prior to such Transfer or in a business generally related to such principal business.

"PTE" is defined in Section 6.2(a).

"Purchasers" is defined in the introductory paragraph of this Agreement.

"QPAM Exemption" means Prohibited Transaction Class Exemption 84-14 issued by the United States Department of Labor.

"Ratable Portion" means, in respect of any holder of any Note and any Transfer contemplated by the definition of Debt Prepayment Application, an amount equal to the product of

(a) the Net Proceeds being offered to be applied to the payment of Senior Debt, multiplied by

(b) a fraction the numerator of which is the outstanding principal amount of such Note and the denominator of which is the aggregate outstanding principal amount of Senior Debt at the time of such Transfer determined on a consolidated basis in accordance with GAAP.

"**Required Holders**" means, at any time, the holders of at least a majority in principal amount of the Notes (without regard to Series) at the time outstanding (exclusive of Notes then owned by the Company or any of its Affiliates).

"**Responsible Officer**" means any Senior Financial Officer or any other officer of the Company with responsibility for the administration of the relevant portion of this Agreement.

"Second Closing" is defined in Section 3.2.

"Second Closing Compensation Amount" is defined in Section 3.3.

"Second Closing Date" is defined in Section 3.2.

"Securities Act" means the Securities Act of 1933, as amended from time to time.

"Senior Debt" means the Notes and any other Indebtedness of the Company or its Subsidiaries that by its terms is not in any manner subordinated in right of payment to any other unsecured Indebtedness of the Company or any Subsidiary.

"Senior Financial Officer" means the chief financial officer, principal accounting officer, treasurer or controller of the Company.

"Series" means any one or more series of Notes issued under this Agreement.

"Series A Notes" is defined in Section 1.

"Series B Notes" is defined in Section 1.

"Significant Subsidiary" means at any time any Subsidiary that would at such time constitute a "significant subsidiary" (as such term is defined in Regulation S-X of the Securities and Exchange Commission as in effect on the First Closing Date) of the Company; provided that Subsidiary Guarantors shall be deemed to be Significant Subsidiaries at all times.

"Source" is defined in Section 6.2.

"Subsidiary" means, as to any Person, any corporation, association or other business entity in which such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries owns sufficient equity or voting interests to enable it or them (as a group) ordinarily, in the absence of contingencies, to elect a majority of the directors (or Persons performing similar functions) of such entity, and any partnership or joint venture if more than a 50% interest in the profits or capital thereof is owned by such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries (unless such partnership can and does ordinarily take major business actions without the prior approval of such Person or one or more of its Subsidiaries). Unless the context otherwise clearly requires, any reference to a "Subsidiary" is a reference to a Subsidiary of the Company.

"Subsidiary Guarantor" means, at any time, any Subsidiary that has executed and delivered the Subsidiary Guaranty or the joinder agreement thereto and has not been released in accordance with Section 9.6.

"Subsidiary Guaranty" is defined in Section 4.10.

"Subsidiary Stock" means, with respect to any Person, the Capital Stock (or any options or warrants to purchase stock, shares or other Securities exchangeable for or convertible into stock or shares) of any Subsidiary of such Person.

"Successor Corporation" is defined in Section 10.2(a).

"**Transfer**" means, with respect to any Person, any transaction in which such Person sells, conveys, transfers or leases (as lessor) any of its property, including, without limitation, any transfer or issuance of any Subsidiary Stock (whether by means of a merger of the issuer of such Subsidiary Stock or otherwise). For purposes of determining the application of the Net Proceeds in respect of any Transfer, the Company may designate any Transfer as one or more separate Transfers each yielding separate Net Proceeds. In any such case, (a) the Disposition Value of any property subject to each such separate Transfer shall be determined by ratably allocating the aggregate Disposition Value of, and the aggregate Net Proceeds attributable to, all property subject to all such separate Transfers to each such separate Transfer on a proportionate basis.

"Transfer Prepayment Date" is defined in Section 8.4(a).

"Transfer Prepayment Offer" is defined in Section 8.4(a).

"USA Patriot Act" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA PATRIOT ACT) Act of 2001 of the United States of America.

"Voting Stock" means, with respect to any Person, Capital Stock of any class or classes of a corporation, an association or another business entity the holders of which are ordinarily, in the absence of contingencies, entitled to vote in the election of corporate directors (or individuals performing similar functions) of such Person or which permit the holders thereof to control the management of such Person, including general partnership interests in a partnership and membership interests in a limited liability company.

"Wholly-Owned Subsidiary" means, at any time, any Subsidiary one hundred percent (100%) of all of the equity interests (except directors' qualifying shares) and voting interests of which are owned by any one or more of the Company and the Company's other Wholly-Owned Subsidiaries at such time.

SCHEDULE 3

Payment Instructions

See attached.

Schedule 3

SCHEDULE 4.9

Changes in Corporate Structure

Schedule 4.9

SCHEDULE 4.10

Subsidiary Guarantors

B & B Insurance Services, Inc. Brown & Brown Insurance of Arizona, Inc. Brown & Brown Insurance of Nevada, Inc. Brown & Brown of California, Inc. Brown & Brown of Lehigh Valley, Inc. Brown & Brown of Washington, Inc. Florida Intracoastal Underwriters, Limited Company MacDuff Underwriters, Inc. Madoline Corporation Peachtree Special Risk Brokers, LLC Program Management Services, Inc. The Young Agency, Inc.

Schedule 4.10

SCHEDULE 5.3

Disclosure Materials

Schedule 5.3

SCHEDULE 5.4

Subsidiaries of the Company and Ownership of Subsidiary Stock

One hundred percent (100%) of the outstanding shares of Capital Stock of each direct subsidiary (that is, those companies listed without any symbol preceding them) are owned by Brown & Brown, Inc.

- = indirect subsidiary, whose outstanding shares of Capital Stock (or, in the case of companies identified as limited liability companies, membership interests) are owned 100% by the direct subsidiary (company listed without any symbol preceding its name) listed above the name of such indirect subsidiary
- = indirect subsidiary whose outstanding shares of Capital Stock are owned 100% by the indirect subsidiary (company with symbol preceding its name) listed above the name of such indirect subsidiary

Acumen Re Management Corporation(DE)

AFC Insurance, Inc. (PA)

B & B Insurance Services, Inc. (FL)

- Ernest Smith Insurance Agency, Inc. (FL)
- Halcyon Underwriters, Inc. (FL)
- Hotel-Motel Insurance Group, Inc. (FL)
- MacDuff Underwriters, Inc. (FL)
 - MacDuff America, Inc. (FL)
 - MacDuff Pinellas Underwriters, Inc. (FL)

Balcones-Southwest, Inc. (TX)

Brown & Brown Agency of Insurance Professionals, Inc. (OK)

- Graham-Rogers, Inc. (OK)
- Graham-Rogers of Arkansas, Inc. (AR)
- Brown & Brown Insurance Agency of Virginia, Inc. (VA)
- Brown & Brown Insurance Benefits, Inc. (TX)
- Brown & Brown Insurance of Arizona, Inc. (AZ)
 - Azure IV Acquisition Corporation (AZ)
 - Brown & Brown of New Mexico, Inc. (NM)
- Brown & Brown Insurance of Georgia, Inc. (GA)
- Brown & Brown Insurance of Nevada, Inc. (NV)
- Brown & Brown Insurance Services of El Paso, Inc. (TX)
- Brown & Brown Insurance Services of San Antonio, Inc.(TX)
- Brown & Brown Insurance Services of Texas, Inc. (TX)
- Brown & Brown Metro, Inc. (NJ)
- Brown & Brown of Arkansas, Inc. (AR)
- Brown & Brown of Bartlesville, Inc. (OK)
- Brown & Brown of California, Inc. (CA)
- Brown & Brown of Central Oklahoma, Inc. (OK)

Schedule 5.4-1

Brown & Brown of Colorado, Inc. (CO) Brown & Brown of Connecticut, Inc. (CT) Brown & Brown of GF/EGF, Inc. (ND) Brown & Brown of Illinois, Inc. (IL) Brown & Brown of Iowa, Inc. (IA) Brown & Brown of Kentucky, Inc. (KY) Brown & Brown of Louisiana, Inc. (LA) Brown & Brown of Michigan, Inc. (MI) Brown & Brown of Minnesota, Inc. (MN) Brown & Brown of Mississippi, Inc. (MS) Brown & Brown of Missouri, Inc. (MO) Brown & Brown of New Hampshire, Inc. (NH) Brown & Brown of New Jersey, Inc. (NJ) Brown & Brown of Lehigh Valley, Inc. (PA) Brown & Brown of New York, Inc. (NY) Brown & Brown of North Dakota, Inc. (ND) Brown & Brown of Northern California, Inc. (CA) Brown & Brown of Ohio, Inc. (OH) Brown & Brown of Indiana, Inc. (IN) Brown & Brown of Southwest Indiana, Inc. (IN) Brown & Brown of Pennsylvania, Inc. (PA) Brown & Brown of South Carolina, Inc. (SC) Brown & Brown of Tennessee, Inc. (TN) Brown & Brown of Washington, Inc. (WA) • Azure VI Merger Co. (CA) Brown & Brown of West Virginia, Inc. (WV) Brown & Brown of Wisconsin, Inc. (WI) Brown & Brown of Wyoming, Inc. (WY) Brown & Brown Premium Finance Co. (VA) Brown & Brown Re, Inc. (CT) Brown & Brown Realty Co. (DE) M&J Buildings, LLC (ND) • Champion Underwriters, Inc. Conduit Insurance Managers, Inc. (TX) Energy & Marine Underwriters, Inc. (LA) John Manner Insurance Agency, Inc. (DE) Lancer Claims Services, Inc. (NV) Madoline Corporation (FL) • Florida Intracoastal Underwriters, Limited Company (FL) Payease Financial, Inc. (OK) Peachtree Special Risk Brokers of New York, LLC (NY) Peachtree Special Risk Brokers, LLC (GA) Peachtree Special Risk Insurance Brokers of NV, Inc. (NV) Peachtree West Insurance Brokers, Inc. (CA) Physician Protector Plan Risk Purchasing Group, Inc. (FL) Poe & Brown of North Carolina, Inc. (NC)

Schedule 5.4-2

Proctor Financial, Inc. (MI) Program Management Services, Inc. (FL) RAI of Mississippi, Inc. (MS) RAI of Oklahoma, Inc. (OK) RAI, Inc. (AR) Risk Management Associates, Inc. (FL) Roswell Insurance & Surety Agency, Inc. (NM) Technical Risks, Inc. (TX) TES Acquisition Corp. (CA) The Flagship Group, Ltd. (VA) The Flagship Group, Ltd. (VA) The Young Agency, Inc. (NY) Title Pac, Inc. (OK) Underwriters Services, Inc. (FL) Unified Seniors Association, Inc. (GA non-profit)

Schedule 5.4-3

SCHEDULE 5.5

Financial Statements

1. 2003 Annual Report to Shareholders, including financial statements

2. Annual Report on Form 10-K for the period ended December 31, 2003, filed March 15, 2004

3. Quarterly Report on Form 10-Q for the period ended March 31, 2004, filed May 10, 2004

Schedule 5.5

Certain Litigation

Patents

ERISA Affiliates and Plans

_

PLAN TYPE	CARRIER/TPA/RECORD KEEPER	PLAN LEVEL (Organization, Group, or Branch)	DESCRIPTION
Medical	United Benefits	Group	Approximately 90 locations
Dental	Guardian	Organization	Organization-wide
Short-term Disability	Self-funded	Organization	Organization-wide
Long-term Disability	Principal	Organization	Organization-wide
Group Term Life	Principal	Organization	Organization-wide
AD&D	Cigna	Organization	Organization-wide
Voluntary Life - Employee	Principal	Organization	Organization-wide
Voluntary Life - Spouse	Principal	Organization	Organization-wide
Voluntary Life - Child	Principal	Organization	Organization-wide
FSA - Premiums	United Benefits	Organization	Organization-wide
FSA - Healthcare	United Benefits	Organization	Organization-wide
FSA - Dependent Care	United Benefits	Organization	Organization-wide
401(K)	Diversified Investment Advisors	Organization	Organization-wide
Defined Benefit - Frozen	Principal	Group	Former Poe Employees
Non-qualified Deferred Comp.	B&B and The Pangburn Company	Employee	Lesueur, Chuck
Non-qualified Deferred Comp.	B&B - self managed	Group	Former Poe Employees
Life Insurance -w/BB as Beneficiary	Multiple	Employee	Current & Former Executives
Medical-Other	Anthem BCBS	Group	VA Branches
Dental-Other	Anthem BCBS	Group	VA Branches
Medical-Other	Preferred Care	Group	Rochester, Wolcott (S. Tier), NY
Medical-Other	Anthem BCBS	Branch	Hartford, CT
Medical-Other	Horizon BCBS of NJ	Branch	Florham Park, NJ
Dental-Other	Horizon BCBS Dental PPO	Branch	Florham Park, NJ
Dental-Other	Horizon BCBS Dental DMO	Branch	Florham Park, NJ
Medical-Other	BCBS of Rochester	Branch	Syracuse, NY
Medical-Other	HealthNOW HMO	Branch	Syracuse, NY
Medical-Other	MVP HMO	Branch	Syracuse, NY
Medical-Other	MVP HMO (different plan)	Branch	Endicott, NY
Medical-Other	Gunderson Lutheran Health Plan	Branch	LaCrosse, WI
Medical-Other	Horizon BCBS of NJ	Branch	Marmora, NJ
Medical-Other	BCBS of WNY	Branch	Buffalo, Jamestown, NY
Medical-Other	Univera Health Care	Branch	Buffalo, NY
Medical-Other	Empire BCBS-HMO	Branch	Clifton Park, NY
Medical-Other	Horizon BCBS of NJ	Branch	New Jersey
Medical-Other	BCBS of Michigan	Branch	Flint,Clarkston, MI
Medical-Other	Capital BCBS	Group	Bethlehem, AFC, PA
Medical-Other	Anthem BCBS PPO	Branch	Manassas, VA
Medical-Other	Anthem Healthkeepers HMO	Branch	Manassas, VA
Medical-Other	Independence Blue Cross	Branch	Philadelphia, PA
Medical-Other	United Healthcare	Branch	Rome, NY

Schedule 5.12-1

Medical-Other Medical-Other Medical-Other Medical-Other Medical-Other Medical-Other Medical-Other Medical-Other Medical-Other Vision-Other Medical-Other Dental-Other Vision-Other Medical-Other Medical-Other Medical-Other Medical-Other Medical-Other Vision-Other Medical-Other Vision-Other Medical Medical

BCBS Branch Capitol Health Plan Branch Humana Branch Kaiser Permanete Branch AmCare Branch Humana Group Principal Group BC of CA HMO Saver/PPO Branch BC of CA California Care Group Vision Service Plan Group BC of CA POS/HMO Group BC of CA POS/HMO Group BC of CA POS/HMO Group Health Plan of Nevada Branch Medica Health Ins Co Branch United Health Care Group BCBS of North Dakota Group Pacificare of WA Group Vision Services Plan Group Predent Buyer Group Group Group Pacific Vison Care Fortis Group CIGNA Healthcare of NH - HMO Group

Panama City, FL Tallahasse, FL Balcones-Southwest, TX Insurance Marketing Group, GA El Paso, TX AZ Branches, El Paso NM Branches Brea, CA Several CA branches Las Vegas, NV Mankato, MN Colorado Branches North Dakota WA Offices WA Offices WA Offices WA Offices WITTA McDuffee

Schedule 5.12-2

Use of Proceeds

The proceeds from the sale of the Notes will be used for general corporate purposes including repayment of existing senior debt and acquisitions.

Existing Indebtedness

Amended and Restated Revolving and Term Loan Agreement, dated as of January 3, 2001, by and among Brown & Brown, Inc. and SunTrust Bank, as amended, with an outstanding principal balance of \$45,000,000.06 and total availability of \$0.

Revolving Loan Agreement, dated as of September 29, 2003, by and among Brown & Brown, Inc. and SunTrust Bank, as amended, with an outstanding principal balance of \$14,940,000 and total availability of \$60,060,000.

Guaranty Agreement, dated as of July 15, 2004, by the following direct or indirect subsidiaries of Brown & Brown, Inc. in favor of the Noteholders (purchasers of the Notes issued pursuant to the terms of the Note Purchase Agreement, dated as of July 15, 2004):

B & B Insurance Services, Inc. Brown & Brown of California, Inc, Brown & Brown of Lehigh Valley, Inc. Brown & Brown of Washington, Inc. Brown & Brown Insurance of Arizona, Inc. Brown & Brown Insurance of Nevada, Inc. Florida Intracoastal Underwriters, Limited Company MacDuff Underwriters, Inc. Madoline Corporation Peachtree Special Risk Brokers, LLC Program Management Services, Inc. The Young Agency, Inc.

EXHIBIT 4.2

FIRST AMENDMENT TO AMENDED AND RESTATED <u>REVOLVING AND TERM LOAN AGREEMENT</u>

THIS FIRST AMENDMENT TO AMENDED AND RESTATED REVOLVING AND TERM LOAN AGREEMENT (the "<u>First Amendment</u>"), dated and effective as of July 15, 2004, is made and entered into by and between **BROWN & BROWN, INC.**, a Florida corporation (the "<u>Borrower</u>"), and **SUNTRUST BANK**, a Georgia corporation (the "Lender").

WITNESSETH:

WHEREAS, on or about January 3, 2001, the Borrower and the Lender entered into that certain Amended and Restated Revolving and Term Loan Agreement (the "Initial Term Loan Agreement") providing for a revolving loan and a term loan all as provided in the Initial Term Loan Agreement; and

WHEREAS, the Initial Term Loan Agreement contained various restrictions on the ability of the Borrower to incur other debt including, but not limited to, a restriction in Section 8.1(f) of the Initial Term Loan Agreement precluding the Borrower from incurring at any time any unsecured indebtedness exceeding the aggregate amount of \$25,000,000; and

WHEREAS, the Borrower desires to incur unsecured Indebtedness up to the principal amount of \$200,000,000 through the issuance of promissory notes to one or more investors (the "2004 Note Offering"); and

WHEREAS, the Borrower and the Lender wish to amend the Initial Term Loan Agreement so as to permit, among other matters, the 2004 Note Offering in accordance with the terms and conditions of this First Amendment.

NOW, THEREFORE, in consideration of the mutual covenants made herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, agree as follows:

1. **Definitions.** Unless defined in this First Amendment, capitalized terms contained herein shall have the meaning set forth in the Initial Term Loan Agreement.

2. Amendment of Existing Definition. The following existing definitions contained in Article I of the Initial Term Loan Agreement are hereby amended:

"<u>Permitted Acquisitions</u>" shall mean the acquisition, by merger, consolidation, purchase or otherwise, by any Consolidated Company of any Person where substantially all the assets or stock of said Person who is not affiliated with the Borrower are purchased, to the extent after giving effect to said acquisition, no Event of Default will occur or be continuing and either (i) the Funded Debt to Consolidated EBITDA Ratio will not be greater than 2.0:1; or (ii) the Funded Debt to Consolidated EBITDA Ratio will be greater than 2.0:1, then, in that event, only to the extent the aggregate value of said acquisitions after said 2.0:1 Ratio is met is not greater than \$75,000,000.

3. <u>New Definitions.</u> The following definitions are added to Article I of the Initial Term Loan Agreement:

"2004 Note Offering" shall mean that certain transaction by which the Borrower has incurred Indebtedness up to the maximum principal amount of \$200,000,000 all pursuant to the 2004 Note Purchase Agreement.

"2004 Note Purchase Agreement" shall mean that certain Note Purchase Agreement between the Borrower and the Purchaser's scheduled thereto and dated July 15, 2004 by which the Borrower has issued both Series A Notes and Series B Notes, as in effect as of the date of this First Amendment, as the same may be amended or modified from time to time.

4. <u>Amendment to Initial Term Loan Agreement</u>. The Initial Term Loan Agreement is hereby amended as follows:

(a) Section 7.8 of the Initial Term Loan Agreement captioned "<u>Maintain the Following Financial Covenants</u>" is hereby amended in its entirety to read as follows:

"Section 7.8 Maintain the Following Financial Covenants.

(a) Consolidated Net Worth of a minimum of the sum of (i) \$400,000,000 <u>plus</u> (ii) 50% of cumulative positive Net Income after September 30, 2003, <u>plus</u> (iii) 100% of net cash raised through contribution or issuance of new equity, <u>less</u> (iv) receivables from affiliates.

(b) A Fixed Charge Coverage Ratio of not less than 2.50 to 1.00 (The Fixed Charge Coverage Ratio means, at the end of any fiscal quarter, the ratio of (a) the sum of (i) Consolidated EBITDA *plus* (ii) Consolidated Rental Expense, both calculated for the period of four consecutive fiscal quarters then ended to (b) the sum of (i) Consolidated Interest Expense *plus* (ii) Consolidated Rental Expense, both calculated for such period.)

(c) A ratio of Consolidated Net Indebtedness as of the end of any fiscal year of the Company to Consolidated EBIDTA, for the period of four consecutive fiscal quarters of the Company ending with and including such fiscal quarter, not greater than 2.75 to 1.00.

The foregoing covenants will be tested quarterly."

(b) There is hereby added the following sentence at the end of Section 7.10 captioned "Additional Guarantors":

"Upon the request of the Borrower, the Lender will from time to time consider releasing the Guaranty of a Subsidiary who is not a Material Subsidiary provided that (i) no Event of Default then exists or would exist immediately after the release of said Subsidiary Guaranty, (ii) the circumstances or conditions by which said Subsidiary was no longer a Material Subsidiary was not otherwise prohibited under this Agreement, and (iii) the remaining Subsidiaries satisfy the financial covenants and requirements set forth in the defined term of "<u>Material Subsidiary</u>". For a Subsidiary to be released hereunder, the written approval and consent of the Lender will be required."

(c) There is hereby added to Section 8.1 of the Initial Term Loan Agreement the following subsection (g):

"(g) Unsecured Indebtedness not to exceed at any time the aggregate amount of \$200,000,000 due under the 2004 Note Offering."

(d) Section 8.8 of the Initial Term Loan Agreement captioned "Optional Prepayments" is hereby amended in its entirety to read as follows:

"Section 8.8 <u>Optional Prepayments</u>. Directly or indirectly, prepay, purchase, redeem, retire, defease or otherwise acquire, or make any optional payment on account of any principal of, interest on, or premium payable in connection with the optional prepayment, redemption or retirement of, any of its Indebtedness, or give a notice of redemption with respect to any such Indebtedness, or make any payment in violation of the subordination provisions of any Subordinated Debt, except with respect to (a) the Obligations under this Agreement and the Notes, (b) prepayments of Indebtedness outstanding pursuant to revolving credit, overdraft and line of credit facilities permitted pursuant to <u>Section 8.1 (c), (d), (g) and (h</u>), (c) Intercompany Loans made or outstanding pursuant to <u>Section 8.1 (e)</u>, (d) Intercompany Loans where both Consolidated Companies are not Credit Parties made or outstanding pursuant to <u>Section 8.1 (e)</u>. (d) Intercompany Loans where both Consolidated Debt, upon the prior written consent of the Lender, (e) Subordinated Debt, upon the prior written consent, of the Lender, and (f) Indebtedness under the 2004 Note Purchase Agreement."

(e) Section 8.11 of the Initial Term Loan Agreement captioned "Additional Negative Pledges" is hereby amended in its entirety to read as follows:

"Section 8.11 <u>Additional Negative Pledges</u>. Create or otherwise cause or suffer to exist or become effective, directly or indirectly, any prohibition or restriction on the creation or existence of any Lien upon any asset of any Consolidated Company, other than pursuant to (a) the terms of any agreement, instrument or other document pursuant to which any Indebtedness permitted by <u>Section 8.2(b)</u> is incurred by any Consolidated Company, so long as such prohibition or restriction applies only to the property or asset being financed by such Indebtedness, (b) any requirement of applicable law or any regulatory authority having jurisdiction over any of the Consolidated Companies, and (c) any such provisions contained in the 2004 Note Purchase Agreement."

(f) Section 8.12 of the Initial Term Loan Agreement captioned "Limitation on Payment Restrictions Affecting Consolidated Companies" is hereby amended in its entirety to read as follows:

"Section 8.12 Limitation on Payment Restrictions Affecting Consolidated Companies. Create or otherwise cause or suffer to exist or become effective, any consensual encumbrance or restriction on the ability of any Consolidated Company to (a) pay dividends or make any other distributions on such Consolidated Company's stock, or (b) pay any indebtedness owed to Borrower or any other Consolidated Company, or (c) transfer any of its property or assets to Borrower or any other Consolidated Company, except in each case any consensual encumbrance or restriction existing under the Credit Documents, or those restrictions as are contained in the 2004 Note Purchase Agreement."

(g) Section 8.17 of the Initial Term Loan Agreement captioned "Guaranties" is hereby amended in its entirety to read as follows:

"Section 8.17 <u>Guaranties</u>. Without the prior written consent of the Lender, extend or execute any Guaranty other than (a) endorsements of instruments for deposit or collection in the ordinary and normal course of business, (b) Guaranties acceptable in writing to the Lender, and (c) Guaranties for obligations of any Consolidated Subsidiary, <u>provided</u>, <u>however</u>, said Guaranteed Indebtedness under this subparagraph (c) will not exceed the aggregate amount of \$10,000,000 without the prior written consent of the Lender, and (d) Guaranties of Subsidiaries in connection with the 2004 Note Purchase Agreement."

(h) Section 8.18 of the Initial Term Loan Agreement captioned "Changes in Debt Instruments" is hereby amended in its entirety to read as follows:

"Section 8.18 <u>Changes in Debt Instruments</u>. Without the prior written consent of the Lender, enter into any amendment, change or modification of any agreement relating to any Indebtedness (excluding Indebtedness described in the 2004 Note Purchase Agreement); <u>provided</u>, <u>however</u>, the foregoing restrictions shall not prohibit any such amendment, change or

modification where (a) it relates solely to an extension of a maturity date if said Indebtedness is not already in default, and (b) other changes in said agreements which are not material; **provided**, **however**, if said amendment, change or modification constitutes the waiver of any default condition under said agreement, notice of said matter along with a copy of said amendment, change or modification shall be given to the Lender."

5. <u>Ratification</u>. Except as modified by this First Amendment, the parties do hereby confirm and ratify the Initial Term Loan Agreement. Hereafter, the term "<u>Term Loan Agreement</u>" means and includes this First Amendment.

6. <u>Conflicts Between Term Loan and Revolving Loan</u>. In addition to the term loan evidenced by the Term Loan Agreement, the Lender has also extended to the Borrower a separate and distinct revolving loan which is evidenced by that certain Revolving Loan Agreement (as amended or modified, the "<u>Revolving Loan Agreement</u>") dated September 29, 2003, as amended by that certain First Amendment to Revolving Loan Agreement dated December 30, 2003 and that certain Second Amendment to Revolving Loan Agreement dated July 15, 2004. To the extent there is a conflict in the provisions of said Term Loan Agreement and Revolving Loan Agreement in the Articles relating to Affirmative Covenants or Negative Covenants (including any definitions relating to or used in said Affirmative or Negative Covenants), the terms and conditions of the Revolving Loan Agreement shall govern.

Signature Page Follows

SIGNATURE PAGE TO FIRST AMENDMENT

IN WITNESS WHEREOF, the parties hereto have caused this First Amendment to Amended and Restated Revolving and Term Loan Agreement to be duly executed and delivered by their duly authorized officers as of the day and year first above written.

BORROWER:

BROWN & BROWN, INC.

By: /S/ CORY T. WALKER

Cory T. Walker, Senior Vice President, Treasurer and Chief Financial Officer

LENDER:

SUNTRUST BANK

By: /S/ EDWARD WOOTEN

Edward Wooten, Director

Address for Notices:

220 South Ridgewood Avenue Daytona Beach, Florida 23115-2412 Attention: Cory T. Walker Telephone No.: (386) 239-7250 Telecopy No.: (386) 239-7252

With a copy to:

Laurel L. Grammig General Counsel **BROWN & BROWN, INC.** 401 East Jackson Street Suite 1700 Tampa, Florida 33602 Telephone No.: (813) 222-4277 Telecopy No.: (813) 222-4464

Address for Notices:

SunTrust Bank Mail Code FL-Orlando-1106 200 South Orange Avenue Tower 10 Orlando, FL 32801 Telephone: (407) 237-6855 Telecopy: (407) 237-4076

EXHIBIT 4.3

SECOND AMENDMENT TO REVOLVING LOAN AGREEMENT

THIS SECOND AMENDMENT TO REVOLVING LOAN AGREEMENT, dated and effective as of July 15, 2004 (the "<u>Second Amendment</u>"), is made and entered into by and between **BROWN & BROWN, INC.**, a Florida corporation (the "<u>Borrower</u>"), and **SUNTRUST BANK**, a Georgia corporation (the "<u>Lender</u>").

WITNESSETH:

WHEREAS, on or about September 29, 2003, the Borrower and the Lender entered into that certain Revolving Loan Agreement (the "<u>Initial Revolving</u> <u>Loan Agreement</u>") providing for a revolving loan up to the maximum amount of \$75,000,000 from the Lender to the Borrower. Subsequently, that Initial Revolving Loan Agreement was amended by virtue of that certain First Amendment To Revolving Loan Agreement (the "<u>First Amendment</u>") dated December 30, 2003. Hereafter, the term "<u>Initial Revolving Loan Agreement</u>" includes the First Amendment; and

WHEREAS, the Initial Revolving Loan Agreement contained various restrictions on the ability of the Borrower to incur other debt including, but not limited to, a restriction in Section 8.1(g) of the Initial Revolving Loan Agreement precluding the Borrower from incurring at any time any unsecured indebtedness exceeding the aggregate amount of \$100,000,000; and

WHEREAS, the Borrower desires to incur unsecured indebtedness up to the principal amount of \$200,000,000 through the issuance of promissory notes to one or more investors (the "2004 Note Offering"); and

WHEREAS, the Borrower and the Lender wish to amend the Initial Revolving Loan Agreement so as to permit, among other matters, the 2004 Note Offering in accordance with the terms and conditions of this Second Amendment.

NOW, THEREFORE, in consideration of the mutual covenants made herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, agree as follows:

1. **Definitions**. Unless defined in this Second Amendment, capitalized terms contained herein shall have the meaning set forth in the Initial Revolving Loan Agreement.

2. <u>Amendment of Existing Definition</u>. The following existing definitions contained in Article I of the Initial Revolving Loan Agreement are hereby amended:

"<u>Material Subsidiary</u>" shall mean (a) each Subsidiary designated as such in <u>Schedule 6.1</u> hereto, and (b) each other Wholly Owned Subsidiary of Borrower, now existing or hereinafter established or acquired, that at any time prior to the Maturity Date, has either (i) at an annualized basis, net income which generates one and one-half percent (1.5%) or more of the Consolidated Net Income; <u>provided</u>, <u>however</u>, if the aggregate net income of the Borrower and its Material Subsidiaries is at any time less than eighty percent (80%) of Consolidated Net Income, then the one and one-half percent (1.5%) trigger set forth herein will be reduced to such a figure so that the aggregate net income of the Borrower and its Material Subsidiaries (based on said reduced trigger amount) is not less than eighty percent (80%) of the Consolidated Net Income, or (ii) any Material Assets. 3. New Definitions. The following definitions are added to Article I of the Initial Revolving Loan Agreement:

"2004 Note Offering" shall mean that certain transaction by which the Borrower has incurred Indebtedness up to the maximum principal amount of \$200,000,000 all pursuant to the 2004 Note Purchase Agreement.

"2004 Note Purchase Agreement" shall mean that certain Note Purchase Agreement between the Borrower and the Purchaser's scheduled thereto and dated July 15, 2004 by which the Borrower has issued both Series A Notes and Series B Notes, as in effect as of the date of this Second Amendment, as the same may be amended or modified from time to time.

4. <u>Amendment to Initial Revolving Loan Agreement</u>. The Initial Revolving Loan Agreement is hereby amended as follows:

(a) Section 6.25 of the Initial Revolving Loan Agreement captioned, "Guarantors – Income Requirement" is hereby amended in its entirety to read as follows:

"Section 6.25 <u>Guarantors – Income Requirement</u>. The aggregate net income of the Borrower and the Guarantors, on an annualized basis, is not less than eighty percent (80%) of the Consolidated Net Income."

(b) Section 7.8 of the Initial Revolving Loan Agreement captioned "Maintain the Following Financial Covenants" is hereby amended in its entirety to read as follows:

"Section 7.8 Maintain the Following Financial Covenants.

(a) Consolidated Net Worth of a minimum of the sum of (i) \$400,000,000 <u>plus</u> (ii) 50% of cumulative positive Net Income after September 30, 2003, <u>plus</u> (iii) 100% of net cash raised through contribution or issuance of new equity, <u>less</u> (iv) receivables from affiliates.

(b) A Fixed Charge Coverage Ratio of not less than 2.50 to 1.00 (The Fixed Charge Coverage Ratio means, at the end of any fiscal quarter, the ratio of (a) the sum of (i) Consolidated EBITDA *plus* (ii) Consolidated Rental Expense, both calculated for the period of four consecutive fiscal quarters then ended to (b) the sum of (i) Consolidated Interest Expense *plus* (ii) Consolidated Rental Expense, both calculated for such period.)

(c) A ratio of Consolidated Net Indebtedness as of the end of any fiscal year of the Company to Consolidated EBIDTA, for the period of four consecutive fiscal quarters of the Company ending with and including such fiscal quarter, not greater than 2.75 to 1.00.

The foregoing covenants will be tested quarterly."

(c) There is hereby added the following sentence at the end of Section 7.10 captioned "Additional Guarantors":

"Upon the request of the Borrower, the Lender will from time to time consider releasing the Guaranty of a Subsidiary who is not a Material Subsidiary provided that (i) no Event of Default then exists or would exist immediately after the release of said Subsidiary Guaranty, (ii) the circumstances or conditions by which said Subsidiary was no longer a Material Subsidiary was not otherwise prohibited under this Agreement, and (iii) the remaining Subsidiaries satisfy the financial covenants and requirements set forth in the defined term of "<u>Material Subsidiary</u>". For a Subsidiary to be released hereunder, the written approval and consent of the Lender will be required."

(d) There is hereby added to Section 8.1 of the Initial Revolving Loan Agreement the following subsection (h):

"(h) Unsecured Indebtedness not to exceed at any time the aggregate amount of \$200,000,000 due under the 2004 Note Offering."

(e) Section 8.8 of the Initial Revolving Loan Agreement captioned "Optional Prepayments" is hereby amended in its entirety to read as follows:

"Section 8.8 <u>Optional Prepayments</u>. Directly or indirectly, prepay, purchase, redeem, retire, defease or otherwise acquire, or make any optional payment on account of any principal of, interest on, or premium payable in connection with the optional prepayment, redemption or retirement of, any of its Indebtedness, or give a notice of redemption with respect to any such Indebtedness, or make any payment in violation of the subordination provisions of any Subordinated Debt, except with respect to (a) the Obligations under this Agreement and the Notes, (b) prepayments of Indebtedness outstanding pursuant to revolving credit, overdraft and line of credit facilities permitted pursuant to <u>Section 8.1 (c), (d), (g) and (h</u>), (c) Intercompany Loans made or outstanding pursuant to <u>Section 8.1 (e)</u>, (d) Intercompany Loans where both Consolidated Companies are not Credit Parties made or outstanding pursuant to <u>Section 8.1 (e)</u>. (d) Intercompany Loans where both Consolidated Debt, upon the prior written consent of the Lender, (e) Subordinated Debt, upon the prior written consent, of the Lender, and (f) Indebtedness under the 2004 Note Purchase Agreement."

(f) Section 8.11 of the Initial Revolving Loan Agreement captioned "Additional Negative Pledges" is hereby amended in its entirety to read as follows:

"Section 8.11 <u>Additional Negative Pledges</u>. Create or otherwise cause or suffer to exist or become effective, directly or indirectly, any prohibition or restriction on the creation or existence of any Lien upon any asset of any Consolidated Company, other than pursuant to (a) the terms of any agreement, instrument or other document pursuant to which any Indebtedness permitted by <u>Section 8.2(b)</u> is incurred by any Consolidated Company, so long as such prohibition or restriction applies only to the property or asset being financed by such Indebtedness, (b) any requirement of applicable law or any regulatory authority having jurisdiction over any of the Consolidated Companies, and (c) any such provisions contained in the 2004 Note Purchase Agreement."

(g) Section 8.12 of the Initial Revolving Loan Agreement captioned "Limitation on Payment Restrictions Affecting Consolidated Companies" is hereby amended in its entirety to read as follows:

"Section 8.12 Limitation on Payment Restrictions Affecting Consolidated Companies. Create or otherwise cause or suffer to exist or become effective, any consensual encumbrance or restriction on the ability of any Consolidated Company to (a) pay dividends or make any other distributions on such Consolidated Company's stock, or (b) pay any indebtedness owed to Borrower or any other Consolidated Company, or (c) transfer any of its property or assets to Borrower or any other Consolidated Company, except in each case any consensual encumbrance or restriction existing under the Credit Documents, or those restrictions as are contained in the 2004 Note Purchase Agreement."

(h) Section 8.17 of the Initial Revolving Loan Agreement captioned "Guaranties" is hereby amended in its entirety to read as follows:

"Section 8.17 <u>Guaranties</u>. Without the prior written consent of the Lender, extend or execute any Guaranty other than (a) endorsements of instruments for deposit or collection in the ordinary and normal course of business, (b) Guaranties acceptable in writing to the Lender, and (c) Guaranties for obligations of any Consolidated Subsidiary, <u>provided</u>, <u>however</u>, said Guaranteed Indebtedness under this subparagraph (c) will not exceed the aggregate amount of \$10,000,000 without the prior written consent of the Lender, and (d) Guaranties of Subsidiaries in connection with the 2004 Note Purchase Agreement."

(i) Section 8.18 of the Initial Revolving Loan Agreement captioned "<u>Changes in Debt Instruments</u>" is hereby amended in its entirety to read as follows:

"Section 8.18 <u>Changes in Debt Instruments</u>. Without the prior written consent of the Lender, enter into any amendment, change or modification of any agreement relating to any Indebtedness (excluding Indebtedness described in the 2004 Note Purchase Agreement); <u>provided</u>, <u>however</u>, the foregoing restrictions shall not prohibit any such amendment, change or modification where (a) it relates solely to an extension of a maturity date if said Indebtedness is not already in default, and (b) other changes in said agreements which are not material; <u>provided</u>, <u>however</u>, if said amendment, change or modification constitutes the waiver of any default condition under said agreement, notice of said matter along with a copy of said amendment, change or modification shall be given to the Lender."

5. <u>Ratification</u>. Except as modified by this Second Amendment, the parties do hereby confirm and ratify the Initial Revolving Loan Agreement. Hereafter, the term "<u>Revolving Loan Agreement</u>" means and includes this Second Amendment.

6. <u>Conflicts Between Term Loan and Revolving Loan</u>. In addition to the term loan evidenced by the Revolving Loan Agreement, the Lender has also extended to the Borrower a separate and distinct term loan which is evidenced by that certain Amended and Restated Revolving and Term Loan Agreement (as amended or modified, the "<u>Term Loan Agreement</u>") dated January 3, 2001, as amended by that certain First Amendment to Amended and Restated Revolving and Term Loan Agreement dated July 15, 2004. To the extent there is a conflict in the provisions of said Term Loan Agreement and Revolving Loan Agreement in the Articles relating to Affirmative Covenants or Negative Covenants (including any definitions relating to or used in said Affirmative or Negative Covenants), the terms and conditions of the Revolving Loan Agreement shall govern.

Signature Page Follows

SIGNATURE PAGE TO SECOND AMENDMENT TO REVOLVING LOAN AGREEMENT

IN WITNESS WHEREOF, the parties hereto have caused this Second Amendment to Revolving Loan Agreement to be duly executed and delivered by their duly authorized officers as of the day and year first above written.

BORROWER:

BROWN & BROWN, INC.

By: /S/ CORY T. WALKER

Cory T. Walker, Senior Vice President, Treasurer and Chief Financial Officer

Address for Notices:

220 South Ridgewood Avenue Daytona Beach, Florida 23115-2412 Attention: Cory T. Walker Telephone No.: (386) 239-7250 Telecopy No.: (386) 239-7252

With a copy to:

Laurel L. Grammig General Counsel **BROWN & BROWN, INC.** 401 East Jackson Street Suite 1700 Tampa, Florida 33602 Telephone No.: (813) 222-4277 Telecopy No.: (813) 222-4464

Address for Notices:

SunTrust Bank Mail Code FL-Orlando-1106 200 South Orange Avenue Tower 10 Orlando, FL 32801 Telephone: (407) 237-6855 Telecopy: (407) 237-4076

LENDER:

SUNTRUST BANK

By: /S/ EDWARD WOOTEN

Edward Wooten, Director

<u>Certification by the Chief Executive Officer</u> <u>Pursuant to Section 302 of The Sarbanes-Oxley Act of 2002</u>

I, J. Hyatt Brown, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Brown & Brown, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2004

/S/ J. HYATT BROWN

J. Hyatt Brown Chief Executive Officer

<u>Certification by the Chief Financial Officer</u> <u>Pursuant to Section 302 of The Sarbanes-Oxley Act of 2002</u>

I, Cory T. Walker, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Brown & Brown, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2004

/S/ CORY T. WALKER

Cory T. Walker Chief Financial Officer

Certification Pursuant to Section 1350 of Title 18 of the United States Code, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

I, J. Hyatt Brown, the chief executive officer of Brown & Brown, Inc., hereby certify, in my capacity as an officer of Brown & Brown, Inc. and to my actual knowledge, that:

(1) the Quarterly Report on Form 10-Q of Brown & Brown, Inc. for the quarterly period ended June 30, 2004 (the "<u>Report</u>") fully complies with the requirements of Section 13(a) or 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and

(2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Brown & Brown, Inc. and its subsidiaries.

Date: August 9, 2004

/S/ J. HYATT BROWN

J. Hyatt Brown Chief Executive Officer

A signed original of this written statement required by Section 906 has been provided to Brown & Brown, Inc. and will be retained by Brown & Brown, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

Certification Pursuant to Section 1350 of Title 18 of the United States Code, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

I, Cory T. Walker, the chief financial officer of Brown & Brown, Inc., hereby certify, in my capacity as an officer of Brown & Brown, Inc. and to my actual knowledge, that:

(1) the Quarterly Report on Form 10-Q of Brown & Brown, Inc. for the quarterly period ended June 30, 2004 (the "<u>Report</u>") fully complies with the requirements of Section 13(a) or 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and

(2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Brown & Brown, Inc. and its subsidiaries.

Date: August 9, 2004

/S/ CORY T. WALKER

Cory T. Walker Chief Financial Officer

A signed original of this written statement required by Section 906 has been provided to Brown & Brown, Inc. and will be retained by Brown & Brown, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.